2014 – 2017 COLLECTIVE BARGAINING AGREEMENT

UNIT S (CLASSIFIED SUPERVISORS)

LOS ANGELES UNIFIED SCHOOL DISTRICT

AND

TEAMSTERS LOCAL 572

TABLE OF CONTENTS

ARTICLE		<u>PAGE</u>	
	AGREEMENT		1
1.	RECOGNITION		1
II.	SEPARABILITY AND SAVINGS		_6
III.	DISTRICT RIGHTS		7
IV.	UNION RIGHTS		_10
V.	GRIEVANCE PROCEDURE		. 12
VI.	WORK STOPPAGE		18
VII.	NON-DISCRIMINATION		_20
VIII.	UNION SECURITY AND DUES DEDUCTION		_21
IX.	HOURS OF WORK		_24
Χ.	EVALUATION PROCEDURES		_28
XI.	TRANSFER PROCEDURES		33
XII.	SUBCONTRACTING		35
XIII.	LEAVES OF ABSENCE		36
XIV.	WAGES AND SALARIES, PAY ALLOWANCES, DIFFERENTIALS AND SPECIAL SALARY PRACTICES		.5 <u>2</u>
XV.	HEALTH AND WELFARE		60
XVI.	HOLIDAYS		71
XVII.	VACATION		73
XVIII.	SAFETY CONDITIONS		<u>79</u>
XIX.	TOOL REPLACEMENT		80
XX.	TUITION REIMBURSEMENT		81
XXI.	RECLASSIFICATION PROCEDURE		83

TABLE OF CONTENTS

ARTICLE XXII.	JOB STEWARDS	<u>PAGE</u>	0.1
AAII.	JOB STEWARDS	•••••	<u>04</u>
XXIII.	ENTIRE AGREEMENT		<u>85</u>
XXIV.	TERM OF AGREEMENT		<u>86</u>
APPENDIX A	MEMORANDUM OF UNDERSTANDING 2014-2017		<u>87</u>
APPENDIX B	TRANSPORTATION ADDENDUM		<u>89</u>
APPENDIX C	FLEET MAINTENANCE ADDENDUM		<u>91</u>
APPENDIX D	SCHOOL ADMINISTRATIVE ASSISTANT ADDENDUM		<u>92</u>
APPENDIX E	PLANT MANAGERS ADDENDUM		<u>94</u>
APPENDIX F	FOOD SERVICES ADDENDUM		<u>96</u>
APPENDIX G	CONSTRUCTION INSPECTORS ADDENDUM		<u>99</u>
APPENDIX H	SENIOR GARDENERS ADDENDUM		<u>100</u>
APPENDIX I	SUPERVISING SPECIAL EDUCATION ASSISTANTS ADDENDUM	1	<u>102</u>
APPENDIX J	2014-2017 SALARY SCHEDULES	•••••	<u>103</u>
APPENDIX K	RESPECTFUL TREATMENT		125
APPENDIX L	SALARY REOPENER MEMORANDUM OF UNDERSTANDING	•••••	126

AGREEMENT

THIS AGREEMENT is made and entered into this <u>3rd</u> day of <u>December</u> 2014, by and between the Board of Education of the Los Angeles Unified School District, which together with its administrative staff and representatives will be referred to in this Agreement as the "District" and Teamsters, Local 572, which together with its officers and representatives will be referred to in this Agreement as "the Union".

ARTICLE I

RECOGNITION

1.0 <u>The Unit</u>: Pursuant to applicable California statutes, regulations, and the Certification of Representation dated July 1, 1996, in PERB Case No. LA-R-1009, the District acknowledges that Teamsters has been certified as the exclusive representative of a bargaining unit comprised of all regular employees in probationary and permanent status, including part-time employees, employed in the following classes:

1161 Accountant 1101 Accounting Analyst 1101 Accounting System Specialist 1127 Accounting System Specialist 1128 Accounting System Specialist 1129 Administrative Assistant, Building 1129 Assistant Contracts Supervisor 1120 Administrative Assistant, Building 1120 Program 1121 Assistant Food Production Manager 1122 Assistant Garage Supervisor 1222 Assistant Gardening Supervisor 1223 Assistant Gardening Supervisor 12328 Assistant Hardware Supervisor 12328 Assistant Logistics Supervisor 12329 Assistant Mail Supervisor 12320 Area Bus Supervisor 12320 Area Bus Supervisor 12320 Area Electrical Supervisor 12321 Area Carpentry Supervisor 12321 Area Food Services Supervisor 12322 Assistant Plant Manager 13323 Assistant Projects Manager 13324 Area Heating and Air Conditioning 14335 Associate Civil Engineer 14346 Associate Civil Engineer 14347 Associate Civil Engineer	2375	Absence Coordinator	3165	Asbestos Abatement Supervisor
1127Accounting System Specialist2057Assistant Contracts Supervisor2170Administrative Assistant, Building Program4329Assistant Food Production Manager S7552406Administrative Secretary I4173Assistant Gardening Supervisor2150Adult Education Administrative Specialist3258Assistant Hardware Supervisor2150Adult Education Administrative Specialist5140Assistant Logistics Supervisor1444Architect2218Assistant Mail Supervisor1438Architectural Associate4574Assistant Outdoor-Education Center1460Architectural Project FacilitatorManager5804Area Bus Supervisor4061Assistant Plant Manager I3271Area Carpentry Supervisor4048Assistant Plant Manager II3206Area Electrical Supervisor1791Assistant Projects Manager4314Area Food Services Supervisor1632Assistant Supervising3211Area Heating and Air Conditioning SupervisorBuilding/Construction InspectorSupervisor1531Associate Civil Engineer	1161	Accountant	5806	Assistant Area Bus Supervisor
Administrative Assistant, Building Program 5755 Assistant Food Production Manager Supervisor Administrative Secretary I 4173 Assistant Gardening Supervisor Adult Education Administrative 3258 Assistant Hardware Supervisor Specialist 5140 Assistant Logistics Supervisor 4474 Architect 2218 Assistant Mail Supervisor 4574 Assistant Outdoor-Education Center 4574 Area Bus Supervisor 4061 Assistant Plant Manager I 3271 Area Carpentry Supervisor 4048 Assistant Plant Manager II 4326 Area Electrical Supervisor 1791 Assistant Plant Manager I 4314 Area Food Services Supervisor 1632 Assistant Supervising 8211 Area Heating and Air Conditioning Supervisor 1531 Associate Civil Engineer	1101	Accounting Analyst	1541	Assistant Civil Engineer
Program 5755 Assistant Garage Supervisor Administrative Secretary I 4173 Assistant Gardening Supervisor Adult Education Administrative 3258 Assistant Hardware Supervisor Specialist 5140 Assistant Logistics Supervisor 1444 Architect 2218 Assistant Mail Supervisor 1438 Architectural Associate 4574 Assistant Outdoor-Education Center 1460 Architectural Project Facilitator Manager 5804 Area Bus Supervisor 4061 Assistant Plant Manager I 3271 Area Carpentry Supervisor 4048 Assistant Plant Manager II 3206 Area Electrical Supervisor 1791 Assistant Projects Manager 4314 Area Food Services Supervisor 1632 Assistant Supervising 3211 Area Heating and Air Conditioning Building/Construction Inspector Supervisor 1531 Associate Civil Engineer	1127	Accounting System Specialist	2057	Assistant Contracts Supervisor
2406Administrative Secretary I4173Assistant Gardening Supervisor2150Adult Education Administrative Specialist3258Assistant Hardware Supervisor1444Architect5140Assistant Logistics Supervisor1438Architectural Associate4574Assistant Mail Supervisor1460Architectural Project FacilitatorManager5804Area Bus Supervisor4061Assistant Plant Manager I3271Area Carpentry Supervisor4048Assistant Plant Manager II3206Area Electrical Supervisor1791Assistant Projects Manager4314Area Food Services Supervisor1632Assistant Supervising3211Area Heating and Air Conditioning SupervisorBuilding/Construction Inspector531Associate Civil Engineer	2170	Administrative Assistant, Building	4329	Assistant Food Production Manager
2150 Adult Education Administrative Specialist Speciali		Program	5755	Assistant Garage Supervisor
Specialist 5140 Assistant Logistics Supervisor 1444 Architect 2218 Assistant Mail Supervisor 1438 Architectural Associate 4574 Assistant Outdoor-Education Center 1460 Architectural Project Facilitator Manager 5804 Area Bus Supervisor 4061 Assistant Plant Manager I 3271 Area Carpentry Supervisor 4048 Assistant Plant Manager II 3206 Area Electrical Supervisor 1791 Assistant Projects Manager 4314 Area Food Services Supervisor 1632 Assistant Supervising 3211 Area Heating and Air Conditioning Building/Construction Inspector Supervisor 1531 Associate Civil Engineer	2406	Administrative Secretary 1	4173	Assistant Gardening Supervisor
1444Architect2218Assistant Mail Supervisor1438Architectural Associate4574Assistant Outdoor-Education Center1460Architectural Project FacilitatorManager5804Area Bus Supervisor4061Assistant Plant Manager I3271Area Carpentry Supervisor4048Assistant Plant Manager II3206Area Electrical Supervisor1791Assistant Projects Manager4314Area Food Services Supervisor1632Assistant Supervising3211Area Heating and Air Conditioning SupervisorBuilding/Construction Inspector531Associate Civil Engineer	2150	Adult Education Administrative	3258	Assistant Hardware Supervisor
1438Architectural Associate4574Assistant Outdoor-Education Center1460Architectural Project FacilitatorManager5804Area Bus Supervisor4061Assistant Plant Manager I3271Area Carpentry Supervisor4048Assistant Plant Manager II3206Area Electrical Supervisor1791Assistant Projects Manager4314Area Food Services Supervisor1632Assistant Supervising3211Area Heating and Air Conditioning SupervisorBuilding/Construction Inspector531Associate Civil Engineer		Specialist	5140	Assistant Logistics Supervisor
1460Architectural Project FacilitatorManager5804Area Bus Supervisor4061Assistant Plant Manager I3271Area Carpentry Supervisor4048Assistant Plant Manager II3206Area Electrical Supervisor1791Assistant Projects Manager4314Area Food Services Supervisor1632Assistant Supervising3211Area Heating and Air Conditioning SupervisorBuilding/Construction Inspector5001531Associate Civil Engineer	1444	Architect	2218	Assistant Mail Supervisor
5804Area Bus Supervisor4061Assistant Plant Manager I3271Area Carpentry Supervisor4048Assistant Plant Manager II3206Area Electrical Supervisor1791Assistant Projects Manager4314Area Food Services Supervisor1632Assistant Supervising3211Area Heating and Air Conditioning SupervisorBuilding/Construction Inspector531Associate Civil Engineer	1438	Architectural Associate	4574	Assistant Outdoor-Education Center
3271Area Carpentry Supervisor4048Assistant Plant Manager II3206Area Electrical Supervisor1791Assistant Projects Manager4314Area Food Services Supervisor1632Assistant Supervising3211Area Heating and Air Conditioning SupervisorBuilding/Construction Inspector531Associate Civil Engineer	1460	Architectural Project Facilitator		Manager
3206 Area Electrical Supervisor 1791 Assistant Projects Manager 4314 Area Food Services Supervisor 1632 Assistant Supervising 3211 Area Heating and Air Conditioning Building/Construction Inspector Supervisor 1531 Associate Civil Engineer	5804	Area Bus Supervisor	4061	Assistant Plant Manager I
4314 Area Food Services Supervisor 3211 Area Heating and Air Conditioning Supervisor 1632 Assistant Supervising Building/Construction Inspector 1531 Associate Civil Engineer	3 <u>27</u> 1	Area Carpentry Supervisor	4048	Assistant Plant Manager II
3211 Area Heating and Air Conditioning Building/Construction Inspector Supervisor 1531 Associate Civil Engineer	3206	Area Electrical Supervisor	1791	Assistant Projects Manager
Supervisor 1531 Associate Civil Engineer	4314	Area Food Services Supervisor	1632	Assistant Supervising
	3211	Area Heating and Air Conditioning		Building/Construction Inspector
222C Avec Metal Ment Companies 1700 Accepte Flactured Francisco		Supervisor	1531	Associate Civil Engineer
3236 Area ivietal work Supervisor 1709 Associate Electrical Engineer	3236	Area Metal Work Supervisor	1709	Associate Electrical Engineer
4016 Area Operations Supervisor 1763 Associate Mechanical Engineer	4016	Area Operations Supervisor	1763	Associate Mechanical Engineer
3276 Area Painting Supervisor 1651 Building/Construction Inspector	3276	Area Painting Supervisor	1651	Building/Construction Inspector
3216 Area Plumbing Supervisor 1650 Building/Construction Inspector	3216	Area Plumbing Supervisor	1650	Building/Construction Inspector

	(Class 1)	1435	Facilities Partnership and
1471	Building/Project Estimator		Development Coordinator
5731	Bus Dispatcher	1813	Facilities Services Coordinator
3192	Carpentry Supervisor (Building	5748	Fleet Technical Support Specialist
	Moving	3251	Floor Covering Supervisor
3148	Carpentry Technical Supervisor	3128	Floor Covering Technical Supervisor
<u> 2620</u>	<u>Cataloguing Supervisor</u>	4330	Food Production Supervisor
3160	Central Shops Supervisor	3121	Food Service Equipment Specialist
2110	Chief Broadcasting Engineer	4291	Food Service Manager I
2511	Chief Clerk	4292	Food Service Manager II
1501	Chief Enrollment Analysis Coordinator	4293	Food Service manager III
5192	Chief Inventory Analyst	4296	Food Service Manager IV
1528	Civil Engineer	4297	Food Service Manager V
2045	Claims Processing Supervisor	4294	Food Service Manager VI
2056	Communications Production	4295	Food Service Manager VII
2030	Facilitator	4312	Food Service Training Specialist
4817	Computer Applications	1071	Food Services Cost Analyst
4017		5756	Garage Supervisor
.=	Support Supervisor	3266	Glazing Supervisor
1792	Construction Safety Specialist	3051	Graphic Arts Machinist Supervisor
2531	Contract Assistant	3256	Hardware Supervisor
1125	Contract Payments Supervisor	5201	Head Stock Clerk
2534	Coordinating Contract Assistant	3137	Heating and Air Conditioning
2138	Coordinator of Telecommunications		Technical Supervisor
	Services	3132	HVAC Technical Supervisor (High-Rise)
1990	Cost Recovery Claims Processing	3134	HVAC Test and Inspection Supervisor
	Supervisor	<u>1782</u>	Industrial Hygienist
2236	Credentials and Contracts	1193	Information Technology Project
1201	Supervisor		Manager
1301	Customer Services Center Supervisor	5196	Instructional Materiel Order
1073	Data Base Specialist		Supervisor
4881	Design Network Engineer	4620	Instructional Media Technician-
4600	Disability Support Program Assistant		Advis <u>e</u> r
4624	Display and Graphic Designer	4156	Integrated Pest Management
5734	Driver Trainer (Bus/Truck)		Program Coordinator
5739	Driver Trainer (Truck)	1151	Intermediate Accountant
3126	Electrical Technical Supervisor	3031	IT Electronics Communications
3522	Electronics Technical Supervisor		Supervisor
5034	Employment Services Manager	3543	IT Field Services Supervisor
1778	Environmental Assessment	4880	IT Infrastructure Project
	Coordinator		Network Engineer
3015	Environmental Compliance	4812	IT Training Project Manager
	Coordinator	4863	IT Training Supervisor
1801	Environmental Health Supervisor	2052	Labor Compliance Officer
2143	Escrow and Title Supervisor	4116	Landscaping Supervisor
1998	Ethics Adviser	5135	Logistics Supervisor
1963	Facilities Cost Analyst	3056	Machinist Supervisor
		2214	Mail Supervisor

3158	Maintenance Materiel Coordinator	5370	Program Coordinator, Out-of-school
3020	Maintenance Planner	5090	Program Evaluation and Research
2183	Marketing Representative		Coordinator
1959	Masterplan Coordinator	1381	Quality Control and Assurance
1757	Mechanical Engineer		Specialist
2385	Medical Administrative Assistant	3523	Radio Communications Supervisor
3140	Metal Technical Supervisor	5211	Receiving Inspector
2561	Microfilm Supervisor	2198	Records Manager
4844	Mobile Device Management Specialist	3226	Refrigeration Supervisor
3081	Musical Instrument Repair	5305	Regional Recreation Director
	Supervisor	1663	Relocatable Housing
3588	Network Operations Supervisor		Manufacturing Inspector
1178	Network Security Administrator	4759	Reprographic Services Coordinator
4317	Nutrition Specialist	1048	Retirement Systems Supervisor
5387	Operations Compliance	3280	Roofing Inspection Supervisor
	Manager (LA's Best)	3281	Roofing Supervisor
4009	Operations Training Specialist	3482	Roofing Technical Supervisor
4591	Outdoor Education Center	4889	SAP Workflow Analyst
	Manager	4325	Satellite Food Service Manager
5307	Outdoor Education Programs	2500	School Administrative Assistant
	Coordinator		
<u>5371</u>	Out-of-School Regional Director	1775	School Traffic Safety Coordinator
<u>5303</u>	Out-of-School Senior Director	1100	Senior Accounting Analyst
3147	Painting Technical Supervisor	5070	Senior Administrative Analyst
3114	Paving Supervisor	2062	Senior Administrative Assistant
1338	Payroll Distribution Supervisor	2544	Senior Budget Technician
4137	Pest Management Supervisor	1472	Senior Building Project Estimator
4054	Plant Manager I	5721	Senior Bus Dispatcher
4047	Plant Manager II	2192	Senior Community Outreach Organizer
4037	Plant Manager III	1147	Senior Data Center Equipment
4021	Plant Manager IV	1147	Operator
3152	Plumbing Technical Supervisor	2615	Senior Early Childhood Education
1121	Principal Accountant	2015	Division Fiscal Technician
5101	Principal Administrative Analyst	1665	Senior Electrical Inspector
2716	Principal Assignment Technician	5033	Senior Employment Services Manager
1220	Principal Auditor	1773	Senior Energy Specialist
2551	Principal Clerk	1780	Senior Environmental Health
1116	Principal Financial Analyst		and Safety Officer
4118	Principal Gardener	1114	Senior Financial Analyst
2251	Principal Personnel Clerk	1128	Senior Fiscal Specialist
2141	Principal Realty Agent	4321	Senior Food Service Training Specialist
5288	Principal Stock Clerk	4174	Senior Gardener
5199	Procurement Customer Services	4998	Senior Human Resources Specialist
	Supervisor	3860	Senior IT Customer Services
<u>1425</u>	Program and Project Controls		Support Representative
	Manager	1365	Senior Labor Compliance Officer

2182	Senior Marketing	2576	Supervising Clerk
	Representative	2575	Supervising Clerk (Non-Typing)
2383	Senior Medical Administrative	1631	Supervising Building/Construction
	Assistant		Inspector
4309	Senior Nutrition Specialist	4818	Supervising Data Center <u>Technician</u>
1686	Senior Painting Inspector	1704	Supervising Electrical Engineer
4143	Senior Pest Management	1664	Supervising Electrical Inspector
	Technician	2773	Supervising Insurance Technician
5779	Senior Power Equipment	<u>2367</u>	Supervising Job Order Contracting
	Mechanic		Specialist
<u>5198</u>	Senior Procurement Customer	3778	Supervising Maintenance Worker
	Services Supervisor	1754	Supervising Mechanical Engineer
3809	Senior Programmer Analyst,	2211	Supervising Microcomputer Support
	(COBOL)		Assistant
3819	Senior Programmer Analyst, Filenet	1303	Supervising Payroll Technician
3807	Senior Programmer Analyst,	2261	Supervising Personnel Clerk
	(JAVA)	1675	Supervising Plumbing Inspector
3810	Senior Programmer Analyst,	5783	Supervising Power Equipment
	(Oracle)		Mechanic
4892	Senior Programmer Analyst (SAP)	5144	Supervising Purchasing Services
3811	Senior Programmer Analyst		Coordinator
	(Visual Basic)	4736	Supervising Records Technician
4436	Senior Radiologic Technologist	4560	Supervising Special Education
2144	Senior Realty Agent		Assistant
1806	Senior Sustainability Specialist	5395	Supervising Staff Development
3336	Senior Tractor and Roller		Coordinator (LA's BEST)
	Operator	1832	Supervising Structural Engineer
4131	Senior Tree Surgeon	1584	Supervisor, Plots Plans and Allotments
5164	Senior Stores Supervisor	1864	Survey Party Chief
4045	Senior Window/Wall Washer	3155	Swimming Pool Technical Supervisor
<u>1126</u>	Special Education Fiscal Analyst	1053	Telecommunications Billing
5175	Stores Supervisor		Supervisor
<u>4974</u>	Strategic Data Analyst	2952	Test Services Coordinator
1836	Structural Engineer	5099	Testing Operations Manager
<u>4736</u>	Student Records and Data Supervisor	1120	Transportation Cost Analyst
2356	Student Support Services	5741	Transportation Inspector
	Program Coordinator	5708	Transportation Planner
4369	Summer Food Service Supervisor	5728	Transportation Routing Center
1129	Supervising Accountant		Supervisor
1316	Supervising Accounting Technician	4127	Tree Maintenance Supervisor
1440	Supervising Architect	5736	Truck Dispatcher
2720	Supervising Assignment Technician	1180	Web Architect
5758	Supervising Automotive Body	2047	Workers' Compensation Claims
	Mechanic		Processing Supervisor
5214	Supervising Fleet-Parts Storekeeper		
1527	Supervising Civil Engineer		
5648	Supervising Classified Training		
	Representative		

- 1.1 <u>Excluded</u>: All other personnel designated as management or confidential within the meaning of Government Code Section 3540.1 and those classes and positions excluded in the Certification of Representation dated July 1, 1996, in P.E.R.B. Case No. 1009.
- 2.0 <u>Changes to the Unit</u>: The parties agree that this represents the appropriate unit. The unit may be revised only by mutual agreement or by a Public Employment Relations Board unit clarification decision, but it is agreed that neither party may file for a unit clarification proceeding involving this unit except when the District creates new classifications or when the Union contends that certain classifications should be accreted to the unit. Disputes over unit composition and alleged violations of this Article are not subject to the grievance and arbitration procedures of this Agreement.
- 3.0 <u>"Employees" Defined</u>: Unless the context clearly indicates otherwise, the terms "employee" or "employees" will normally be used in this Agreement to indicate persons who are included within the unit, and the term "Personnel" will normally be used in a broader sense to include employees as defined above plus all other persons utilized by the District to provide services.
- 4.0 Employees with more than one (1) job assignment who function for a majority of the work period in any of the classifications listed in Section 1.0 shall be considered in the unit. Should an employee's job involve an equal number of hours in different assignments, the employee shall be considered as included in the unit only if that employee has functioned in one of the foregoing classifications for the longest period of time based upon the date of regular assignment.

ARTICLE II

SEPARABILITY AND SAVINGS

- 1.0 If any provision of this Agreement should be held invalid by operation of law or by any tribunal of competent jurisdiction, or if compliance with or enforcement of any provision should be restrained by any tribunal of competent jurisdiction pending a final determination as to its validity, the remainder of this Agreement or the application of such Article or Section as to persons or circumstances other than those as to which it has been held invalid or as to which compliance with or enforcement of has been restrained, shall not be affected thereby.
- 2.0 In the event of any such invalidation of any provision of this Agreement, the parties agree to meet and negotiate within thirty (30) days of such invalidation for the purpose of arriving at a satisfactory replacement for such provision. This Article shall not be subject to the grievance and arbitration provisions of Article V (Grievance Procedure).

ARTICLE III

DISTRICT RIGHTS

- 1.0 <u>General</u>: The intention of this Article is to provide that the District retains all rights and powers which have not been limited by the other Articles of this Agreement. The provisions of this Article are not intended to expand the rights of the District beyond statutory and constitutional limits, or in any manner to waive or diminish the rights of the Union or the employees as provided in the other Articles of this Agreement. In the event that there is a conflict between the rights of the District under this Article and the rights of the Union or employees as set forth elsewhere in this Agreement, the provisions of the other Articles of this Agreement shall prevail.
- 2.0 <u>District Rights</u>: It is agreed that all matters which are beyond the scope of negotiations under Government Code Section 3543.2, and also all rights which are not expressly limited by the terms of this Agreement are retained by the District. Such retained rights include, but are not limited to, the right to determine the following matters:
 - a. The legal, operational, geographical, and organizational structure of the District, including the chain of command, division of authority, organizational divisions and subdivisions, external and internal boundaries of all kinds, and advisory commissions and committees;
 - b. The financial structure of the District, including all sources and amounts of financial support, income, funding, taxes and debt, and all means and conditions necessary or incidental to the securing of same, including compliance with any qualifications or requirements imposed by law or by funding sources as a condition of receiving funds; all investment policies and practices; all budgetary matters and procedures, including the budget calendar, the budget formation process, accounting methods, fiscal and budget control policies and procedures, and all budgetary allocations, reserves, and expenditures, apart from those allocated to fund the express wage and benefit obligations of this Agreement;
 - c. The acquisition, disposition, number, location, types and utilization of all District properties and equipment, whether owned, leased, or otherwise controlled, including all facilities, grounds, parking areas and other improvements, and the type of personnel, work, services, and activity functions assigned to such properties;
 - d. All services to be rendered to the public and to District personnel in support of the services rendered to the public, the nature, methods, quality, quantity, frequency and standards of service, and the personnel, facilities, vendors, supplies, materials, vehicles, equipment and tools to be used in connection with such services; the subcontracting of services to be rendered and functions to be performed, including educational, support, construction, maintenance and repair services, subject only to Code restrictions upon same;

Article III - District Rights

- e. The utilization of personnel not covered by this Agreement, including but not limited to consultants and personnel occupying positions listed as "Excluded" in Article I (Recognition), to do work on a non-regular and limited basis which is normally done by employees covered hereby, subject to provisions of the Education Code covering this issue, and the methods of selection and assignment of such personnel;
- f. The educational policies, procedures, objectives, goals and programs, including those relating to student conduct and discipline, student transportation, food services, racial and ethnic balance, extra-curricular activities, and emergency situations; and the substantive and procedural rights and obligations of students, parents, employees and the public with respect to such matters;
- g. The selection, classification, direction, promotion, demotion, discipline, termination and retirement of all personnel of the District subject only to applicable law; affirmative action and equal employment policies and programs to improve the District's utilization of women and minorities; the assignment of employees to any locations subject only to Article XI (Transfer Procedures); and also to any facilities, classrooms, functions, activities, departments, tasks, or equipment; the staffing levels, work loads, and the number of employees; and the determination as to whether, when and where there is a job opening.
- h. The job classifications and the content and qualifications thereof; the rates of pay for any new classifications implemented during the term of the Agreement. In case of a dispute regarding this sub-paragraph, the parties retain the rights held prior to the execution of this agreement to petition to PERB.
- i. The duties and standards of performance for all employees; and whether any employee adequately performs such duties and meets such standards, subject only to Article X (Evaluation Procedures).
- j. The dates, times, and hours of operation of District facilities, functions, and activities; work schedules; school calendar; the assignment of paid duty days beyond the regular assigned duty year; the assignment of overtime, if any, subject only to Article IX (Hours of Work) and Article XVI (Holidays).
- k. Safety and security measures for employees, students, the public, properties, facilities, vehicles, materials, supplies, and equipment, including the various rules and duties for all personnel with respect to such matters, subject only to Article XVIII (Safety Conditions).
- I. The rules, regulations and policies for all employees, students, and the public, subject only to the express limitations contained in this Agreement.

8

m. It is understood that several of the above-mentioned reserved rights are exercised in conjunction with or subject to Personnel Commission powers,

Article III - District Rights

functions and obligations, and where that occurs the above-mentioned rights of the District are intended to include the rights of the Commission.

- n. All other rights of the District not expressly limited by the language of this Agreement are also expressly reserved to the District even though not enumerated above, and the express provisions of this Agreement constitute the only contractual limitations upon the District's rights. The exercise of any right reserved to the District in a particular manner or the non-exercise of any such right shall not be deemed a waiver of the District's right or preclude the District from exercising the right in a different manner.
- 3.0 The right to "determine" as used above in Section 2.0 includes the exclusive right to establish, change, modify, or discontinue in whole or in part, temporarily or permanently, any of the above matters.
- 4.0 <u>Effect on Grievance Procedure</u>: The contractual rights of the Union and the employees are set forth in the other Articles of this Agreement and this Article is not a source of such rights. Accordingly, no grievances may be filed under this Article. However, nothing in this Article shall prevent the filing of grievances under Articles of this Agreement which have not been excluded from the grievance procedure.

ARTICLE IV

UNION RIGHTS

- 1.0 Access: Any authorized Union representative shall have the right of reasonable access to District facilities, including employee mailboxes for the purpose of contacting employees and transacting matters. Upon arriving at a work site, the representative shall first report to the office of the site administrator and state the intended purpose and estimated length of visit. The representative may contact employees during duty free lunch periods, before and after employees' hours of service, or when the employee is not engaged in duties. The representative shall not interrupt any employee's duties or assignments.
- 2.0 <u>Bulletin Boards</u>: The Union shall have the right to post notices of official Union matters on a designated bulletin board or a section of a designated bulletin board established for the Union's exclusive use at each work site where employees are assigned.
- 3.0 <u>Released Time for Negotiations</u>: No more than six (6) negotiating team employee representatives designated by the Union shall be released from duty with no loss of pay and with mileage reimbursement for the purpose of attending negotiation meetings with the District pursuant to this Agreement. The Union and the District may agree that additional employees shall receive such released time.
- 4.0 <u>List of Employees</u>: The Union shall be provided on a quarterly basis a current list of names, employee numbers, social security numbers, classifications, addresses, District e-mail addresses and work locations of all employees covered by this Agreement. This report shall also include date of hire in classification for each employee.
- 5.0 <u>Copies of Agreement</u>: An electronic copy of this Agreement will be provided by the District to the Union for its ratification meeting.
- 6.0 <u>School-Based Management</u>: No decision by a site council or any other District-sponsored committee to grant a School-based Management waiver request which conflicts with the express terms of this Agreement shall be approved without the consent of the Union.
- 7.0 <u>Special Committees</u>: From time to time, the District and the Union may establish ad hoc joint management -- bargaining unit committees to discuss matters of mutual concern. Such committees shall in no way be construed as bargaining committees nor shall they be empowered to modify, in any way, the terms and conditions of this Agreement.
- 8.0 <u>Teamsters Local 572 Elected Officers</u>: Where attendance at Union events by Teamsters Local 572 elected officers is required, reasonable released time to attend such events, not to exceed a total of five (5) days per fiscal year, will, subject to the provisions herein, be given to any Unit S member who is a Teamsters Local 572 elected officer. The Union will reimburse the District for the cost of the attendee's salary and benefits, and all

Article IV - Union Rights

expenses of attending will be borne by the Union or the employee. If more than one (1) Unit S member is a Teamsters Local 572 elected officer, no more than one (1) Unit S member per branch or local district will be given released time at any time, and in any event no more than three (3) such Unit S members will be given release time District-wide in any fiscal year. The Union will give the Office of Staff Relations five (5) working days advance notice of the name of the employee designated to attend any event and the duration of the released time requested. The supervisor and the employee will, where practicable, modify the employee's work schedule so as to minimize the amount of released time required for attendance.

- 9.0 <u>Attendance at District Meetings</u>: Subject to the provisions herein, employees designated by the Union may attend District meetings as follows:
- 9.1 When the Union is invited by the District to send an employee to participate in a District-sponsored meeting, one (1) designated employee will be given reasonable released time to attend.
- 9.2 If the Union desires that a designated employee attend any other District meetings (other than negotiations) such as Board meetings, Board Committee meetings, or Personnel Commission meetings, the Union may request reasonable released time approval for one (1) such designated employee to attend. Reasonable released time approval will be granted where the designated employee is a speaker on the meeting agenda. No employee will be granted released time more frequently than two (2) times per fiscal year. Expenses of attending any of the foregoing meetings will be borne by the Union or the employee, and, in any event, the number of released time approvals will not exceed six (6) in a fiscal year. It is provided, however, that if the agenda item on which the designated employee is scheduled to speak is not discussed at the meeting the employee attends, the employee's attendance will not count toward the two (2) times or six (6) times limitations noted above. The Union will reimburse the District for the salary and benefits of any employee approved for released time.
- 9.3 The Union will give the Office of Labor Relations three (3) working days advance notice of the name of the employee designated to attend any meeting, or as much advance notice as the Union has of the meeting. The supervisor and the employee will, where practicable, modify the employee's work schedule so as to minimize the amount of released time required for attendance. The parties understand that the immediate needs of a particular work group or area may, at any given time, prevent the release of a particular employee, in which case the Union will be advised and may designate another employee to attend. If the District reasonably determines that the operational needs of a particular work group or area are such that no designated employee can be released to attend, no released time will be granted. Employees are expected to return to work after attending the meeting, if practicable.

ARTICLE V

GRIEVANCE PROCEDURE

- 1.0 <u>Grievance and Parties Defined</u>: A grievance is defined as a claim that the District has violated an express term of this Agreement and that by reason of such violation the grievant's rights under this Agreement have been adversely affected. Grievances as so defined may be filed by:
 - a. An employee;
 - b. The Union on behalf of an identified employee(s); or
 - c. The Union on its own behalf as to alleged violations of rights granted to the Union in this Agreement.
- 1.1 All other matters and disputes of any nature are beyond the scope of this grievance procedure, including but not limited to those matters for which other methods of adjustment are provided by the District, such as reductions in force; examination procedures, results and references; performance evaluations; disciplinary matters; and complaints by one employee about another. Also excluded from this grievance procedure are those matters so indicated elsewhere in this Agreement. Claimed violations of Article VII (Non-Discrimination) are to be handled through the Equal Opportunity Program Section or the grievance procedure subject to Article V (Grievance Procedure).
- 1.2 The respondent in any grievance shall be the District itself rather than any individual administrator.
- 1.3 If the same grievance or essentially the same grievance is filed by more than one (1) employee, then, by written agreement with the Union, the District may process any or all such grievances as one (1) grievance. In such event, the final determination of the grievance shall apply to all grievances.
- 1.4 The filing or pendency of a grievance shall not delay or interfere with implementation of any District action during the processing thereof unless the parties agree to the contrary.
- 1.5 Processing and discussing the merits of a grievance shall not be considered a waiver by the District of the defense that the matter is neither grievable nor subject to arbitration under this Agreement or that the grievance should be denied for other reasons which do not go to the merits.
- 2.0 <u>Representation Rights in the Grievance Procedure</u>: If an administrator with the authority and responsibility to adjust a grievance is absent during the time specified for meeting his/her responsibility under these procedures and no

mutual agreement has been reached for a time extension, the District shall designate a representative to assume this responsibility. The grievant must be present at each step of the grievance procedures unless excused by the District.

- 2.1 At all grievance meetings under this Article, the grievant shall be entitled to be accompanied and/or represented by either a job steward or a Union representative. A grievant shall also be entitled to represent himself or herself. The administrator shall have the right to be accompanied by another administrator or District representative. By mutual agreement other persons such as witnesses may also attend grievance meetings.
- 2.2 When a grievant is not represented by the Union, the District shall not agree to a final resolution of the grievance until the Union has received a copy of the grievance, been notified of the proposed resolution and been given an opportunity to state in writing its views on the matter, provided however, that the grievance may be withdrawn by the grievant at anytime which shall terminate the grievance procedure.
- 3.0 Released Time for Employees: Grievance meetings and hearings will be scheduled by the District at mutually convenient times and places during regular District business hours. Such meetings will be scheduled so as to minimize interference with regular employee duties. If a grievance meeting or hearing is scheduled during duty hours, reasonable employee released time, including necessary travel time without loss of salary, will be provided to the grievant, Job Steward and to any witness who attends by mutual agreement. When grievance meetings are scheduled outside the employee's duty hours, the meeting shall be held immediately preceding or following the employee's shift, except that grievance meetings scheduled following completion of an employee's shift may be held at a time to allow the employee adequate time to travel to the meeting on non-duty hours. Any of the foregoing employees required to travel to meetings or hearings pursuant to this section shall receive mileage reimbursement.
- 4.0 <u>Confidentiality</u>: From the time a grievance is filed until it is finally resolved, neither the Union, nor the grievant, nor the District shall publicly disclose or discuss the grievance or evidence regarding the grievance (e.g., specific facts, positions of the parties, merits, etc.). This prohibition is not intended to restrict normal interviewing of witnesses and other necessary preparations for the hearing or internal communication by the Union or the District for the purpose of evaluating, pursuing or resolving grievances. Moreover, nothing in this provision shall prohibit the internal disclosure by either the District or the Union of the general fact that a grievance has been filed regarding a particular contractual dispute and that the parties are utilizing the grievance process in an attempt to resolve that dispute.
- 5.0 <u>Effect of Time Limits</u>: If a grievance is not processed by the grievant at any step in accordance with the time limits of this Article, it shall be deemed withdrawn. If the District fails to respond to the grievance in a timely manner at any step, the running

of its time limit shall be deemed a denial of the grievance and termination of the step in question, and the grievant may proceed to the next step. All time limits and grievance steps may be shortened, extended or waived, but only by mutual written agreement.

- 6.0 <u>"Day" Defined</u>: A "day" for purposes of this Article is defined as any day of the calendar year except Saturdays, Sundays, and legal (or school) holidays.
- 7.0 <u>Required Informal Discussion</u>: Before filing a formal written grievance under Step One, a grievant must attempt to resolve the dispute by presenting the grievance orally to the immediate administrator and discussing the grievance with the administrator. The written grievance must be filed within the time limits required under Step One whether or not the grievant is able to utilize these informal efforts.
- 8.0 <u>Step One</u>: Within fifteen (15) days, as defined in Section 6.0, after the grievant or the Union knew or reasonably should have known of the occurrence of the facts upon which the grievance is based, the grievance must be presented in writing to the immediate administrator on the District Grievance Procedure Form stating the facts surrounding the grievance, identifying the specific provisions of this Agreement which are alleged to have been violated and the remedy requested. The form shall be signed and dated by the grievant. A meeting between the grievant and the immediate administrator shall take place within five (5) days from presentation of the grievance, and the administrator shall reply in writing within five (5) days following the meeting. Unless there is a mutual written agreement to the contrary, Step One shall terminate at the close of business on the ninth (9th) day following the Step One meeting.
- 8.1 If a grievance does not relate to the immediate administrator and the remedy requested is not within the authority of the immediate administrator, the grievance may, if the grievant desires, be filed with the administrator who has such responsibility and authority.
- 9.0 <u>Step Two</u>: If the grievance is not resolved in Step One, the grievant may, within five (5) days after the termination of Step One, present the written grievance to the grievant's Division Head or Local District Superintendent, or designee. Within five (5) days from receipt of the grievance, a meeting shall take place to discuss the matter and the administrator shall reply in writing within five (5) days following the meeting. Unless there is a mutual written agreement to the contrary, Step Two shall terminate at the close of business on the ninth (9th) day following the Step Two meeting.
- 10.0 <u>Step Three</u>: If the grievance is not resolved in Step Two, the grievant may, within five (5) days after the termination of Step Two, present the written grievance to the Deputy Superintendent or designee. If, at his or her discretion, the Deputy Superintendent or designee desires, a meeting may take place within five (5) days from receipt of the grievance. The Deputy Superintendent or designee shall reply in writing to the grievance within five (5) days after the meeting or, if no meeting is held, within five (5) days after receipt of the grievance. Unless there is a mutually written agreement to the

contrary, Step Three shall terminate at the close of business on the ninth (9th) day following the Step Three meeting, or if no meeting is held, on the fourteenth (14th) day following receipt of the grievance.

11.0 <u>Request for Board Review</u>: If the Union is not satisfied with the decision at Step Three, the Union, with the concurrence of the grievant, may submit the matter to the Office of Labor Relations for a Board of Review. This request must be made in writing within five (5) days after the termination of Step Three.

Within five (5) days from the date the request for a Board of Review is received by the Office of Labor Relations, a meeting shall be arranged with the parties to the grievance, or their representatives, for the selection of the chairman of the Board of Review. The chairman may be jointly agreed upon by the parties or shall be selected from the following list of persons by alternately striking names until one remains.

Howard Block George Marshall
Bonnie Castrey Kenneth Perea
Joe Henderson Michael Prihar

Edgar Jones

The party who strikes the first name shall be determined by lot. If the chairman indicates that he/she will not be available for hearing within a reasonable time, not to exceed sixty (60) days; the parties shall proceed to select another chairman as indicated above. The District and the Union shall each pay one-half (½) of the fees of the chairman of the Board of Review. Each party shall bear the expense of the presentation of its own case. All decisions and rulings shall be made by majority decision of the Board of Review.

- 12.0 <u>Board of Review</u>: The Board of Review shall be composed of a chairman and two (2) members. The Union and the District shall each appoint one (1) member, who shall be an employee or an administrator of the District, to serve on the Board of Review, provided, however, that no employee may be selected by any grievant to serve on a Board of Review more than twice in any calendar year.
- 12.1 The hearing shall be under the direction of the Chairman who shall conduct all matters in accordance with the rules and procedures prescribed in Section 11513 of the Government Code except as otherwise indicated in this Article. Sessions of the Board of Review shall be private with attendance limited to the parties to the grievance and their representatives, if any, and witnesses while testifying.
- 12.2 The Office of Labor Relations shall be responsible for the arrangements for the hearing, the maintenance of records, and such other services required by the chairman for the fulfillment of the chairman's responsibilities.
- 12.3 The parties shall exchange lists of proposed witnesses not later than five (5) days prior to the first (1^{st}) date of the hearing.

- 12.4 Neither party shall communicate with the Chairman without first contacting the other party to explain the purpose of the intended communication.
- 12.5 Unless the parties mutually agree otherwise, a hearing shall be scheduled within sixty (60) days from selection of the arbitrator, but shall not be scheduled during the summer or off-track time. The decision shall be issued within thirty (30) calendar days after final submission of the case. Arbitrators who fail to meet the deadline for decision shall, unless the parties have mutually extended this deadline, be deemed ineligible for selection for new cases until such time as the decision is submitted.
- 13.0 Optional Preliminary Hearing on Issues Which Do Not Involve Merits of Grievance: If the District claims that the grievance should be dismissed for reasons which do not go to the merits (e.g., mootness, timeliness, matter beyond the scope of procedure, or breach of confidentiality provisions) the District may cause its claim to be heard and ruled upon by the Board of Review prior to a hearing on the merits. If the District plans to invoke this separate preliminary hearing, it shall so advise the Union prior to the selection of the Chairman. Immediately after selection of the Chairman for the preliminary hearing, either the Union or the District may require that a different Chairman be selected to hear the merits in the event that such a hearing is required. There shall be at least fifteen (15) days between the Board of Review's decision on the preliminary matter(s) and any hearing on the merits. The preliminary hearing is optional to the District, and if not utilized, the District shall not be precluded from raising its arbitrability defenses at the regular hearing, provided that it gives the Union ten (10) days' notice of its intention to do so. Moreover, both the Union and the District shall retain all rights they have under law to pursue issues relating to arbitrability of a grievance.
- 14.0 <u>Limitations Upon the Board of Review</u>: The Board of Review shall have no power to alter, add to, or subtract from any of the terms of this Agreement or of this Article, but shall only determine whether an express term of the Agreement has been violated as alleged in the grievance. Past practice of the parties in interpreting and applying the terms of this Agreement may be relevant evidence, but shall not be used so as to justify or result in what is in effect a modification (whether by revision, addition or detraction) of the terms of this Agreement. The Arbitrator shall have no power to render an award on any grievance occurring before or after the term of this Agreement or to grant a remedy exceeding that sought by the grievant.
- 15.0 <u>Effect of Board of Review Award</u>: The Board of Review's decision shall be final and binding upon the grievant(s), the District and the Union. The California law on final and binding arbitration awards between a school district and an employee organization shall be applicable to such a decision.
- 15.1 Except as provided above, a final and binding award which determines the merits of a dispute shall be conclusive on the grievant(s), the District and

the Union in any subsequent proceedings, including disciplinary and termination proceedings.

- 15.2 Unless otherwise indicated in this Agreement, this grievance procedure is to be the employees' and the Union's sole and final remedy for any claimed breach of this Agreement.
- 16.0 <u>Expenses:</u> All fees and expenses of the Chairman shall be shared equally by the Union and the District. Each party shall bear the expense of presenting its own case. A transcript of proceedings shall not be required, but either party may order a transcript at its own expense. If the other party at any time desires a copy of the transcript, it must share equally the cost of the reporter and transcription.
- 16.1 <u>Rescheduling/Cancellation Expenses</u>: Should one (1) of the parties request the Board of Review be either rescheduled or cancelled, the requesting party shall bear the rescheduling/cancellation fee determined by the Chairman, should there be such a fee. Should the parties mutually agree to reschedule or cancel the Board of Review, the costs shall be borne equally by the parties.
- 17.0 <u>Grievance Files</u>: The District's Office of Staff Relations shall maintain a file of all grievance records and communications separate from the personnel files of the grievant(s), and grievance documents and decisions shall not be included in the personnel file unless it is reasonably necessary or appropriate to do so. Grievance documents, including arbitration awards, shall never be placed in an employee's examination folder except that any evaluation which is sustained through the grievance procedure may be placed in the employee's examination folder.
- 18.0 <u>No Reprisals</u>: There shall be no reprisals against an employee for utilizing these grievance procedures or for assisting a grievant pursuant to these procedures.

ARTICLE VI

WORK STOPPAGE

- 1.0 <u>No Strikes</u>: Apart from and in addition to any existing legal restrictions upon and remedies for work stoppage, the Union agrees to the following:
 - a. During the term of this Agreement, neither the Union nor its respective offices or representatives shall urge, call, sanction or engage in any work stoppage, slowdown, or other concerted interference with normal District operations for any cause whatsoever. In the event of any actual or threatened strike, slowdown, or other work stoppage, the Union and its officers, representatives and affiliates shall take all reasonable steps within their control to avert or end the same; and
 - b. Any employee engaging in any strike, slowdown, or other work stoppage may be subjected to discipline or termination under applicable law.
 - c. Neither the District nor the Union shall discriminate against a unit member for honoring or refusing to honor a picket line of another Teamster organization:
 - (1) The picket line has been sanctioned by Joint Council No. 42;
 - (2) The District and Union have previously conferred and agreed that unit members need not cross the particular picket line (with such agreement by the District not unreasonably withheld, e.g., such agreement may be withheld when there is no reasonable alternative source for the particular goods or services);
 - (3) In no case shall the rights provided to a unit member pursuant to this section be employed by the unit member for longer than two (2) hours in any one (1) day or in any case where to do so would compromise the health or safety of students or staff; and
 - (4) The District shall not compensate a unit member for time spent by the unit member exercising his or her rights under this section.
- 2.0 <u>No Lockouts</u>: The District agrees that it shall not engage in a lockout of unit members during the term of this Agreement. The term "lockout" is intended to cover a situation where the employer refuses to permit employees to work in an effort to obtain bargaining concessions from the Union.

Article VI – Work Stoppage

- 3.0 Disputes arising under this Article are to be handled according to appropriate legal proceedings rather than the grievance and/or arbitration procedures of Article V (Grievance Procedure).
- 4.0 No employee shall be required to perform clearly identifiable struck work of employees of a different bargaining unit, unless such work is generally included in the bargaining unit employee's job description. In no such case shall a bargaining unit member permanently replace another bargaining unit member on strike.

ARTICLE VII

NON-DISCRIMINATION

- 1.0 Pursuant to applicable Federal and State Laws, the District and the Union agree not to discriminate against any employee based upon race, color, creed, national origin, gender, age, physical disability, mental disability, medical condition, sexual orientation, marital status, <u>religious creed</u> or union affiliation.
- 2.0 Employees may grieve alleged violations of this Article through Steps I, II, and III of the grievance procedures of Article V (Grievance Procedure). Any such grievance may, at the Union's request, then proceed to arbitration pursuant to Article V (Grievance Procedure), Sections 11.0 through 16.0 upon execution of a separate written agreement by the individual grievant to be bound by the arbitration award as a final and binding resolution of the dispute. If the employee does not want to use the grievance procedure, the employee may process any claimed violations through the appropriate statutory procedures or through the District Equal Opportunity Program Section.

ARTICLE VIII

UNION SECURITY AND DUES DEDUCTION

- 1.0 <u>Voluntary Authorization:</u> The District shall deduct Union membership dues for each pay period worked in the amount specified by the Union from the salary of each employee who has submitted a written authorization.
- 2.0. <u>Exclusive to Union:</u> Payroll deductions for membership dues from employees shall be exclusive on behalf of the Union and no membership dues deductions are to be made on behalf of any other employee organization as defined in Government Code Section 3540.1(d).
- 3.0 <u>Remittance to Union:</u> A deposit approximating the amount of dues so deducted shall be remitted to the Union on payday, and the reconciled amount will be supplied to the Union within thirty (30) days after the deductions are made, together with a list of affected employees.
- 4.0 <u>Dues Deductions:</u> In instances where dues deduction is not taken from an employee who has a valid authorization form on file, the missed deductions(s) will be taken from subsequent salary payment and remitted to the Union.
- 5.0 A dues deduction may only be revoked by an employee in writing during the thirty (30) day period commencing ninety (90) days before the expiration of the Agreement and/or upon expiration of the agreement. The dues deduction shall automatically terminate if an employee terminates employment or otherwise ceases to be a member of the bargaining unit.
- 6.0 Agency Fee Obligation: Those employees who are currently members of the Union, and who have a dues deduction in effect, shall continue to have such dues deducted from their salary payments. Commencing within thirty (30) days of a final certification of agency fee election results or within thirty (30) days of an employee's initial employment, whichever is later, and continuing throughout the term of this Agreement, each employee (as defined in Article I [Recognition] of this Agreement) is required as a condition of continued employment either: (a) to be a member in good standing of the Union, or (b) to satisfy the agency fee financial obligations set forth in Section 6.1 below, unless qualified for religious exemption as set forth in Section 6.2 below.
- 6.1 Unless the employee has (a) voluntarily submitted to the District an effective dues deduction request, or (b) individually made direct financial arrangements satisfactory to the Union as evidenced by notice of same by the Union to the District, or (c) qualified for exemption based upon religious grounds as provided in Section 6.2 below, the District shall process a mandatory agency fee payroll deduction in the appropriate amount, and forward that amount to the Union. The amount of agency fee to be charged shall be determined by the Union subject to applicable law: it shall therefore be an amount not to exceed the normal periodic membership dues, initiation fee and general

Article VIII - Union Security and Dues Deduction

assessments applicable to Union members. As to non-members who object to the fee and general assessment applicable to the Union spending their agency fee on matters unrelated to collective bargaining and contract administration, the amount of agency fee charged shall not reflect expenditures which the courts or PERB have determined to be non-chargeable, including political contributions to candidates and parties, members-only benefits, charitable contributions and ideological expenditures and to the extent provided by law, shall not reflect expenditures for certain aspects of lobbying, ballot measures, publications, organizing and litigation. The Union shall comply with applicable law regarding disclosure and allocation of its expenses, notice to employees of their right to object, provision for agency fee payers to challenge the Union determinations of amounts chargeable to the objecting nonmembers, and appropriate escrow provisions to hold contested amounts while the challenges are underway. The foregoing description of permissible agency fee charges and related procedures is included herein for informational purposes as a statement of applicable law, and is not intended to change applicable law or to provide any contractual terms or enforcement procedures under this Agreement. The District will promptly remit to the Union all monies deducted, accompanied by a list of employees for whom such deductions have been made.

6.2 <u>Religious Exemption from Agency Fee Obligations:</u>

a. Any employee who is a member of a religious body whose traditional tenets or teachings include objections to joining or financially supporting employee organizations shall not be required to meet the above agency fee obligations, but shall pay in lieu thereof (by means of mandatory payroll deduction) an amount equal to the agency fee, to a non-religious, non-labor charitable organization exempt from taxation under Section

501(c)(3) of the Internal Revenue Code, as designated by the employee. Board of Education approved examples of such organizations are:

Brotherhood Crusade United Negro College Fund United Way

- b. To qualify for the religious exemptions, the employee must provide to the District, with a copy to the Union a written statement of objection, along with verifiable evidence of membership in a religious body as described in Section "a" above.
- c. An employee utilizing this religious exemption status who requests the Union to utilize the grievance/arbitration provisions on the employee's behalf, shall be subject to charges by the Union for the reasonable cost of using such procedures.
- 6.3 <u>Implementation Dates</u>: Any of the above-described payment obligations applicable to employees shall be processed by the District with the payroll

Article VIII - Union Security and Dues Deduction

immediately following the effective date of the payment requirement provided that the information is on file with the Payroll Branch by the deadline for filing time reports.

6.4 <u>Indemnity/Hold-Harmless</u>: The Union agrees to indemnify and hold the District harmless against any and all liabilities (including reasonable and necessary costs of litigation) arising from any and all claims, demands, suits, or other actions relating to the District's compliance or attempted compliance with either this Article or the requests of the Union pursuant to this Article, or relating to the conduct of the Union in administering this Article. The Union shall have the right to determine and decide all matters relating to settlement and conduct of litigation with respect to this Article. In no case shall District funds be involved in any remedy relating to this Article. Any underpayments to the Union resulting from the District's failure to make a required deduction shall be remedied by additional deductions from the affected employee(s).

Any overpayments to the Union resulting from excessive deductions shall be remedied either by refund from the Union to the affected employee(s) or by a credit against future payments by the affected employee(s).

- 6.5 The District will furnish any information needed by the Union to fulfill the provisions of this Article.
 - 6.6 The District shall deduct agency fees for twelve (12) pay periods.

ARTICLE IX

HOURS OF WORK

1.0 <u>General Provisions</u>:

- 1.1 The work year of employees shall be determined by the District in accordance with the Assignment Bases established in Board Rule 1990. The District reserves the right to add new bases or modify existing bases to meet the operational needs of the District. If, during the term of this agreement the Board of Education takes action to exempt another bargaining unit from Board Rule 1990's provisions regarding the placement of employees on unpaid service for up to five (5) days during the school year, then this unit shall be similarly exempt.
- 1.2 The workweek of employees shall normally be Monday through Friday, provided, however, that the District may establish a different workweek for particular employees, classes, or shifts as required to meet the operational needs of the District.
- 1.3 Full-time employment for employees shall be based on a forty (40) hour workweek of eight (8) hours per day, or a thirty-five (35) hour workweek of seven (7) hours per day, exclusive of meal periods. Employees' daily hours of work and shifts shall be established at the discretion of the District to meet the operational needs of the District in a manner consistent with applicable law, provided, however, that nothing contained herein shall be construed as a guarantee by the District of a certain number of paid hours per day or days per week.
- 1.4 Nothing contained herein precludes the District from establishing a ten (10) hours per day, forty (40) hours per week schedule for certain classifications or for employees within certain classifications, or any other similar flexible workweek, provided, however, that certain benefits such as holidays and bereavement leave shall be granted on a properly pro-rated basis so as not to advantage or disadvantage such employees in relation to other employees assigned the same number of hours per week. Such a schedule change, however, will not occur without the concurrence of the concerned employees as ascertained through the Union.
- 1.5 For the purpose of computing hours worked, time during which an employee is excused from work but remains in paid status, e.g. because of holidays, vacation, or paid leaves of absence, shall be considered as time worked by the employee.
- 1.6 Prior to any substantial change of a permanent nature that affects an employee's work week or group of employees' work week, daily hours of work, and/or work shift, the employee(s) involved shall be given reasonable advance notification. The Union shall be advised and provided an opportunity to consult with the District regarding such a substantial change for a group of employees. (However,

Article IX - Hours of Work

these consultation rights shall not operate to stop or stay the change during such consultation.)

- 2.0 Overtime: Overtime only applies to employees in this unit employed in a classification receiving overtime prior to the execution of this agreement, i.e., employees employed in classifications which are not designated by the Personnel Commission in Rule 596 as executive/administrative. Employees employed in classifications which are not entitled to receive overtime, i.e., employees employed in classifications designated as executive/ administrative in Personnel Commission Rule 596, shall earn vacation credit at the rate of .09232 hours of credit for each hour for which pay is received, in an amount not to exceed one hundred ninety two (192) hours [or twenty-four (24) days] per year. This accrual rate shall supersede (not be added to) the accrual rates set forth in Article XVII, (Vacation) Section 1.1 for those employees employed in such overtime exempt classifications. Overtime must be approved in advance by the employee's direct supervisor or designee and (except in exceptional and unusual circumstances) confirmed in writing by the appropriate administrator or designee. An employee who works overtime without such approval shall not be compensated for such work. Overtime eligible employees shall not be disciplined for requesting overtime approval prior to working overtime or for requesting payment for pre-approved overtime.
- 2.1 To the extent practicable, the District shall use reasonable efforts to distribute overtime work equitably among the qualified employees of an office, operational unit, or work group with consideration given to District need and employee availability in making the distribution. Upon reasonable notice, an employee shall be required to work overtime as needed. Reasonable notice shall be deemed to be no less than twelve (12) hours in advance except in cases of emergency or when necessary to meet unanticipated peak work loads.
- 2.2 Employees assigned to a workday of seven (7) hours or more and a workweek of thirty-five (35) hours or more shall receive compensation at a rate equal to one and one-half (1-1/2) times the regular rate of pay, or shall be provided compensatory time off in a manner consistent with applicable State and Federal laws at the rate of one and one-half (1-1/2) times the hours worked, for work authorized and performed on the sixth (6th) and seventh (7th) days following the commencement of the regular workweek, or for hours worked in excess of eight (8) hours in one (1) day or in excess of forty (40) hours in any calendar week. Earned compensatory time off shall be scheduled in a manner consistent with State and Federal Laws.
- 2.3 Employees assigned an average workday of four (4) hours or more but less than seven (7) hours and a workweek of twenty (20) hours or more but less than thirty-five (35) hours shall be compensated at a rate equal to one and one-half (1-1/2) times the regular rate of pay, or shall be provided compensatory time off in a manner consistent with applicable State and Federal Laws at the rate of one and one-half (1-1/2) times the hours worked, for any work authorized and performed on the sixth (6th) and seventh (7th) days following the commencement of the regular

Article IX - Hours of Work

workweek, or for hours worked in excess of eight (8) hours in one (1) day or hours worked in excess of forty (40) hours in a calendar week. Earned compensatory time off shall be scheduled in a manner consistent with State and Federal Laws.

- 2.4 Employees assigned an average workday of less than four (4) hours shall be compensated at a rate equal to one and one-half (1-1/2) times the regular rate of pay, or shall be provided compensatory time off of one and one-half (1-1/2) times the hours worked, for any work authorized and performed on the seventh (7th) day following the commencement of the regular workweek, or for hours worked in excess of eight (8) hours in one (1) day or hours worked in excess of forty (40) hours in a calendar week.
- 2.5 When an employee is authorized and required to work on any day recognized as a holiday under this Agreement, he/she shall be compensated, in addition to regular pay received for the holiday, at the rate of one and one-half (1-1/2) times the regular rate of pay for actual hours worked.
- 2.6 For any compensatory time-off earned pursuant to Sections 2.2, 2.3, and 2.4 of this Article, the District shall, within twelve (12) months from the date it is earned, or earlier if required by law, either pay the employee or allow time off at the appropriate rate.
- 3.0 <u>Call Back Time</u>: Employees who are called back to work outside of their regular work hours shall be guaranteed a minimum of two (2) hours pay or compensatory time off at the appropriate rate in a manner consistent with applicable State and Federal Law. However, this section shall apply only to employees in this unit in a classification entitled to overtime pursuant to section 2.0 above.
- 4.0 <u>Unpaid, Duty-Free Time:</u> Employees who are overtime eligible and who have an on-site obligation which includes unpaid time each day shall be entitled to duty free time during that unpaid period. This unpaid time shall be scheduled in consultation with the direct supervisor and, except in unusual and infrequent circumstances pre-approved by the direct supervisor, the unpaid time shall not be taken during the first (1st) or last hour of the assignment. Employees who are prevented from taking this unpaid time and who are required to perform work duties during that time may complete this unpaid time later that same day (scheduled in accordance with the direct supervisor). All on-duty time shall be counted as time worked and be compensated in accordance with this Article and State and Federal Laws. If, due to unusual circumstances and after pre-approval by the employee's direct supervisor, the unpaid time is scheduled immediately prior to the end of the employee's work day, then the employee shall be free to leave the work site without penalty.

Article IX – Hours of Work

- 4.1 The duty-free unpaid period shall not be extended for the purpose of lengthening the employee's work day without incurring overtime, nor shall the employee be required to take a duty-free period longer than one (1) hour. It is provided, however, that as to Assistant Area Bus Supervisors, the District may for budgetary considerations utilize split shifts. Prior to doing so, the District will advise the union and provide the union an opportunity to consult. However, such consultation will not delay implementation.
- 5.0 Employees shall be compensated for all hours worked in accordance with this Agreement and applicable law, including work performed during otherwise unpaid, duty-free meal periods and work performed before and after assigned shifts, including work related, District initiated telephone calls made to employees before and after assigned shifts. Any calls compensable under this section shall be paid on the basis of the actual time spent on the call.

ARTICLE X

EVALUATION PROCEDURES

- 1.0 <u>Schedule</u>: Employees shall be evaluated by administrators in accordance with the following schedule: The District agrees to send a reminder to all administrators of bargaining unit employees immediately before the time that annual evaluations are to be prepared stating that failure to follow provisions outlined in this Article can result in a grievance or appeal filed by the employee. A goal of the employee evaluation is to help each employee perform his/her job more effectively to the mutual benefit of the employee and the District. The District and the Union agree that a primary purpose for evaluations is: a) to provide the employee with information regarding the status of his/her performance and quality of work; and b) to provide the immediate administrator with current reports and permanent records on the performance of the employee. An evaluation shall not be used as discipline in and of itself. However an evaluation may be used as evidence that an employee was provided notice of performance deficiencies.
 - a. Probationary employees shall be given performance evaluations no less than twice during their probationary period. However, if during the probationary period any items on the evaluation form are rated unsatisfactory, then the employee may be evaluated every month during the remainder of the probationary period.
 - b. Permanent employees shall be given performance evaluations at least once every year.
 - c. The District shall make a reasonable effort to issue the employee's annual evaluation at least twenty (20) working days prior to the end of the employee's assignment basis for that school year. The parties realize that because of year-round schools and different work schedules, employees' assignment bases may end at different times.

Evaluations may be issued within the last twenty (20) days or in the next evaluation period in cases where the employee is unavailable. If an employee receives the annual evaluation within the last twenty (20) working days, the further processing of any timely appeal or timely grievance shall be postponed, unless otherwise requested by the employee or the Union, until the employee's assignment basis begins again.

In an effort to create a positive evaluation process for everyone involved, the District has formed the Classified Performance Framework Advisory Committee to discuss plans and ideas to incorporate into the employee evaluation process. Once the Advisory Committee concludes its discussions, the parties agree that any items within the scope of bargaining that affect unit members will be negotiated as they arise, outside of the normal bargaining timelines.

- 2.0 <u>Procedure to be Followed</u>: Performance evaluation reports, including annual evaluations and any interim evaluations, shall be made on forms prescribed by the District. At the discretion of the evaluator and the employee, a meeting may be convened early in the evaluation period to discuss performance goals for the upcoming evaluation period. Such meeting should, when possible, occur within the first thirty (30) days of the evaluation period. Thereafter during the evaluation period, the evaluator and employee may agree to modify or change any performance goals originally set.
 - a. Evaluations shall be based upon observation or knowledge and not upon unsubstantiated charges or rumors. No evaluation shall be based upon derogatory materials in the employee's personnel file unless the employee has previously been given sufficient prior notice of same, an opportunity to review and comment upon them, and had such comments attached to the materials. For purposes of this Article, "Personnel File" refers to an employee's specific personnel file maintained by the District in the Classified Assignments Branch, as well as any such personnel file maintained by the appropriate division, branch or school site.
 - b. The evaluator (generally the immediate administrator) shall discuss the written performance evaluation report with the employee. However, in the event the parties are unable to meet, the administrator shall arrange for a discussion at a later date. Both the evaluator and the employee will sign the evaluation. The signature of the employee means only that the employee has received a copy of the evaluation. The employee may attach any written comments to the evaluation at the employee's option at the time of the conference or at a later date. The original evaluation shall be placed in the appropriate division, branch or school site file. The employee shall have the right to inspect the original. Upon request, each employee shall be informed as to the file in which the original performance evaluation is kept. Copies of the evaluation together with any attachments will be distributed as follows: one (1) copy to the evaluator.
 - c. If any category on the performance report is rated lower that "meets standards" the following will be included in the evaluation:
 - (1) statement of the problem or concern;
 - (2) the desired improvement;
 - (3) suggestions as to how to improve; and
 - (4) provisions for assisting the employee.
 - d. If on an annual evaluation, an evaluator rates an employee's overall performance as lower than "meets standards", the evaluator shall note on the evaluation whether or not the employee has been previously advised of the specific deficiencies which form the basis for that lower than "meets

standards" rating. If the employee was not previously advised so, then the evaluator shall note why not.

- e. Upon request by an employee to the Personnel Commission, a performance evaluation form shall be forwarded to, and completed by, any administrator under whom the employee has worked during the evaluation period for at least ninety (90) calendar days.
- 3.0 Appeal: If the employee disagrees with the evaluation, he/she shall have the right to appeal the evaluation in writing to the appropriate Local District Superintendent, or division head, or designated representative within ten (10) working days of receipt of the evaluation. A meeting may take place within five (5) working days from receipt of the appeal if the employee so requests. The employee may be represented in this meeting by the Union if the employee so desires. The reviewer shall reply in writing to the employee within ten (10) working days after the meeting, or if no meeting is held, within ten (10) working days after receipt of the written appeal. The decision of the reviewer shall be final.
- 3.1 No grievance arising under this Article shall challenge the substantive objectives, standards, or criteria determined by the evaluator or the District, nor shall it contest the judgment of the evaluator. Grievances concerning evaluations filed under Article V (Grievance Procedure) shall be limited to a claim that the procedures of this Article have not been followed.
- 4.0 <u>Notice of Unsatisfactory Service or Act</u>: An employee given a Notice of Unsatisfactory Service or Act that does not recommend disciplinary action shall have the same appeal rights as outlined in Section 3.0, above. A formal grievance concerning such a Notice of Unsatisfactory Service or Act filed under Article V (Grievance Procedure) shall be limited to a claim that the procedures in Section 2.0 above, have not been followed.
- 5.0 An employee shall be provided a copy of all adverse written material prior to or at the time they are placed in his/her Personnel File.
- 5.1 The employee shall have the right to sign or initial any such adverse material and prepare a written response which shall be attached to the material. Upon reasonable prior notice an employee shall have the right to inspect his/her personnel file (as defined in Section 2.0 (a) above) during the normal office hours of the office in which the personnel file is located without loss of pay. The employee's Union representative shall have the right, with the written consent of the employee, to inspect his/her personnel file. Employees will not be charged for the first five (5) pages of materials in the personnel file which they request. An employee shall also be entitled to a copy of any document the employee is requested to sign.

6.0 <u>Conference Memos</u>: The purpose of a conference memo is to inform the employee in writing about perceived deficiencies, where appropriate, to provide constructive assistance to the employee to improve, and to document the communication on a reasonably current basis.

A conference memo is a written record about work performance issues issued after a face-to-face meeting or a telephone discussion if the employee or supervisor is unable to personally meet.

When the District determines that a formal conference memo is to be issued, the following procedures shall apply:

- a. The conference memo will be signed by the issuing administrator. The employee shall have the right to sign the conference memo.
- b. A copy of the memo will be given to the employee. The administrator shall make reasonable efforts to obtain the signature of the employee acknowledging receipt of the memo or a witness that the conference memo was delivered.
- c. Any written response from the employee shall be attached to the memo and retained with the file copy.

Neither the District nor the employee shall consider the conference memo by itself to constitute discipline. Any use of conference memos in subsequent disciplinary proceedings shall be to establish that the employee had notice of the concerns expressed in the conference memo. The underlying facts which gave rise to the conference memo (e.g., absence or tardiness record or the employee conduct at issue) may be presented in the disciplinary proceedings.

d. With the exception described below, annual evaluations shall be based, on conference memos issued in prior annual evaluation periods, but such conference memos may only be referred to as establishing prior notice to the employee of the concerns set forth. Generally, in order to provide an opportunity for remediation of job-related deficiencies, administrators shall not refer to a conference memo issued for the first time within thirty (30) days of the employee's annual evaluation. However, if conduct arose during that period of time which was of significant nature, such reference is permissible. When a conference memo issued within the last thirty (30) days is not referenced in the current year's evaluation, it may be referenced in the next year's annual evaluation.

Any conference memo for which there is no repetition of the concern, event, conduct or incident which gave rise to the conference memo, except those relating to serious misconduct such as theft, substance abuse, or violence, shall be void after three (3) years. Following expiration of this three (3) year period, upon request of the employee, such conference memos shall be removed from the employee's files.

- 7.0 A "letter or reprimand" will be identified as such. If the District determines that a "letter of reprimand" is to be used, the issuance procedures set forth in Sections 6.0 a, b, c, and d above will be followed.
- 7.1 Any "letter of reprimand" for which there is no repetition of the concern, event, conduct or incident which gave rise to the "letter of reprimand," except those relating to serious misconduct such as theft, substance abuse, or violence, shall be void after five (5) years. Following expiration of this five (5) year period, upon request of he employee, such "letters of reprimand" shall be removed from the employee's files.
- 8.0 Prior to taking disciplinary (suspension, demotion, or dismissal) action against an employee, the responsible administrator shall advise the employee that disciplinary action may be taken and schedule a meeting to discuss the matter. The employee shall, upon request, be entitled to be accompanied at this meeting, or at an investigatory meeting with the responsible administrator, which the employee reasonably believes will result in discipline, by a Local 572 representative or job steward from the site. Non-availability of the employee or representative for more than a reasonable time shall not delay appropriate action, if any. This right shall not extend to routine conferences or any other meetings or to any conferences conducted under the Evaluation Procedures of this Article. However, upon mutual agreement between the employee and the administrator a union representative may attend such meetings. Such agreement shall not be withheld by the administrator in violation of applicable law. Claimed violations of this Section shall be presented through appropriate disciplinary appeals.
- 9.0 Employees required to attend meetings scheduled by the District pursuant to this Article shall be paid appropriate mileage pursuant to Article XIV.
- 10.0 Discussions between a Unit employee and District supervision concerning the employee's unsatisfactory work performance or work-related problems shall, to the extent practicable, be conducted privately. For the purpose of this Section, "privately" means either a private location, or a location which may be in public view, but is not within earshot of other employees.

ARTICLE XI

TRANSFER PROCEDURES

- 1.0 For the purposes of this Article, "transfer" means a permanent change of an employee's work location without a change in his/her classification or shift. Transfers may be initiated either by written request of the employee (voluntary transfers) or by the District (involuntary transfers). For purposes of this Article, "work location" means any work site such as a school, area office, or administrative office or an employee reporting location.
- 2.0 <u>Involuntary Transfers</u>: An involuntary transfer of an employee is one instituted by the District. Involuntary transfers may occur when in the best interest of the District. The District shall make a reasonable effort to notify the employee at least five (5) working days prior to the effective date of an involuntary transfer. Any employee who is involuntarily transferred shall, upon request, be informed in writing of the reason(s) for the transfer from the appropriate administrator. No employee shall be involuntarily transferred for punitive or disciplinary reasons or in reprisal for the exercise of any right provided by this agreement. This restriction on involuntary transfers shall not apply when the District determines that keeping the employee at his/her current location would be detrimental to the health, welfare or safety of the employee, administrators, students, or other employees.
- 2.1 The District will make a reasonable effort to make transfers pursuant to Section 2.0 within the same geographic administrative region or a proximate geographic administrative region.
- 3.0 <u>Voluntary Transfers</u>: When a permanent employee wishes to transfer, he/she must complete the appropriate District form and submit it to his/her immediate administrator for processing. Within ten (10) working days of receipt by the designated administrator of a request for transfer (change of work location), the designated administrator will provide the employee with an acknowledgement of receipt of such request. If the employee does not receive acknowledgement of the receipt, the request will be deemed received ten (10) working days after the delivery of the request by the employee to the designated administrator.

The appropriate branch head or designee must approve the request to place the employee on the transfer eligibility list, but may defer approval of release to transfer for up to six (6) months. A deferral will not be based on punitive or discriminatory reasons. An employee whose transfer request is deferred shall receive a copy of that deferred transfer request stating the reasons for the deferral from the employee's site or designated administrator within ten (10) working days of the receipt of the transfer request.

3.1 A file of names of employees with approved transfer requests shall be maintained by the Personnel Commission. When an appointing authority

Article XI - Transfer Procedures

advises the Personnel Commission that a position needs to be filled, the Personnel Commission will provide the appointing authority with a certified list of eligible transfer candidates along with other certified eligible candidates.

- 3.2 Employees requesting transfer may be subject to interview by the appropriate administrator.
- 3.3 If an employee is rejected for appointment to a specific vacancy, he/she shall be entitled to know the reason(s) for the rejection, if requested. Transfer appointments will not be denied for punitive or discriminatory reasons.
- 3.4 Transfer requests shall remain on file for one (1) year unless the employee terminates employment, takes a reduction to limited-term status, changes classification, or accepts or declines an offer of transfer.
- 3.5 Employees must accept or decline a transfer offer within three (3) working days from the date the offer is made. Those accepting a transfer must be able to report to the new location within ten (10) working days.
- 3.6 A change in an employee's work shift that required payment of a work differential as specified in Article XIV, (Wages) Section 4 shall not be considered a transfer for the purposes of this Article. Assignments to new or vacant positions for which a work differential is designated, other than a temporary assignment of less than twenty (20) working days, shall be made on the basis of class seniority from among those employees within the appropriate classification who submit written requests for reassignment to the location of the new or vacant position.

ARTICLE XII

SUBCONTRACTING

During the life of this Agreement, the District shall not subcontract or contract out for services or work currently performed by employees in this bargaining unit except in compliance with state law.

ARTICLE XIII

LEAVES OF ABSENCE

- 1.0 "Leaves of Absence" Defined: Probationary and permanent employees shall be eligible for certain paid and unpaid leaves of absence. A leave is an authorized absence from a job classification granted to probationary or permanent employees, for a specified purpose and period of time, with the right to return to active service unless the employee's service would otherwise have been terminated. Leaves are either "permissive" or "mandatory". As to permissive leaves, the term "may" is used and the District retains discretion as to whether they are to be granted, and as to the starting and ending dates of the leave. As to mandatory leaves, the term "shall" is used and the District has no discretion as to whether the leave is to be granted to a qualified employee.
- 2.0 <u>Rights Upon Return</u>: Any employee returning from a leave of ninety (90) days or less will be returned to the location from which the leave was taken except that the employee may be transferred if such a transfer would have been made if the employee had been on duty. An employee returning from a leave of more than ninety (90) days will have return rights only to a position in his/her class.
- 3.0 <u>Restrictions</u>: An unpaid leave of absence may not be converted to a paid leave of absence, except in the case of pregnancy disability as provided in Section 9.3 of this Article.
- 4.0 <u>Applications</u>: Applications for permissive leaves of absence must be submitted on or before the dates established by this Article. Exceptions may be made in the sole discretion of the District.

5.0 <u>Notification Requirements</u>:

- 5.1 Unless otherwise provided in this Article, an employee must make every reasonable effort to contact and notify the appropriate administrator or designee the working day prior to the beginning of an absence, but, except in exceptional circumstances, notification should not be later than one hour before the beginning of the employee's assignment for that day, except that specific reporting requirements may be established for certain classes and positions_requiring adherence to strict time schedules. Unless such notice is given, failure to return to work after the fifth (5th) consecutive working day of absence may be considered as being absent without leave and subject to termination of employment.
- 5.2 All employees returning to service must notify the appropriate administrator or designee at least one (1) hour before the end of the regular working day prior to the day of anticipated return. If such notice is not given and both the employee and a substitute report for duty, only the substitute is entitled to work and to be paid for that day.

- 6.0 <u>Cancellation or Early Return From Leave</u>: A request by an employee for cancellation of or early return from a leave once commenced or for cancellation of a request for a leave shall be granted unless there are no vacancies in the job classification. Exceptions may be made in the sole discretion of the District.
- 7.0 Expiration of Leave: Except in the case of illness or industrial injury/illness leave, or as otherwise provided in this Article, twenty (20) days before the expiration of a leave for ninety (90) days or more, or five (5) days before expiration of a leave for at least twenty (20) but less than ninety (90) days, the employee should make every effort to notify the Personnel Commission of his or her intention to return, or request an extension of leave, if eligible. Unless such notice is given, failure to return to work upon expiration of the leave may be considered resignation from service.
- 8.0 <u>Bereavement Leave (Paid)</u>: An employee is entitled to a paid leave of absence from the District, not to exceed three (3) days, on account of the death of a member of the employee's immediate family provided acceptable proof of death and relationship is provided if requested and the leave of absence commences within ten (10) calendar days of notification of the death. If more than one such death occurs simultaneously, the leave may be taken consecutively. If out-of-state travel or more than two-hundred (200) miles one-way travel is required and requested, an additional two (2) days shall be granted. The immediate family is defined as the following relatives of the employee:
 - a. Spouse or cohabitant who is the equivalent of a spouse;
 - b. Parent (includes in-law, step, foster, of cohabitant who is the equivalent of a spouse);
 - c. Grandparent (includes in-law, step);
 - d. Child (includes son/daughter-in-law, step and foster child);
 - e. Grandchild (includes grandchild of spouse, step and grandchildren);
 - f. Brother;
 - g. Sister; and
 - h. Any relative living in the employee's immediate household.

A permanent employee may interrupt or terminate vacation to take bereavement leave.

9.0 Pregnancy and Related Disability Leave (Paid and Unpaid):

- 9.1 <u>Paid Disability Leave</u>: For that period of time during which the employee is physically disabled and unable to perform her regular duties due to pregnancy, miscarriage, childbirth and recovery therefrom, she shall be permitted to utilize her illness leave pursuant to Section 11.0 of this Article.
- 9.2 <u>Physician Certifications</u>: A pregnant employee shall be permitted to continue on active duty until such date as she and her physician determine that she must absent herself due to pregnancy disability, provided that she can and does continue to perform the full duties and responsibilities of her position. The employee must also supply to the District her physician's certification as to the beginning and the ending dates of actual pregnancy-related disability for which paid illness absence is claimed and her physician's release to return to active duty.
- 9.3 <u>Optional Unpaid Portion</u>: A pregnant employee in active status shall, upon request, be granted an unpaid pregnancy leave prior to the period of actual disability, and still qualify for paid illness absence during the actual disability. This is the only exception to Section 3.0 of this Article.
- 10.0 <u>Child Care Leave (Unpaid)</u>: An unpaid leave may be granted to a permanent employee to care for such employee's own (including adopted; under legal guardianship) child of under three (3) years of age. Proper written application must be submitted to the Personnel Commission at least ten (10) working days prior to the commencement of such leave. The leave, together with any renewal thereof, shall not exceed thirty-nine (39) calendar months in duration.
- 11.0 <u>Illness Leave (Paid)</u>: An eligible employee shall be granted a leave of absence because of illness, or injury, or quarantine of the employee.
- 11.1 Each employee shall accrue 0.05 hours of full-pay illness absence credit for each hour for which salary is received excluding overtime.
- 11.2 At the beginning of the first pay period of each fiscal year upon initial regular appointment, reemployment or reinstatement, each employee who has accrued fewer than the number of full-pay illness absence hours equivalent to one hundred (100) days shall be credited with the number of half-pay illness absence days which, when added to the accrued full-pay illness absence days equals the equivalent of one hundred (100) days of full and half-pay illness absence days.
- 11.3 At the beginning of the first pay period of each fiscal year upon initial regular appointment, reemployment or reinstatement, each employee in paid status shall receive credit for full-pay illness leave of absence up to thirteen (13) days (pro-rated for those employed less than a full year) prior to accrual. However, an employee who uses such a credit prior to actual accrual shall not accrue or be credited with additional leave until the negative balance has been restored. If an employee is paid for more than the illness absences to which entitled, or terminates employment prior to

accruing leave taken in advance, the employee shall be required to refund to the District the salary to which not entitled. This requirement shall be waived in the event of the employee's death or physical or mental disability which precludes the employee from returning to District employment.

- 11.4 Unused full-pay illness absence credit shall be cumulative from year to year without limitation. Half-pay illness credit shall not be cumulative from year to year.
- When a permanent employee is absent under this Section and such 11.5 absence is properly verified, the employee will receive his or her full normal pay up to the total of the employee's full-pay illness benefits. Full-pay illness benefits shall be used before available half-pay benefits may be used. Additional days of illness absence will be at half-pay up to the total of half-pay days credited, if available, unless the employee requests use of an accrued vacation which he or she may have. The amount of paid illness absence taken in any pay period shall not be in excess of the illness absence accumulated by the close of the pay period immediately preceding the illness absence, except as provided in Section 11.3. An initial probationary employee must render service and shall not be eligible to be paid for more than the equivalent of six (6) days of full-pay illness leave until the first (1st) day of the pay period following completion of one hundred and thirty (130) days of paid service in regular assignments. Half-pay illness leave shall not be paid during this time. When all paid and unpaid leaves of absence and vacation benefits have been exhausted, a regular employee who is unable to assume the duties of his/her position shall be placed on a reemployment list for a period of thirty-nine (39) months as if he/she were being laid off. An employee on a reemployment list shall have the same rights and benefits as an employee laid off for lack of work or lack of funds.
- 11.6 An employee who is absent shall be required to certify the reason for absence on the appropriate form. Also, the District may verify any claimed illness, injury, or disability under this Section before authorizing any compensation.
- 11.7 An employee absent from duty for an illness, injury or surgery for more than five (5) consecutive working days shall be required to submit a signed attending physician's statement or appropriate health form to the immediate administrator. The immediate administrator may refer, upon cause, the returning employee for health approval prior to readmission or following readmission. A returning employee referred for health approval by the immediate administrator shall be medically evaluated within a reasonable period of the returning employee's submission of all necessary and required documentation for that evaluation. A returning

employee who receives a medical evaluation for health approval shall receive notice of the results of that evaluation within a reasonable period following receipt by the evaluator of all necessary and required documentation to complete that evaluation. A denial of return from leave of absence based on District health standards may be appealed to the Personnel Commission in accordance with Personnel Commission Rule 836.

- 11.8 If a permanent employee resigns and returns within thirty-nine (39) months of the last date of paid service to permanent status, the number of hours for which the employee was entitled to full-pay illness absence shall be restored, unless the employee's illness balance had been transferred to another agency or used in computation of retirement allowance.
- 11.9 A permanent employee who has exhausted all accumulated illness leave privileges, vacation, and other available paid leaves may be granted additional unpaid illness leave for a period not to exceed six (6) months. Such leave may, upon request, be renewed for two (2) additional six (6) month periods. The total of all unpaid illness leave shall not exceed eighteen (18) months. Until notified to the contrary, the employee may properly assume the leave has been granted.
- 12.0 <u>Industrial Injury/Illness Leave (Paid)</u>: An employee who is absent from District service because of an injury or illness which arose out of and in the course of employment, and for which temporary disability benefits are received under the workers' compensation laws, shall be entitled to a paid leave of absence under the following conditions:
 - a. Allowable paid leave of absence shall be for up to sixty (60) working days for the same injury or illness;
 - b. Allowable paid leave of absence shall not be accumulated from year to year;
 - c. An employee absent under this Section shall be entitled to receive such portion of the salary due for any pay period in which the absence occurs as, when added to the temporary disability indemnity, if any, required under State law, will result in a payment of not more than the employee's salary as of the date of injury or illness. If, as a result of the temporary disability payments, the employee receives more than his/her normal gross salary in any given pay period, any overpayment shall be deducted from future salary warrants. If such recovery occurs more than two (2) pay periods after the overpayment, the District may within its discretion, upon written request by the affected employee, establish a repayment plan for such overpayments not to exceed three (3) pay periods.
 - d. When an authorized leave of absence continues into the next fiscal year, the employee shall be entitled to only the amount of unused leave of absence due for the same illness or injury; and
 - e. Each employee who has received a work-related injury or illness that requires medical attention or absence from work for more than the day of the occurrence must complete a written report of injury or illness on a form to be

provided by the District. This written report must be submitted to the immediate administrator within two (2) working days after the occurrence if the employee is physically able to do so. The site administrator shall, as a result of his or her own investigation, complete the Employer's Report of Occupational Injury or Illness, and shall attach the employee's report thereto. The employee must also report as soon as possible for examination and treatment by a physician who is on the District's Emergency Medical Panel.

Extension of Industrial Injury Leave (Paid): If the employee was 12.1 physically injured during an act of violence related to and during the performance of assigned duties, then the leave of absence may be extended beyond the initial sixty (60) day period up to an additional one hundred twenty (120) days. In order to qualify for such an extension the employee must have: (1) notified the site administrator and appropriate law enforcement authorities within twenty-four (24) hours of the incident if the employee was physically able to do so; (2) completed the employee's written report and reported for treatment as required in "e" above; (3) submitted the Special Physical Injury/Alleged Act of Violence form to the Office of Risk Management and Insurance Services within thirty (30) days of the incident; and (4) submitted to the District a District-approved leave of absence form. The leave of absence form is to be filed with the District in a timely manner so that the District has adequate time to review and process the claim prior to the effective date of the leave extension. Determination whether the injury was the result of an act of violence, and whether the act of violence was related to and during the performance of duties (but not whether it is compensable under workers' compensation laws), shall be made by the Office of Risk Management and Insurance Services. If the employee disagrees with the determination as to whether the injury was the result of an act of violence, the employee will have the right to appeal the determination in writing to the Deputy Director of the Office of Risk Management and Insurance Services or designee within ten (10) working days of receipt of the determination. The Deputy Director of the Office of Risk Management and Insurance Services or designee will reply in writing to the employee within fifteen (15) working days after receipt of the written appeal. Such decision shall be final and not subject to the Grievance Procedure. A determination that the injury is disabling beyond the sixty (60) day period and approval of the paid leave extension shall be contingent upon the employee qualifying for payment of temporary disability benefits under applicable

workers' compensation laws. An employee may be required during the extended period to be evaluated by the District at any time. The District shall continue to advise employees of the requirements of this Section.

12.2 Upon exhaustion of the above-authorized industrial injury/illness leave benefits, the employee shall be permitted to utilize accrued illness benefits or vacation benefits, if any. If the employee continues to receive temporary disability indemnity, the employee shall be paid for any illness and vacation benefits which, when added to the temporary disability indemnity, will result in a payment of not more than full normal salary.

- 12.3 An employee absent under this Section shall remain within the State of California unless the District authorizes the travel outside the State.
- 13.0 <u>Personal Necessity Leave (Paid)</u>: An employee shall, subject to the limits set forth below, be granted a paid personal necessity leave when the gravity of the situations described below requires the personal attention of the employee during assigned hours of service:
 - a. Death or serious illness of a member of the employee's immediate family. (Immediate family is defined as in Section 8.0 of this Article.);
 - b. On a maximum of two (2) occasions during a school year (up to accumulative total of eight [8] hours in a school year), to attend the funeral of a close friend or relative not included in the definition of immediate family (immediate family as defined in Section 8.0 of this Article).
 - c. Accident involving the employee's person or property or the person or property of a member of the employee's immediate family;
 - d. Birth of the employee's child;
 - e. Adoption of a child;
 - f. Religious holiday of the employee's faith;
 - g. Imminent danger to the home of an employee occasioned by a disaster such as flood, fire, or earthquake;
 - h. Other significant event of a compelling nature to the employee, the gravity of which is comparable to the above, which demands the personal attention of the employee during assigned hours and which the employee cannot reasonably be expected to disregard, limited to one (1) occasion in any school year.
 - <u>i</u>. Verifiable automobile failure (including flat tires) up to two (2) hours if the employee's automobile is required to be used for work purposes on that day;
 - j. An appearance of the employee in court or administrative tribunal as a litigant, party or as a witness under an official governmental order for which salary is not otherwise permitted, provided that:
 - (1) Each day of necessary attendance must be certified by the clerk or other authorized officer of a court or other governmental jurisdiction;

- (2) In any case in which a witness fee is payable, such fee shall be collected by the employee and remitted to the Accounting and Disbursements Division; and
- (3) The employee must return to work in cases where it is not necessary for him to be absent the entire day.
- \underline{k} . One (1) of the days allowed for personal necessity leave may be taken for registration or final examinations in District recognized institutions of higher learning. Verification of the registration or examination schedule may be required by the appropriate administrator.
- <u>I</u>. Required attendance at the employee's child's or ward's classroom and meeting with the school administrator because of suspension pursuant to Education Code Section 48900.1.
- <u>m</u>. Up to one (1) day of paid personal necessity leave and additional hours of accrued vacation or unpaid leave not to exceed a total of eight (8) hours per calendar month, forty (40) hours per school year for attendance at the school of the employee's own child, ward, or grandchild for purposes of a school activities leave provided by Section 230.8 of the Labor Code. The employee must notify the immediate administrator or designee at least five (5) working days prior to the absence. This advance notice requirement shall not be applicable in the event of unforeseeable circumstances or emergencies, in which case the employee shall provide as much notice as reasonably possible. The administrator or designee and employee must agree on the date and time of the leave and the employee must provide written verification from the school visited, upon request of the administrator or designee.
- n. An employee shall be allowed up to six (6) additional days of personal necessity leave in any calendar year to attend to the illness of a child, parent or spouse of the employee as provided by Section 233 of the Labor Code. All existing contractual conditions for use of illness leave shall apply to this leave as well. Use of illness leave under this Section 13.0 shall not extend the maximum period of leave to which as an employee is entitled under Article XIII, Section 20.0, Family Care and Medical Leave.
 - o. To be sworn in as a Citizen of the United States.
- 13.1 The following limits and conditions are placed upon allowing a personal necessity leave or absence:
 - a. The total number of days allowed for such leave shall not exceed the number of days granted pursuant to Personnel Commission Rule 807.

- b. The days allowed shall be deducted from and may not exceed the number of full-pay days of accrued illness leave to which the employee is entitled;
- c. The personal necessity leave may not be granted during a strike, demonstration or any work stoppage involving the Union;
- d. Written request on the appropriate form shall be filed with the appropriate administrator no less than five (5) working days in advance of a religious holiday, court appearance, or school visit pursuant to 13.0(k) above; and
- e. The employee may be required to verify the nature of such necessity.
- 14.0 <u>Personal Leave (Unpaid)</u>: An unpaid leave may, at the discretion of the District, be granted to a permanent employee for a period not to exceed fifty-two (52) consecutive calendar weeks, except as provided in "f" below, for a specific personal reason satisfactory to the District, including but not limited to the following:
 - a. To be with a member of the immediate family who is ill;
 - b. To accept an opportunity of a superior character which will result in the employee rendering more effective service on return to the District;
 - c. To rest, subject to approval by the District;
 - d. To remain with spouse if a change of residence is required;
 - e. To pursue a program of study in residence in an approved institution of higher learning or under a fellowship foundation approved by the State Board of Education;
 - f. To serve as a State Legislator--such leave shall be renewed annually during tenure of office, the above limitation notwithstanding; or
 - g. To serve in an elective position in the city, county, state or federal government, other than the State Legislature.

Applications must be filed with the Classified Personnel Assignments Branch and are subject to cancellation in the event of layoff.

15.0 <u>Military Leave</u>: An appropriate military leave of absence shall be granted to any qualified employee in accordance with the provisions of the Education Code and Military and Veterans Code. The District may grant leave under this section in excess of the statutory requirement.

- an employee to appear, in response to a subpoena duly served, when other than a litigant: (a) in a case before a grand jury; (b) in a criminal case before a court within the State; or (c) in a civil case in a court within the county in which the employee resides or outside of said county if within one hundred fifty (150) miles of place of residence. Leave shall be granted for the days of attendance in court as certified by the clerk or other authorized officer of such court or grand jury or by the attorney for the litigant in the case. In any case in which witness fees are payable, such fees shall be collected by the employee and remitted to the Accounting and Disbursements Division. An employee whose regular assignment is to other than the day shift will be reassigned to the day shift on each day that such court subpoena occurs. Subject to the possibility of making reasonable travel arrangements, the employee shall be required to report for work during the balance of her/his assigned working day or week when her/his presence is not required pursuant to said subpoena.
- 17.0 <u>Jury Duty Leave (Paid)</u>: A paid absence or leave shall be granted to any employee required to render jury service in any court within the state. An employee shall provide to his/her supervisor no less than five (5) working days notice of a summons to jury service. However, if the summons to the employee does not allow for at least five (5) working days notice, the employee shall notify his/her supervisor immediately upon receipt of the summons.
- 17.1 All jury fees received shall be remitted to the Accounting and Disbursements Division with the following exceptions:
 - a. Mileage fee reimbursement;
 - b. Fees earned on holidays, vacation or any day an employee is not in paid status;
 - c. That amount of jury fee which exceeds employee's daily gross earnings.
- 17.2 An employee who is normally assigned to the "B" or "C" night shift during jury service shall be temporarily reassigned to the "A" day shift during the term of his/her jury service.
- 17.3 Subject to the possibility of making reasonable travel arrangements, the employee shall be required to report for work during the balance of her/his assigned working day or week when her/his presence is not required for jury duty.
- 18.0 <u>Peace Corps, Red Cross and Merchant Marine Leaves</u>: Permanent employees covered by this Agreement shall be granted an unpaid leave of absence not to exceed twenty-five (25) months to serve in the Peace Corps. During any period of war or national emergency, unpaid Red Cross Leave or unpaid Merchant Marine Leave shall be

granted to any employee who enters the full-time paid service of the American Red Cross or the U. S. Merchant Marine in accordance with the provisions of the Military and Veterans Code and the Education Code.

19.0 Miscellaneous Leaves:

- 19.1 <u>Employment Examination</u>: Upon giving his/her immediate administrator advance notice of not less than two (2) working days, an employee shall be permitted a paid absence to take an examination or participate in other District employment procedures during working hours. If less than two (2) days' notice is given by an employee, permission to participate without loss of pay is subject to approval by his/her immediate administrator.
- 19.2 <u>Annual Physical Examination</u>: A permanent employee shall be granted up to one (1) day per year with pay for the purpose of a comprehensive physical examination provided that verification of such an examination is submitted to the District.
- 19.3 <u>Witness</u>: An employee who is subpoenaed to be a witness in the appeal by another employee of a decision of the Workers' Compensation Appeals Board arranged by the District's Insurance Section may attend without loss of salary.
- 19.4 <u>Epidemics and Emergencies</u>: An employee with regular status shall be paid her/his regular salary for any period during which she/he is unable to work at her/his regular place of employment because it is closed by the District due to quarantine, epidemic, or other conditions involving the health or safety of students or employees. To be eligible for such pay the employee must be ready, able and willing to

perform her/his customary or other reasonable and suitable duties at different work locations as designated by the District. Nothing contained herein shall be construed to limit the authority of the District to make temporary assignments of employees to different or additional locations, shifts, or work duties for the purposes of meeting emergencies.

20.0 <u>Family Care and Medical Leave</u>: An unpaid Family Care and Medical Leave shall be granted, to the extent of and subject to the restrictions as set forth below, to an employee who has been employed for at least twelve (12) months and who has served for one hundred thirty days (130) workdays during the twelve (12) months immediately preceding the effective date of the leave. For purposes of this section, furlough days and days worked during off-basis time shall count as "workdays". Family Care and Medical Leave absences of twenty (20) consecutive working days or less can be granted by the immediate administrator or designee. Leaves of twenty (20) or more consecutive working days can be granted only by submission of a formal leave application to the Classified Personnel Assignments Branch.

20.1 Definitions: For purposes of Family Care and Medical Leave, the following definitions shall apply: (1)"Child" means a biological, adopted or foster child; a stepchild; a legal ward; or a child of an employee standing "in loco parentis," such child being either under eighteen (18) years of age or an adult dependent who is incapable of self care due to a mental or physical disability; (2)"Spouse" means a husband or wife of an employee; (3)"Parent" means a biological, foster, or adoptive parent; a person who stood "in loco parentis" to the employee when the employee was a child; a stepparent; or a legal guardian; and does not include a parent-in-law; (4) "Family member" means "child," "spouse," or "parent" as defined above; (5) "Serious health condition" means an illness, injury, impairment, or other condition that involves either "in-patient care" or "continuing treatment"; (6) "Inpatient care" means a stay in a hospital or other medical facility and includes any subsequent treatment in connection with inpatient care; (7) "Continuing treatment" means treatment by a "health care provider" that involves one or more of the following: (a) a period of incapacity of more than three (3) consecutive calendar days (as well as any subsequent treatment or period of incapacity relating to the same condition) that also involves either two or more treatments by a "health care provider," or treatment by a "health care provider" on at least one (1) occasion that results in a regimen of continuing treatment under the supervision of a "health care provider"; (b) any period of incapacity due to pregnancy (including morning sickness); (c) any period of incapacity or treatment for an incapacity due to a chronic health condition that requires periodic visits for treatment, which continues over an extended period of time, and may cause episodic (i.e., a period of incapacity for less than three (3) days rather than a continuing incapacity (such as asthma, diabetes, and migraine headaches); (d) a period of incapacity that is long-term due to a condition for which treatment may not be effective; and (e) any period of absence to receive multiple treatments, including treatment of a condition that would likely result in a period of incapacity for a period of

more than three (3) days if not treated; (8) "Health care provider" means an individual holding either a physician's and surgeon's certificate or an osteopathic physician's and surgeon's certificate issued pursuant to Article 4 of Chapter 5 of Division 2 of the California Business and Professions Code, or any other individual duly licensed to practice medicine in another state or jurisdiction who directly treats or supervises the treatment of the serious health condition, or by any other person determined by the Secretary of Labor to be capable of providing health care services. The definition includes podiatrists, dentists, clinical psychologists, optometrists, chiropractors (limited in scope), nurse practitioners, nurse midwives, and certain Christian Science practitioners.

20.2 <u>Reasons for Leave</u>: Family Care and Medical Leave may be granted for reason of the birth of a child of the employee, or the placement of a child with an employee in connection with the adoption or foster care of the child by the employee. If the leave is taken for any of these reasons, the leave must be concluded within twelve (12) months of the birth, the adoption, or the foster care placement of the child. In addition, leave may be granted because of the serious health condition of a child of the employee, the employee's own serious health condition, or the care of a parent or spouse who has a serious health condition.

20.3 Length of Leave: The leave, together with any renewal thereof, shall not exceed the number of days equivalent to a total of twelve (12) normally scheduled workweeks in a twelve (12) month period measured forward from the beginning date of the employee's first Family Care and Medical Leave, effective July 1, 2007. An employee will be entitled to twelve (12) weeks of leave during the 12-month period beginning on the first date Family Care and Medical Leave is taken; the next 12-month period would begin the first time Family Care and Medial Leave is taken after completion of any previous 12-month period. For the period of time up to, and including June 30, 2007, the leave, together with any renewal thereof, shall not exceed the number of days equivalent to a total of twelve (12) normally scheduled workweeks in a fiscal year. An employee will retain the full benefit of twelve (12) weeks of leave under whichever calculation method (either fiscal year or 12-month period measured forward) affords the greatest benefit to the employee during a 60-day transition period. This transition period shall be from July 1, 2007 through August 31, 2007.

Any leave an employee takes for the reasons specified in Section 20.2 above will be counted against the employee's annual leave entitlements under the federal Family and Medical Leave Act of 1993 and the California Family Rights Act of 1991, as amended. This leave runs concurrently with any other leave the District offers for which the employee is qualified. Leave caused by pregnancy, childbirth or related medical conditions under Section 9.0 of this Article is separate and apart from the provisions of Family Care and Medical Leave herein. Employees are entitled to the leave allowed under Section 9.0 and, in addition, up to the full twelve (12) weeks of Family Care and

Medical Leave. However, leave taken on account of pregnancy, childbirth, or related medical condition will be counted against the employee's annual leave entitlement under the federal Family and Medical Leave Act of 1993.

20.4 <u>Intermittent Leave</u>: The leave may be taken intermittently or on a reduced work schedule. If the leave is taken for reason of the birth, adoption, or foster care placement of a child of the employee, the basic minimum duration of the leave shall be two (2) weeks; however, the District shall grant the employee leave of less than two (2) weeks' duration on two occasions. If the leave is taken for a serious health condition of the employee or of the employee's family member, leave may be taken intermittently or on a reduced schedule when medically necessary, as determined by the health care provider of the person with the serious health condition. An employee may take such leave for as short a time as one (1) hour (can be less than one [1] hour, if necessary).

If an employee does take intermittent or a reduced-schedule leave that is foreseeable based on a planned medical treatment of the employee or the employee's family member or for the birth, adoption, or foster care placement of a child, the District has the right to transfer temporarily the employee to an available alternative position for which the employee is qualified and which better accommodates the recurring periods of leave during the duration of the intermittent or reduced-scheduled leave. The alternative position must have equivalent pay and benefits but does not have to have equivalent duties. The alternative position may include the altering of the employee's current job.

The District may also transfer the employee to a part-time job with the same hourly rate of pay and benefits. Upon the conclusion of the intermittent or reduced-schedule leave, the District will place the employee in the same or equivalent job the employee had when the leave started.

20.5 <u>Notification</u>: If the need for the Family Care and Medical Leave is foreseeable more than thirty (30) calendar days prior to the employee's need for leave, the employee shall give at least thirty (30) days notice. If less than thirty (30) days, the employee must provide the immediate supervisor with as much advance notice as possible but, at the least, within two (2) business days of learning of the need for the leave. These advance notice requirements shall not be applicable in the event of unforeseeable circumstances or emergencies. Whenever possible, if the need for leave is foreseeable due to a planned medical treatment or supervision, the employee must make a reasonable, good faith effort, subject to the approval of the employee's or family member's health care provider, to schedule the treatment or supervision to avoid disruption to the District's operations. In giving notice, the employee must include the qualifying event for which the leave is needed, e.g., birth of a child, serious health condition of parent, etc.

Medical Certification: For leaves to care for a child, spouse or 20.6 parent who has a serious health condition, the employee must submit to the immediate administrator or, if applying for a formal leave must attach to the leave application, medical certification from the health care provider which includes: (1) the date, if known on which the serious health condition commenced; (2) the probable duration of the condition; (3) an estimate of the time that the health care provider believes the employee needs to care for the individual; and (4) a statement that the serious health condition warrants the participation of the employee to provide care. If the leave is for the serious health condition of the employee, the employee must submit to the immediate administrator and/or, if applying for a formal leave, must attach to the leave application, medical certification as specified in (1) and (2), above, plus a statement that, due to the serious health condition, the employee is unable to work at all or is unable to perform one or more of the essential functions of the employee's position. Medical certification must be submitted no later than fifteen (15) calendar days after the leave request has been made. If the deadline by which the employee is to submit the medical certification is after the leave has started, the employee will be considered to have taken Family Care and Medical Leave pending the District's receipt of the proper certification. However, if the employee fails to provide proper certification, the employee will be treated as if he or she did not qualify for, and thus never took, Family Care and Medical Leave, will be treated as if he or she sought a leave of absence under another provision of this Agreement, and will not be given the protections set forth in this Article.

In the case of leave due to a serious health condition of the employee, the District reserves the right to require, at its own expense, that the employee obtain the opinion of a second or even third health care provider designated by the District but not employed on a regular basis by the District. The second health care provider, if required, shall be

selected by the District. The third health care provider, if necessary, shall be jointly approved by the District and the employee and this provider's opinion shall be binding. If the employee's leave has already begun during this medical review process, the employee will be considered to have taken Family Care and Medical Leave, pending the result of the examinations by the second and, if necessary, third health care provider.

If additional leave beyond that provided in the certification is required, the employee must submit a new certification by the relevant health care provider.

- 20.7 <u>Restrictions</u>: In the event that parents who are both District employees each wish to take Family Care and Medical Leave for the birth, adoption, or foster care placement of their child, the combined total amount of leave that will be granted such employees will be twelve (12) workweeks during a 12-month period, as defined in Section 20.3 above. These employees will still be eligible to take the remainder of their individual twelve (12) workweek allotment for Family Care and Medical Leave for a purpose other than the birth, adoption or foster care placement of a child.
- Compensation: The Family Care and Medical Leave shall be an unpaid leave. An employee who takes Family Care and Medical Leave and who has accrued vacation may elect, or the District may require, the employee to utilize vacation for this purpose, in lieu of unpaid status. An employee who takes leave for the employee's own serious health condition may elect, or the District may require, the employee to utilize accrued illness days for the leave. During the leave, the District will continue to provide the health benefits package, and maintain the District contribution obligation pursuant to Article XV (Health and Welfare), during the Family Care and Medical Leave (except as provided below) to an employee who is otherwise eligible for health benefits. However, an employee who does not return from such leave, or who works less than thirty (30) days after returning from the leave (unless the employee retires within thirty [30] days after returning from leave) will be required to reimburse the District for the District's cost of providing the health benefits package. The District, however, will not provide such health benefits for an employee for any leave period beyond twelve (12) workweeks. Accordingly, if an employee combines pregnancy leave with a Family Care and Medical Leave, the employee will only be entitled to continued health benefits for the first twelve (12) workweeks of leave. Thereafter, the District will provide the employee with health benefits to the same extent and under the same conditions as it provides to employees on other, similar leaves of absence.
- 20.9 <u>Seniority</u>: Accrual of seniority credit for the period of Family Care and Medical Leave shall be in accordance with Personnel Commission Rule 740.
- 20.10 <u>Certification to Return to Work</u>: The provisions of Section 11.6 and 11.7 shall apply to employees returning to work from a Family Care and Medical Leave (absence) due to the employee's own serious health condition.
- 20.11 <u>Early Return From Leave</u>: If the amount of leave needed is actually less than initially requested, the employee must notify the District of such an occurrence.

Once the employee provides such notification, the District must reinstate the employee to the same or equivalent position within two (2) days.

- 21.0 <u>Charter School Leave (Unpaid)</u>: An employee shall, subject to the limits set forth below, be granted an unpaid leave to serve in an assignment at a Board of Education-approved Charter School:
 - a. The leave shall be for a minimum of one (1) year. The leave shall be extended upon request of the employee; however, the total period of leave shall not exceed the duration of the initial charter or five (5) years whichever occurs first;
 - b. For an employee not assigned to a school or program that is being converted to a Charter School, the leave shall commence at the beginning of the next school year (July 1); exceptions may be provided in the sole discretion of the District;
 - c. Salary and benefits received by the employee during the period of leave shall not be the responsibility of the District, but shall be established and provided by the Charter School in accordance with the Charter School petition approved by the Board of Education;
 - d. Return from Leave to District service shall be in accordance with the provisions contained in this Article unless the employee has been laid off by the District;
 - e. Upon return to District service from a Charter School leave, no employee shall receive more favorable treatment than employees in the same classification who remain with the District;
 - f. Employees separated involuntarily from their Charter School assignment may be subject to administrative or disciplinary action by the District for conduct which occurred at the Charter School in the same manner as if the conduct had occurred while the employee was actively employed by the District.
- 22.0 <u>Break in Service</u>: Periods of paid leave of absence shall not be considered a break in service. A supervisor on any paid leave status shall continue to earn seniority and vacation credit.
- 23.0 To the extent that Personnel Commission Rule 790 grants additional leave rights greater than those included in this contract and enjoyed by bargaining unit members prior to this Agreement, those rights shall continue as long as they continue in the Personnel Commission Rules.

ARTICLE XIV

WAGES AND SALARIES, PAY ALLOWANCES, DIFFERENTIALS, AND SPECIAL SALARY PRACTICES

- 1.0 <u>Wages and Salaries</u>: The wages and salaries for Unit employees have been negotiated in good faith between the Teamsters and the District and shall be as set forth in Appendices A and J of this Agreement. The wages and salaries set forth in Appendices A and K are intended to, and do, meet any prevailing wage obligations which are or may be imposed upon the District.
- 1.1 All employees in this unit in classes that are currently eligible to receive semi-monthly payment of wages and salaries shall continue to receive such semi-monthly payments.

2.0 Miscellaneous Paid Allowances

- a. The District shall, for the duration of this Agreement only, continue to provide and/or make available to employees the uniforms, uniform allowance and protective gear which the District currently provides to unit employees under District practice.
- 2.1 <u>Uniforms</u>: If distinctive uniforms are required for an employee, the cost of purchase, lease, or rental of uniforms, identification badges, emblems, and cards for the employee shall be borne by the District. Such items provided by the District shall be returned to the District upon separation from the service or termination of the assignment.
- 2.2 <u>Mileage Reimbursement</u>: Employees who are required to use their personal vehicles for District business shall be reimbursed at the Internal Revenue Service established standard business rate for such usage for all miles driven in <u>District service</u>.
- a. Daily Flat Rate: In recognition of the fact that unit employees use their own vehicle to haul District authorized materials, tools, supplies, and equipment, effective July 1, 2015, employees in the classifications listed below shall be eligible for "flat rate" mileage in addition to the per mile reimbursement specified in 2.2 above. Flat rate mileage shall be authorized for eligible field assigned employees for each day or part of a day of such use of their personal vehicle. Those eligible employees shall receive flat rate mileage each day their personal vehicle is driven in District service. The rate shall be \$7.00 per day for vehicles hauling in excess of 75 pounds for District authorized materials, tools, supplies and equipment.

Area Carpentry Supervisor

Area Electrical Supervisor

Area Heating and Air Conditioning Supervisor

Area Operations Supervisor

Area Painting Supervisor

Area Plumbing Supervisor
Assistant Hardware Supervisor
Associate Electrical Engineer
Carpentry Supervisor (Building Moving)
Carpentry Technical Supervisor

<u>Central Shops Supervisor</u> <u>HVAC Test and Inspection Supervisor</u>

<u>Computer Inventory Hardware Control</u>
<u>Painting Technical Supervisor</u>

<u>Supervisor</u> <u>Paving Supervisor</u>

<u>Design Network Engineer</u>

Electrical Engineer

Electrical Engineer

Plumbing Technical Supervisor

Roofing Inspection Supervisor

<u>Electrical Technical Supervisor</u> <u>Roofing Supervisor</u>

<u>Electronics Technical Supervisor</u>
Food Services Manager Series

Roofing Technical Supervisor
School Administrative Assistant

<u>Glazing Supervisor</u>
<u>Hardware Supervisor</u>
<u>Senior Electrical Inspector</u>
<u>Senior Painting Inspector</u>

<u>Heating and Air Conditioning Technical</u>
Supervising Electrical Engineer
Supervising Electrical Inspector

HVAC Technical Supervisor (High-Rise) Supervising Plumbing Inspector

b. The foregoing shall apply provided that the employee's vehicle has the appropriate weight rating to carry/haul tools and/or materials being transported. The District shall not be liable for loss or injury deemed inappropriate or unintended by vehicle manufacturer.

3.0 Pay Differentials - General

- 3.1 An earned salary differential in addition to the regular rate of pay specified in Appendices A and J shall be paid to affected employees under the conditions and in the amount specified in this Article.
- 3.2 Assignment to a new or vacant position for which a salary differential is designated, other than a temporary assignment of less than twenty (20) working days or a bilingual differential, shall be made on the basis of seniority among those employees in the appropriate class who request such an assignment.
- 3.3 Long-term salary differentials as designated in this Article shall be based on the special requirements of a particular position or the authorized use of special skills by a particular incumbent for twenty (20) consecutive working days or more and for which payment shall be continued during paid absences of the employee. An employee receiving a long term differential shall not lose such compensation if temporarily assigned, for twenty (20) working days or less, to duties not entitled to such compensation.
- 3.4 A differential authorized under this Article shall not affect salary allocation upon change of assignment.
- 3.5 Differentials for which certification by an administrator is required shall be withdrawn upon certification by the administrator.

- 3.6 <u>Bilingual Differential</u>: A regular employee shall be paid a long-term salary differential for using bilingual skills upon certification from the appropriate District Superintendent or division or branch head that in addition to regular duties of the class, the employee is frequently called upon to speak, interpret, and write a non-English language or to converse fluently in a non-English or sign language. In order to qualify for a bilingual differential, the employee must meet English and non-English or sign language proficiency standards prescribed by the Personnel Commission. Such English and non-English language proficiency standards shall include required reading, writing, and/or oral communication abilities which must be satisfactorily demonstrated pursuant to District examination procedures.
 - a. Eligible full-time employees shall be paid at the rate of forty-six (46) dollars per pay period if required to speak, read, and write a non-English language or use sign language with Level I proficiency, or twenty-eight (28) dollars per pay period if only required to converse in a non-English language or use sign language with Level II proficiency.
 - b. The differential for eligible part-time employees shall be prorated at the same rate that the number of hours of their regular assignment bears to a regular eight (8) hours per day assignment.
 - c. An approved differential shall become effective on the first (1st) day of the pay period following completion of provisions in Section 3.6 above and shall continue during paid absences, provided, however, an appointing authority may certify that a previously approved differential may continue uninterrupted for employees who are reassigned, transferred or promoted to another position requiring the same bilingual skills. The effective date shall be communicated in writing to the effected employee as soon as practicable following completion of the verification procedure(s). This differential shall not affect salary allocation upon change of assignment.
- 4.0 <u>Shift Differential</u>: Shift differentials apply only to classifications eligible to receive such differentials prior to the execution of this agreement.
 - a. Except as provided in paragraph "b" below, non-flat-rated employees who work one-half or more of their assigned time between 5 p.m. and midnight shall receive a shift differential of one step on the salary schedule for their class. Non-flat-rated employees who work one-half or more of their assigned time between midnight and 7 a.m. shall receive a shift differential of two (2) steps on the salary schedule for their class.
 - b. Non-flat-rated employees who work one-half or more of their assigned time between 5 p.m. and midnight and who are on the fifth (5th) step of the salary schedule of their class shall receive a shift differential of five and one-

half percent (5½%). Such employees who work one-half or more of their assigned time between midnight and 7 a.m. and who are on the fourth (4^{th}) or fifth (5^{th}) step of the salary schedule for their class shall receive a shift differential of eleven percent (11%).

- c. The District agrees to continue its practice of paying flat-rated employees who work one-half or more of their assigned time between 5 p.m. and midnight a shift differential of five and one-half percent (5½%); flat-rated employees who work one-half or more of their assigned time between midnight and 7 a.m. shall receive a shift differential of eleven percent (11%).
- d. The provisions of Personnel Commission Rule 586 --Salary Differential for Night Work shall continue to apply to this unit in the same manner as those provisions applied prior to this Agreement unless and until Personnel Commission Rule 586 is changed or amended in which case the parties shall meet and negotiate regarding such changes or amendments.
- 4.1 <u>Salary Placement</u>: Entry-level placement on the salary schedule shall be at the lowest step of the schedule for the classification or at the rate established for the classification, unless the District authorizes hiring at a higher rate.
- 5.0 <u>Step Advancement on the Salary Schedule</u>: A probationary or permanent regular employee, including a flat hourly rate employee who changes to a rate on the salary schedule, shall be advanced to the next highest step as of the pay period following completion of one hundred and thirty (130) days in a paid status in regular assignment(s) in the class, and to higher steps in subsequent years as to the numbered pay period corresponding to the pay period of the last advancement providing the employee completed one hundred and thirty (130) days in paid status in the interim period.
- 5.1 A day in paid status for purposes of this Section shall be defined as any day for which pay is received, including:
 - a. Limited term assignments in the same, equal, or higher class;
 - b. In the event of demotion following promotion to a regular position, time spent in a higher class;
 - c. Time spent on industrial accident/illness, military, Peace Corps, Red Cross or Merchant Marine leaves.
- 6.0 <u>Salary Placement Upon Promotion or Reclassification</u>: Upon promotion or reclassification to a higher class, an employee shall advance to that step of the new salary schedule which is at least two and three-fourths percent (2 3/4 %) percent

above his/her rate of pay, but not to exceed the maximum rate of pay established for the higher class. Such employee shall then receive a step advancement, if applicable, effective as of the first day of the pay period after completion of one hundred and thirty (130) days in paid status in regular assignments in the higher class, exclusive of overtime. A new cycle for subsequent step advancements will thus be established.

- 6.1 A payment shortage in a regular salary warrant received by an employee shall be corrected within thirty (30) days after it is reported to the Payroll Services Branch by the employee's time reporting person, as long as it falls within the terms of 8.0 below.
- 7.0 <u>Supplemental Pay Warrant</u>: A permanent regular employee who does not receive a scheduled pay warrant or receives an underpayment of at least thirty five (35%) of their normal net pay because of problems involving assignment, time reporting, or payroll processing, may request a Supplemental Pay Warrant for hours reported and approved by the employee's work location. The request will be processed and a warrant made available for pick-up after 3 p.m. on the third work day following the request unless the employee requests that the warrant be mailed.
- 7.1 A Supplemental Pay Warrant cannot be made for a pay warrant that has been issued but is subsequently unaccounted for (e.g., lost, delayed in route, stolen after receipt, etc.) or in cases where garnishments, tax liens or the like are being processed.
- 7.2 In the case of a salary warrant issued and mailed but later lost or stolen, a replacement warrant will be issued no later than seven (7) calendar days after the employee submits a Lost Warrant Affidavit form to the Payroll Services Branch.
- 7.3 The District will provide notification to an employee in the event of a garnishment or tax lien.
- 8.0 Payroll Errors Limitations Upon Recovery: Any payroll or other salary errors claimed by an employee against the District in a timely manner as provided in the grievance procedure of Article V shall be corrected retroactively up to a maximum of three (3) years from the date of claim. In the event of an error in favor of an employee, the District shall be limited in its retroactive recovery against the employee to a three (3) year period dating from the discovery of the error. Except for recovery of the amounts received by the employee as temporary disability payments from the District's agent, the District will notify an employee of an overpayment via the employee's location timekeeper or by direct written communication. Written communication will be provided when an overpayment exceeds one hundred dollars (\$100). Where in the judgment of the District the circumstances warrant, the District shall allow the employee to establish a reasonable method of repayment with the Payroll Services Branch.
 - 8.1 <u>Salary Overpayments</u>: Except as set forth in Article XIII, (Leaves)

Section 12.0 (c), when a salary overpayment error has been discovered, the Payroll Branch will notify the employee in writing of the amount and circumstances related to the overpayment and will recommend a suggested method for recovery of the overpayment.

- (1) For cases in which the amount and circumstances are such that it is probable that the employee was unaware of a salary overpayment, two hundred dollars (\$200) per pay period will be the normal limit on repayment deductions. However, in such cases the repayment may be accelerated upon termination of paid status or may be larger than two hundred dollars (\$200) per pay period if necessary to recover the full overpayment.
- (2) The employee may request consideration of alternative methods for recovery of overpayments provided that the time frame for recovery does not exceed the period of time during which the overpayment occurred. If no request is made for an alternative method of recovery within ten (10) calendar days, the recovery shall commence effective with the next pay period using the method recommended by the District in its written notice to the employee.
- (3) When the amount and circumstances are such that the employee knew or should have known that there was an overpayment, the recommended recovery payment will be as much as the entire amount. In such cases, however, the District will work out a suitable recovery payment schedule with the employee which may be as much as the entire amount within one (1) pay period.
- 9.0 <u>Temporary Work in a Vacant Position in Another Classification:</u> When an employee is temporarily assigned to work in an existing vacant position in another classification for at least five (5) consecutive days, or for a number of hours during the pay period which corresponds to five (5) days or more, a limited-term assignment shall be processed in accordance with the provisions of Personnel Commission Rule 585, "Compensation for Employees in Limited-Term Assignments." Copies of this rule shall be available for inspection at all schools and offices. Additional copies may be obtained through the Office of Staff Relations.
- 10.0 <u>Longevity Increment</u>: Longevity increments apply to classifications in this unit eligible to receive this increment.
 - 10.1 The longevity increment shall become effective on the first (1st) day of the second Special School Month following completion of the qualifying number of years of service.

- 10.2 A "year of service" for the purpose of the longevity increment shall be defined as paid service in regular status for 130 days or more within the fiscal year, including time served in probationary or permanent certificated services; however, total assignment hours annually shall not exceed 2080 hours for years of service credit.
- 10.3 The longevity increment shall be part of the employee's basic wage for the purpose of computing overtime but shall not affect salary allocation upon promotion or reclassification to a higher class. Employees paid less than eight (8) hours per day shall receive a proportionate amount of the applicable increment.
- 10.4 The longevity increment schedule for years of qualifying District service shall be:

\$25 per pay period (equivalent to \$.15625 per hour) after 10 years \$30 per pay period (equivalent to \$.18750 per hour) after 15 years \$35 per pay period (equivalent to \$.21875 per hour) after 20 years \$40 per pay period (equivalent to \$.25000 per hour) after 25 years \$45 per pay period (equivalent to \$.28125 per hour) after 30 years

- 11.0 Other Salary Differentials, Pay and Practices: The provisions of Personnel Commission Rule 589 -- Salary Differential for Construction Inspectors with Major Project Assignments, 587-- Salary Differentials for Supervising Special Education Assistants using Specialized Health Care Skills, 575--Salary Differentials for School Administrative Assistant Mentors, 597--Salary Differentials for Training Responsibility and 599 -- Salary Differentials Based upon Variations in Responsibility shall continue to apply to unit member in the same manner as the Rules applied prior to this Agreement unless and until such Personnel Commission Rules are changed or amended, in which case, the parties shall meet and negotiate regarding such changes as or amendments.
- 12.0 Retirement Benefits: In order to address recent changes to CalPERS under the Public Employee Pension Reform Act of 2013 (PEPRA), effective July 1, 2014, all CalPERS eligible bargaining unit employees shall pay one hundred percent (100%) of his/her individual employee contribution to the California Public Employees' Retirement System (CalPERS) at the rate fixed and prescribed by law.
 - 13.0 Pay Day: The Friday payday procedures shall be retained.
- 14.0 <u>Automatic Deposit (EFT):</u> The District shall continue its current practice of making automatic deposits of salary warrants to employees' accounts in a banking institution that participates in the District's direct deposit program.
- 15.0 Bargaining unit members who have been afforded District vehicles for use in the maintenance of licenses or certificates shall continue to receive said benefits.

16.0 Any change in District Vehicle allocation to and/or assessment of employees in this unit shall be discussed with the Union prior to implementation.

ARTICLE XV

HEALTH AND WELFARE

- 1.0 <u>District Contribution Obligations</u>: (as to all eligible District personnel): The District contribution rate and all other matters set forth herein shall be in accordance with the "Health Benefits Agreement between LAUSD And The Unions/Associations Representing District Employees" Agreement on Health and Welfare in Appendix K which is attached hereto for informational purposes only.
- 2.0 <u>Plan Revisions Through the District-wide Health and Welfare</u> <u>Committee</u>: A District-wide Health Benefits Committee shall be formed.
 - a. Composition -- Each union shall be entitled to one (1) HBC member for every 5,000 unit members represented or fraction thereof. The District shall be an official member of the HBC; the District and each union shall have one vote a piece. The District shall provide resource staff as determined by the HBC, and shall provide adequate paid release time for those HBC members who are employees of the District.
 - b. Decision Making -- Consensus shall be used in all HBC deliberations. If a consensus decision cannot be reached, then in the alternative, each union and the District shall have one (1) vote apiece. Any recommended changes to the existing kinds and levels of benefits shall require a 2/3 vote of the members present and voting.
 - c. The HBC may investigate the creation during the term of this Agreement of a joint Employer Health and Welfare trust. Such Trust might include other public or private sector employees as determined by the HBC. The HBC shall review all existing contracts prior to expiration. No contract shall be for more than one (1) year, or awarded without open bid, except upon HBC approval.
 - d. The HBC shall investigate the feasibility of providing benefits to unbenefited part-time employees.
 - e. Benefit Eligibility -- During the term of this Agreement there shall be no changes in the eligibility requirements for District Benefits (see Section 3.0 below).
 - f. Effective no later than January 1 of any plan year, Teamsters shall have the option of informing (in writing) the District and the other unions participating the HBC of its intent to remove its pro rata share of District Health and Welfare Plan health benefit expenditures and assume the responsibility of providing a benefit program for its members, or having the District design and administer a health benefit program for its members. However, any removal of its pro rata share of District health benefit expenditures shall be subject to the

precondition of nine (9) months' written notice to the District regarding Teamsters' intent to do so. Further, its removal shall only be effective on a January 1 plan anniversary date.

- g. The District and the unions/associations will develop plans to address unfunded liability GASB 45 issues through the HBC.
- 3.0 <u>Eligibility for Plans</u>: Eligibility requirements for employees and dependents shall be as provided in the applicable plan and also as follows:
 - a. Every employee who is assigned half-time or more of a full-time assignment in one (1) class, in a status other than substitute, temporary, extra, exchange or relief, shall be eligible to enroll in a plan. The percentage of assignment shall be determined by the District. For employees attaining eligibility under this paragraph, the enrollment year shall be January through December.
 - b. Employees who do not qualify under the preceding paragraph, but who in the previous school year were in paid status for 800 or more hours as a result of any one assignment or any combination of assignments. For employees attaining eligibility under this paragraph, the enrollment year shall be September through August.
 - c. In order to remain eligible, the employee must be in paid status within the assignment basis. However, an employee in an unpaid status who later receives compensation from the District for the unpaid period shall be entitled to reimbursement of direct premium payments made which correspond to the period for which such compensation is allowed. To obtain such reimbursement, the employee shall file application therefore with Benefits Administration.
 - d. In situations where employees are married to one another or share a domestic partner relationship and are covered by the same plan with one (1) listed as a dependent, the dependent shall not, upon divorce, upon termination of the domestic partnership or upon the retirement or death of the spouse/domestic partner, lose any rights the employee would otherwise have had as an eligible employee or retired employee.
 - 4.0 <u>Retirement Benefit Coverage</u>: Qualified employees who retire from the District receiving a PERS/STRS allowance for either age or disability shall be eligible to continue District-paid hospital/medical, dental and vision coverage in which the employee was enrolled at the time of retirement. For the purposes of this section, qualifying years consist of school years in which the employee was in paid status for at least 800 hours and was eligible for District-paid insurance coverage. The following shall not count toward, but shall not

constitute a break in the service requirement: (a) time spent on authorized leave of absence and, (b) any time intervening between resignation and reinstatement with full benefits within thirty (39) months of the last day of paid service. The employee must meet the following requirements:

- a. For employees hired prior to March 11, 1984, five (5) consecutive years of qualifying service immediately prior to retirement shall be required in order to qualify for retiree health benefits for the life of the retiree.
- b. For employees hired on or after March 11, 1984, but prior to July 1, 1987, (ten) 10 consecutive years of qualifying service immediately prior to retirement shall be required in order to qualify for retiree health benefits for the life of the retiree.
- c. For employees hired on or after July 1, 1987, but prior to June 1, 1992, fifteen (15) consecutive years of qualifying service immediately prior to retirement or ten (10) consecutive years immediately prior to retirement plus an additional ten (10) years which are not consecutive shall be required in order to qualify for retiree health benefits for the life of the retiree
- d. For employees hired on or after June 1, 1992, but prior to March 1, 2007, years of qualifying service and age must total at least eighty (80) in order to qualify for retiree health benefits for the life of the retiree. For employees who have a break in service, this must include ten (10) consecutive years immediately prior to retirement.
- e. Employees hired on or after March 1, 2007, but prior to April 1, 2009, shall be required to have a minimum of fifteen (15) consecutive years of service with the District immediately prior to retirement, in concert with the "Rule of 80" eligibility requirement (section 4.0 (d) above) to receive employee and dependents' health and welfare benefits (medical dental and vision) upon retirement as provided for in this Agreement.
- f. For employees hired on or after April 1, 2009, years of qualifying service and age must total at least eighty-five (85) in order to qualify for retiree health benefits. This must include a minimum of twenty-five (25) consecutive years of service with the District immediately prior to retirement.
- g. In order to maintain coverage, the retiree must continue to receive a PERS/STRS allowance and must enroll in those parts of Medicare for which eligible.
- h. Employees on "Continuation of Enrollment" pursuant to Section 7.0 below shall, if otherwise qualifying under this section, be eligible for

coverage under the District paid insurance plans upon receiving a PERS/STRS retirement allowance.

- 5.0 Enrollment: For the hospital-medical, dental and vision care plans, an unenrolled employee eligible for enrollment may submit application for enrollment in a plan at any time. However, an employee who has previously been enrolled in a plan during the current enrollment year must, upon re-enrollment in that same enrollment year, select the same plan. Such an employee must wait until the next open enrollment period to effect a change of plans. The District shall process applications so as to make coverage effective on the earliest practicable date consistent with the plan provisions, and in no case shall this be later than the first (1st) day of the calendar month following the receipt of the completed application.
- 5.1 Eligible dependents may be enrolled by the employee in the hospital-medical, dental, and vision care plans at any time provided the eligible employee submits a "Request for Change of Dependent Status" form and proof of eligible status as described below.

Newborn children of the employee are automatically covered for the first thirty days following birth, provided that an application for dependent coverage is received by Benefits Administration before the end of the 30-day period.

a. <u>Documentary Proof of Status Required for Dependents</u>

<u>Dependents</u>	Documents Required (copy)
Legal Spouse	State-or County-Issued Marriage Certificate
Domestic Partner	Notarized "Declaration of Domestic Partnership" At least two of the documents listed in Section 5.1, b. (9) below
Child, to age 26*	Birth Certificate (in case of newborn, evidence of birth until birth certificate is available)
Stepchild, to age 26	Birth Certificate and income tax return showing dependent status
Adopted Child, to age 26	Adoption papers

Child who is a Legal Court order establishing Ward to age 26 legal guardianship

NOTE: The children of a domestic partner are **not** eligible for coverage unless they have been adopted by the employee or the employee is the legal guardian. In such cases, the required documentation for adoption or legal guardianship must be provided.

b. A domestic partner of the same or opposite sex of an eligible employee may be covered as a dependent if all of the following criteria are met.

The employee and his/her partner:

- (1) have shared a regular and permanent residence for the past twelve (12) months immediately preceding the application for coverage with the LAUSD;
- (2) are engaged in an exclusive, committed relationship for mutual support and benefit to the same extent as married persons and intend to stay together indefinitely;
- (3) are jointly responsible to each other for basic living expenses; basic living expenses are defined as the expenses supporting daily living, i.e., shelter, food, clothing (contributions need not be equal);

- (4) are not currently married to another person;
- (5) have not signed a declaration of a domestic partnership with another individual in the previous twelve (12) month period;
- (6) are at least eighteen (18) years of age;
- (7) are not blood relatives any closer than would prohibit legal marriage in the state of residence;
- (8) are mentally competent to consent to a contract;
- (9) are financially interdependent as proven by providing at least two (2) of the following documents: common

^{*}All references to age 26 in this Article is intended to comply with the Patient Protection and Affordable Care Act dated March 23, 2010.

ownership of real property or a common leasehold interest in real property; common ownership of a motor vehicle; joint bank account or joint credit account; designation as a beneficiary for life insurance or retirement benefits.

- c. No other dependents or family members are eligible for coverage, except that disabled children who meet the disability standards of the plan(s) and who have been enrolled prior to age twenty-six (26) or, who were first enrolled as eligible full-time students prior to the disabling condition, may continue to be covered beyond age twenty-six (26).
- d. If spouses/domestic partners are both District employees and each is covered both as an employee and as a dependent, the District will pay \$1000 to the dependent per coverage year who agrees to accept coverage under the same plan as his/her spouse/domestic partner, thereby creating coverage for one as the employee and one as the dependent.
- e. If a District employee agrees to waive coverage from the District and accepts coverage solely under a plan of his/her spouse's/domestic partner's employer (not the District), the District will pay \$1000 to the employee, for each coverage year waived.
- 5.2 It is the responsibility of the employee to notify Benefits Administration immediately regarding the termination of his/her domestic partner relationship. The employee must submit LAUSD Form "Request for Change of Dependent Status". The coverage for a domestic partner shall end on the last day of the month in which the relationship and/or living arrangement terminates and/or for which either party is no longer eligible for coverage.
- 5.3 For an employee whose spouse/domestic partner has other health insurance coverage, reimbursement will be limited to the maximum percentage allowed by the primary health plan. An employee whose spouse/domestic partner is also a District employee will not be covered as both an employee and as a dependent within the same plan. A married couple who both work for the district or domestic partners who both work for the District may include their qualifying children on their individual policies, but such children may not be covered more than once within the same plan.
- 5.4 Once each year there shall be an open enrollment period during which an enrolled employee may change hospital-medical benefit plans, dental plans and/or vision care plans. Benefits Administration shall establish and announce the date of said open enrollment period.

6.0 Life Insurance

- 6.1 <u>District-paid Life Insurance:</u> For the District-paid basic life insurance plan, all eligible employees are automatically covered. No application is necessary to obtain this benefit.
- 6.2 <u>Employee-paid Life Insurance</u>: Eligible employees may enroll in the employee-paid life insurance plan without evidence of insurability provided that a completed application is received by the third party administrator of the life insurance plan no later than sixty (60) days from the date the employee is first eligible. Employees not submitting applications during the period specified above may enroll by providing evidence of good health acceptable to the plan. Application for the employee-paid life insurance shall be processed to provide coverage at the earliest date consistent with the plan provided and payroll deduction schedules.

Employees participating in the employee-paid life insurance plan may also purchase spouse, domestic partner and/or dependent children coverage. Dependents eligible pursuant to 5.1 above may be enrolled without evidence of insurability in the following circumstances:

- An application for such coverage is made simultaneously with the employee's initial enrollment.
- The eligible dependents are acquired after the point of initial enrollment by the employee. The application for such enrollment, however, must be received by Benefits Administration within thirty (30) days of the acquisition of such dependent(s).
- Newborn children of the employee are automatically covered for the first thirty days following birth, provided that an application for dependent coverage is received by Benefits Administration before the end of the thirty (30) day period.
- 6.3 <u>Conversion of Life Insurance (District-paid and Employee-paid)</u>
 <u>Enrollment:</u> An employee whose life insurance enrollment terminates because of (a) failure to make direct payments when required, (b) termination of employment, or (c) loss of eligibility, shall be given the opportunity to convert, at the employee's expense, to a permanent form of insurance (other than term insurance) pursuant to the provisions of the plan.

6.4 <u>Continuation of Enrollment (Life Insurance)</u>

a. With respect to the District-paid life insurance plan, coverage for an employee on an unpaid leave of absence other than for illness or industrial injury/illness shall not be provided until such time as the employee returns to

active service in an eligible assignment. Coverage for an employee on an unpaid leave of absence for illness or industrial injury/illness shall continue for one (1) year after which termination of coverage shall be processed and a conversion plan offered upon request.

- b. With respect to the employee-paid life insurance plan, employees who receive no salary or who receive insufficient salary to permit deduction of the required premium after all other deductions are made may continue coverage for a period not to exceed one (1) year by making direct payments of the appropriate premiums by check or money order payable to the plan and sent to the administrator of the life insurance plan.
- 7.0 <u>Continuation of Enrollment (Health Benefits)</u>: With respect to the hospital-medical, dental and vision care plans, if an employee is in an unpaid status and not eligible for District contribution, the employee may arrange for continuance of enrollment under COBRA (see 9.0 9.3 below).
 - 7.1 With respect to employees who decline to make the above continuation payments, coverage shall be terminated and they shall not be eligible to re-enroll in a plan until returning to active service in an eligible assignment and, with respect to the employee-paid life insurance plan, submitting evidence of good health acceptable to the plan.
 - 8.0 <u>Termination of Enrollment</u>: The enrollment of an employee shall terminate:
 - a. For failure of the employee to make payment as provided under Sections 6.3 and 9.0, in which case coverage shall terminate at the close of the month for which the last premium was paid;
 - b. At the request of an employee, in which case coverage shall terminate at the close of the accounting cycle in which the request was submitted;
 - c. Upon termination of employment, in which case coverage shall terminate at the close of the month in which the employment termination was effective; except for District paid life insurance in which case coverage shall terminate on the date the employee ceases to be employed.
 - d. In the event of the employee's loss of eligibility, in which case coverage shall terminate at the close of the enrollment year, except for the District-paid life insurance plan, which shall terminate coverage on the date of loss of eligibility; and
 - e. For District-paid life insurance, upon the employee's loss of eligibility or termination of employment, in which case coverage shall terminate on the date the employee ceases to be eligible or employed.

- 8.1 With respect to hospital-medical plan coverage, if the employee's participation is terminated at the plan's request for other than non-payment of premium, the employee may enroll in another of the District's hospital and medical plans by making proper application to Benefits Administration.
- 9.0 <u>COBRA</u>: Pursuant to the Consolidated Omnibus Budget Reconciliation Act (COBRA), and comparable State law, eligible employees or dependents may have continuation of coverage for a given period of time at their own expense under the District's health, dental and vision care plans in the event of termination of coverage due to one of the following causes: Death of covered employee, termination of covered employee (under certain conditions) or reduction in covered employee's hours of employment, divorce or legal separation of the covered employee, or a dependent child ceasing to be eligible for coverage as a dependent child under the District's health and welfare plans. In accordance with COBRA regulations, domestic partners are not considered qualified beneficiaries and are ineligible for COBRA continuation coverage.
 - 9.1 The monthly premium for continued coverage shall be determined at the time of eligibility and shall be subject to change; however, the premium charged to employees will not exceed one hundred (100%) percent of the premium paid by the District plus the amount allowed by law for employees and/or dependents in a comparable status. The continuation coverage shall be the same as the coverage available to continuing employees, regardless of the employee's health at the time.
 - 9.2It shall be the responsibility of the employee or the dependent to notify Benefits Administration of a divorce, legal separation or loss of eligibility of a dependent child at the time of such an event. At the time of eligibility for continuation coverage, and upon such notification, an election form shall be provided by the District.
 - 9.3 COBRA shall be administered pursuant to federal law, and all decisions and rules with respect to eligibility, premium costs, qualification for benefits, and level of benefits shall be in accordance with published federal government guidelines. Accordingly, it is expressly understood that all such matters, as well as any other questions or issues relating to COBRA, are excluded from the grievance and arbitration provisions of Article V (Grievance Procedure).

10.0 Miscellaneous Provisions

10.1 If any medical plan premium is refunded by a Plan carrier/administrator, it shall be retained by the District, unless it is the result of a payment made under section 9.0 above by an employee in which case it shall be refunded to the employee. If any injury or illness is caused or alleged to be caused by

any act or omission of a third party, payments will be made according to the terms of the Plan for the services of physicians, hospitals and other providers; however, the Plan Member must reimburse the Plan for any amount paid by the Plan, up to the amount of any settlement or judgment the Member, the Member's estate, parent or legal guardian receives from or on behalf of the third party on account of such injury or illness. The Plan may, in its discretion, condition payment upon execution by the Member, the Member's estate, parent or legal guardian of an agreement; (1) to reimburse the Plan accordingly; and (2) to direct the Member's attorney to make payments directly to the Plan.

10.2 The controlling documents regarding all health plans are the applicable contracts between the District and the carriers/plan administrators. All disputes regarding coverage and benefits are to be resolved under the plan's own grievance procedures rather than under Article V (Grievance Procedure) of this Agreement.

11.0 <u>Employee Assistance Program</u>

a. <u>General</u>: An Employee Assistance Program (EAP) shall be available for employees and dependents. The objectives of the program are to provide confidential, professional counseling and referral services for a wide range of employee concerns including but not limited to: personal, marital and family problems; psychological and emotional problems; alcohol and substance abuse and dependency, or problems arising out of financial or legal matters.

The program is designed to provide employees with the information, resources and opportunities to resolve personal, family and work problems before job performance is affected and to assist employees in correcting problems contributing to substandard performance. The program is not intended to supplant other forms of assistance or medical referrals currently permitted under this Agreement, Board Rules or the law.

- b. <u>Confidentiality</u>: The program is to be based upon strictest confidentiality and privacy, so that appropriate assistance can be offered by the District and/or sought by employees without adverse effect upon the employment relationship. EAP personnel shall not divulge to the District information relating to the decision of any individual employee to seek or decline EAP assistance, referrals or follow-up treatment. Neither the employee nor the District may refer to the following matters in any evaluation or disciplinary action, or appeal:
 - (1) The decision of employees to utilize EAP services, or not to do so or actual participation in the EAP;
 - (2) The recommendation by anyone, including

Article XV - Health and Welfare

administration, that an employee should utilize EAP services, or the failure to make such a recommendation.

The existence of the EAP shall stand as conclusive evidence that the District has offered appropriate assistance for any personal problems which may have a bearing upon job performance of employees eligible for EAP and which fall under the purview of EAP.

- c. <u>Employee Accountability</u>: Employees remain personally accountable for their job performance. The existence and/or utilization of the EAP and other services shall not serve as an excuse for inadequate job performance or as a defense in any evaluation or disciplinary action.
- 12.0 Eligible unit employees may enroll in the IRS 125 Flexible Spending Account program.
- 13.0 <u>State Disability Insurance</u>: For unit employees not currently eligible for disability insurance, coverage shall commence as soon as administratively practicable following certification from that a majority of such affected employees in the bargaining unit have voted to participate in the program. Upon district receipt of such certification, the following shall apply:
 - a. The District agrees that affected unit employees shall be enrolled in the Disability Insurance Program for public school employees administered by the Employment Development Department of the State of California and that all premium costs of this Program shall be borne by the employees through individual payroll deductions.
 - b. agrees that the Disability Insurance Program is administered by the Employment Development Department of the State of California and that all decisions and rules with respect to eligibility, premium costs, qualifications for benefits, level of benefits, and the administration of the program is the responsibility of the Employment Development Department. Accordingly, it is expressly understood that all such matters, as well as any other questions or issues relating to Disability Insurance of the Employment Development department are excluded from the grievance and arbitration provisions of Article V (Grievance Procedure).
 - c. In order to implement the Disability Insurance Program specified in Section 13.0 above, the District at its sole discretion may enter into and unilaterally may amend, alter, or modify any contract or contracts with the Employment Development Department for Disability Insurance coverage.

ARTICLE XVI

HOLIDAYS

1.0 <u>Holidays</u>: An employee in a regular assignment or in an assignment in lieu of his/her regular assignment shall receive holiday pay for those holidays listed below and for other holidays declared by the Board of Education, the Governor of California, or the President of the United States which come within the employee's assignment period, subject to the conditions listed in "a" through "c".

January 1	New Year's Day*
That date in January declared by the	
Board	. Martin Luther King, Jr. Day*
Third Monday in February	Presidents Day*
Last Monday in May	Memorial Day*
July 4	Independence Day
That date declared by the Board	Admission Day
First Monday in September	Labor Day
November 11	Veterans Day*
That Thursday in November	
proclaimed by the President	Thanksgiving Day*
Day following Thanksgiving Day	Thanksgiving Friday*
December 25	Christmas Day*
That date declared by the	
BoardAlter	nate Lincoln Day Observance

- a. It is recognized by the parties that employees in this unit who are assigned to year-round schools may not receive all of the foregoing holidays on the days specified depending upon the particular calendar for each year-round school. Such employees shall, however, receive the same number of holidays according to their assignment basis as other employees in the same classification and on the same assignment basis.
- b. The employee must have been in paid status for a portion of the working day immediately preceding or succeeding the holiday, provided that an employee on a military leave of absence entitled to compensation under Article XIII (Leaves of Absence) shall only receive pay for the portion of the holiday period needed to meet the total time for which compensation is required by law.
- c. An employee whose regular work schedule is less than five (5) days per week and forty (40) hours per week shall not be entitled to pay for any holiday observed on the employee's regularly scheduled day off.
- d. An employee in paid status during any portion of the working day of his/her normal assignment immediately preceding or succeeding

Article XVI - Holidays

the school holiday of December 25 and January 1 shall receive pay for the two (2) holidays.

- 2.0 Friday shall be the observed holiday for all purposes for holidays which fall on a Saturday; Monday shall be the observed holiday for all purposes for holidays which fall on a Sunday.
- 3.0 If a holiday occurs while an employee is on vacation or other paid leave, that day will be credited and paid as a holiday.

^{*}Employees assigned to C Basis would be eligible for all of the holidays listed above which are indicated by an asterisk(*)

ARTICLE XVII

VACATION

- 1.0 An employee shall earn vacation for active service in a regular assignment or in an assignment in the same or another class in lieu of the employee's regular assignment in accordance with Section 1.1. Active service means all of the time for which pay is received.
- 1.1 Accrual of vacation shall be determined based on the factors and in the manner set forth in the following table:

Employee's <u>Years of Service</u>	Vacation Accrual Factor Based on 40 Hour Workwee	<u>!k</u>			
Less than 4 years	.03846				
4 or more years					
but less than 15	.05770		Employee's		
15 years but less			Hours of		Employee's
than 16	.06155	Χ	Paid Status	=	Hours of
16 years but less			Exclusive of		Accrued
than 17	.06539		Overtime		Vacation
17 years but less					
than 18	.06923				
18 years but less					
than 19	.07308				
19 years or more	.07693				

For example, a full-time twelve (12) month employee will accrue vacation annually as follows:

1 through 4 years	. 10 days
5 through 15 years	. 15 days
16 years	16 days
17 years	17 days
18 years	18 days
19 years	19 days
20 years or more	. 20 days

1.2 The vacation accrual factor for employees assigned a regular workweek of less than forty (40) hours during the first four (4) years of service shall be:

37.5 hours but less than 40 hours	.04087
35 hours but less than 37.5	.04379
Less than 35 hours	.03846

During subsequent years of service vacation accrual shall be at the rate of the forty (40) hour workweek above.

- 1.3 A "year of service" for the purpose of this Article shall be defined as paid service in regular status for one hundred thirty (130) days or more within the fiscal year, including time served in probationary or permanent certificated service; however, total assignment hours annually shall not exceed two thousand and eighty (2080) hours for years of service credit.
- 1.4 No employee shall be permitted to accrue vacation in an amount greater than that which the employee earns in eighteen (18) pay periods (the employee's "vacation cap amount"). Once the employee has accrued vacation in an amount equal to the employee's vacation cap amount, the employee shall cease to accrue vacation until the employee uses vacation in an amount sufficient to reduce the employee's accumulated vacation balance below the employee's vacation cap amount. All appropriate adjustments shall be made annually at the end of each fiscal year so that the employee's earned vacation balance carried forward to the next fiscal year shall not exceed the employee's "vacation cap amount."
- 1.5 Consistent with the eighteen (18) pay period vacation cap amount set forth in Section 1.4 above, the following procedure for scheduling of vacation time shall be in effect:
 - a. Step One: By March 15 of each school year, administrators shall issue an annual vacation calendar for the next school year. The calendar will include the following:
 - (1) A list of all dates when vacation cannot be taken due to operational needs.
 - (2) A list of all dates when vacation may be taken by all employees.
 - (3) A list of all dates when a part of the staff may take vacation indicating any limits on the number of employees who may take vacation or on the amount of vacation taken.
 - b. Step Two: By April 15 of each school year or two weeks after the adoption of school calendars, whichever occurs later, each employee shall provide to his/her appropriate administrator or designee a proposed written vacation usage schedule for the following school year, which schedules vacation for the school year in amount necessary to assure the employee will not exceed the vacation cap amount.

The vacation days identified in the employee-submitted vacation schedule shall be scheduled in a manner consistent with the provisions of Section 1.6 through Section 1.9 below. Unless otherwise directed by the employee's appropriate administrator, the proposed vacation schedule for B, C, and E basis employees shall include the appropriate winter and spring recess days within the employee's basis, and for A and M basis employees, up to five (5) days during winter recess.

1.6

- Within fifteen (15) calendar days of receipt of the employee's vacation usage schedule, the appropriate administrator shall provide a written acknowledgment either approving the employee's submitted vacation usage schedule for the following school year, or disapproving the submitted schedule and providing a basis in writing for that denial. Timely submitted vacation schedules shall not be denied for reasons other than workload, scheduling conflicts or where the proposed schedule for vacation would substantially interfere with the operation of the employee's work unit. Changes in preapproved vacation schedules will not be made by the District except for critical operational necessity or an emergency that would substantially interfere with the operation of the employee's work unit. Vacations in progress shall not be canceled for reasons other than a declared state of emergency. Except as provided in Section 1.8, any scheduling conflict(s) between or among employees working in the same unit or office as to when vacation can be taken shall be decided by site or work unit seniority within classification. In the event of a tie, the scheduling conflict shall be determined by lot.
- b. An employee whose previously approved vacation has been changed due to a critical operational necessity shall have the right, prior to filing a formal grievance pursuant to the grievance procedure, to meet with the employee's appropriate administrator and the appropriate Division Head or designee to attempt to informally resolve the appropriateness of the vacation change. The meeting shall occur and the decision of the Division Head or designee shall be provided within five (5) days of the employee's request for the meeting. Nothing herein shall alter the fifteen (15) day time limit for filing a written grievance as required by Article V, (Grievance Procedure).

1.7

- a. Once an employee's vacation schedule is submitted and approved pursuant to the above, no change can be made by the employee without submission of an alternate vacation schedule for the date(s) in question. The requested modification(s) shall not be unreasonably denied.
- b. An employee who has submitted a vacation schedule and received approval for that schedule, and who is thereafter denied the right to take that vacation because of an extension of the time when vacation is precluded pursuant to Section 1.5 a. (1) above, shall be entitled to submit an alternate vacation schedule, which shall not be unreasonably denied.

- c. An employee assigned to a school site operating on a traditional/LEARN calendar and who submits and receives approval of the employee's vacation request, shall be entitled to resubmit a vacation request if the school site thereafter changes to a multi-track calendar and as a result the employee's previously approved vacation is disapproved. The employee's resubmitted vacation request shall not be unreasonably denied.
- 1.8 An employee who is prevented or prohibited from taking vacation previously approved by the employee's appropriate administrator shall be permitted to exceed by that amount the vacation cap amount for the school year in question, and shall be granted a preference the following year in scheduling vacation so as to assure the employee's ability to schedule sufficient vacation to reduce the employee's vacation accumulation below the vacation cap amount. However, such relief from the vacation cap amount must first be pre-approved in writing by the Superintendent or designee.
- 1.9 The District shall be permitted (but not required) to schedule and require employees to take vacation under the following circumstances:
 - a. When the employee fails to provide an annual vacation schedule per Section 1.5 above;
 - b. When the employee has accrued vacation in an amount equal to or greater than the vacation cap amount, as provided in Section 1.8 above;
 - c. When the employee is sent home pending the results of a disciplinary investigation (with the vacation used to be restored to the employee's vacation balance if the investigation does not lead to discipline); and
 - d. During periods within the employee's assignment basis when the District is closed, when the employee's work site is closed, or when there is a lack of work (unless the employee and the appropriate administrator agree that the employee may go unpaid during such a period). Where assignment of mandatory vacation is necessary due to lack of work as determined by management, volunteers shall be considered first; preference shall be given to employees in the affected classification at the site with the highest site seniority in classification. If the number of volunteers is insufficient, assignment of mandatory vacation shall be to those employees with the highest vacation balance. Exception to the foregoing may be made considering special needs, attendance records, the individual employee's vacation balance, and/or previously approved scheduled vacation.
- 1.10 Vacation may be interrupted or terminated in order to begin illness leave, bereavement leave, jury duty leave or military leave.
 - 1.11 Except as set forth in Section 1.13 below, in computing pay for

vacation, all applicable salary differentials shall be included and vacation shall be paid at the base salary rate in effect at the time the vacation is taken.

1.12 When a regular employee whose assignment is other than A Basis is given an off-basis assignment, that employee shall earn vacation in accordance with the schedule set forth in Section 1.1 above.

1.13

- a. Notwithstanding the foregoing provisions and in order to facilitate a complete transition from an unlimited vacation accrual system to the above-described eighteen (18) pay period vacation cap system, the District shall, for each employee employed by the District as of the adoption of this agreement by the Board of Education, calculate the employee's total accrued vacation as of June 30, 1994 (the "1994 accrual bank"). The District will then credit each employee with their 1994 accrual bank as vested vacation to be paid out at the time the employee separates from the District, but at the employee's salary rate in effect as of June 30, 1995.
- b. In order to encourage employees to draw from their 1994 accrual bank (and thereby reduce the District's current unfunded vacation liability), should an employee utilize any vacation from their 1994 accrual bank during the employee's employment with the District including vacation hours used during 1994-95, that vacation shall be paid out at the employee's current salary rate at the time the vacation is utilized and deducted from the 1994 accrual bank annually at the end of the fiscal year.
- c. The amount of vacation from the 1994 accrual bank which may be utilized by an employee during any school year shall be limited to twenty (20) days. This limitation shall include vacation used in lieu of half-pay illness days pursuant to Article XIII (Leaves), Section 11.5. Exceptions may be made at the sole discretion of the District, but must be pre-approved in writing by the Superintendent or designee.
- 1.14 Except as set forth in Section 1.13 above with respect to employees' 1994 accrual bank, on separation from service, the dollar value of the employee's vacation balance shall be paid as a lump sum at the employee's salary rate at the time of such separation (pursuant to Section 1.11 above).

1.15 <u>Attendance Incentive Plan</u>

a. A vacation-earning employee who accumulates a total of fifty (50) days or more days of full-pay illness absence credit earned subsequent to June 30, 1995 shall, on a one-time basis as of June 30 of the school year in which he or she accumulated those fifty (50) days, be credited with two (2) additional days of vacation. An employee, whose full pay illness absence credit earned subsequent to June 30, 1995, thereafter drops below fifty (50) or more days shall

not be entitled to additional vacation under this section, except pursuant to paragraph (b).

- b. Each additional increment of twenty-five (25) days of unused full-pay illness absence credit beyond fifty (50) days and earned subsequent to June 30, 1995, shall entitle the employee to one (1) additional vacation day.
- c. At the option of the appropriate Division Head, B, C, and E basis employees may be paid for the additional days of vacation earned in Sections 1.15(a) and 1.15(b) on the basis of their daily rate of pay during the preceding fiscal year.

ARTICLE XVIII

SAFETY CONDITIONS

- 1.0 The responsibility for providing for safe working conditions that are in conformance with applicable law and which are within fiscal constraints shall be the District's. Employees shall be responsible for complying with safety procedures and practices and for reporting any unsafe condition, facility, or equipment of which he/she is aware. The District shall be responsible for informing employees of necessary safety procedures and practices. There shall be no reprisal against an employee for reporting any real or potentially unsafe condition, facility, or equipment.
- 2.0 Unit members shall be allowed reasonable access and use of staff restrooms and toilet facilities and drinking water.
- 3.0 <u>Emergency Use of Telephone</u>: Except in cases of emergency, employees shall not use District telephones for personal calls. At work locations where no public telephone is available, the District shall make a reasonable effort to provide access to District telephone for emergency calls. Charges incurred for any such personal calls shall be collected from the employee and remitted to the Accounting and Disbursements Division.
- 4.0 All newly purchased new school buses shall be equipped with tinted glass.
- 5.0 The District shall continue to observe legal environmental regulations regarding water and air pollution at bus yards.
- 6.0 The District shall endeavor to provide any needed supplemental training in safety procedures to employees at the Newman Nutrition Center and the Manufacturing Kitchen in the disassembly of machines for cleaning.
- 7.0 No bargaining unit member shall be required to transport hazardous material in his/her personal vehicle in violation of Cal/OSHA regulations.

ARTICLE XIX

TOOL REPLACEMENT

- 1.0 Subject to Section 3.0 below, the District will repair or replace (or pay the cost of repairing or replacing) employees' hand tools which are broken or damaged in District service or lost through verified theft from District property not the result of the employee's negligence. Such hand tools must be those the employee is required to possess by the District as specified in writing (tools of the trade) and must be listed on approved inventory lists. All employees required to supply their own hand tools shall be given an inventory list form which will include value, and information regarding hand tools as indicated in this Article. Any broken or damaged hand tools replaced with comparable valued hand tools shall become the property of the District.
- 2.0 It is understood that employees will be responsible for taking reasonable care of their tools and that they will be held responsible for carelessness, neglect, and misuse. The District reserves the right to review cases where there are repeated or high frequency claims for hand tools.
- 3.0 If employees' hand tools are damaged beyond repair or lost through verified theft, the replacement value of the tools, as determined at the time of damage or theft (including a normal allowance for depreciation) shall be paid, subject to a one hundred dollars (\$100) deductible. Claims of less than one hundred dollars (\$100) shall not be processed. The maximum payment approved by a division head for any one loss shall not exceed five hundred dollars (\$500). Losses in excess of five hundred dollars (\$500) but not to exceed three thousand dollars (\$3,000) shall be reimbursed only with the prior approval of the branch head. A written request for reimbursement for damage to or theft of tools shall be filed by the employee with the Insurance Section of the Contract and Insurance Services Branch within sixty (60) calendar days of the date of loss and shall be signed by the employee's immediate supervisor and the branch head. The Insurance Section shall conduct such investigation as may be necessary. Reimbursement is provided only when approval for the use of tools was given before the tools were brought to the school or office or other District property and only when the value of the tools was agreed upon by the employee and the supervisor and approved by the branch head on the inventory list. If either the deductible or the maximum reimbursement amount described above changes for bargaining unit members supervised by members of this bargaining unit, then that change shall also be implemented for employees in this unit.
- 4.0 Where the claim involves a theft, the employee must submit a police report and include the DR number in the claim.
- 5.0 The employee must assign to the District the right of subrogation to the extent of any payment made by the District.

ARTICLE XX

TUITION REIMBURSEMENT

- 1.0 <u>Tuition Reimbursement</u>: The District may grant tuition reimbursement to permanent Unit employees under the conditions specified below:
 - a. Programs eligible for reimbursement shall include, but not be limited to, courses of study at approved academic institutions, seminars and training institutes conducted by recognized professional associations, conferences, meetings and such other training programs designed to upgrade the classified service or encourage retraining of employees who may otherwise be subject to layoff as the result of technological change.
 - b. Costs that may be reimbursed are tuition, other mandatory fees, books, and other training materials that are required for the specific course. Traveling expenses, parking fees, student body fees, the cost of paper, pens, notebooks, equipment, and other costs shall not be reimbursed.
 - c. Approval for reimbursement shall be obtained on the appropriate form signed by Division head or designee before the commencement of the course or program. Approval shall be at the sole discretion of the District. If a request for reimbursement is not approved, the employee shall be entitled upon request to know the reasons for the disapproval.
 - d. The District encourages its employees to obtain work- specific degrees and additional professional certifications for the purpose of increasing the employee's knowledge, understanding and skills as related to the employee's employment by the District. The course(s) or program must be directly related to the employee's current or future service within the District and for such purpose.
 - e. The course(s) or program shall not be taken during the employee's assigned duty hours.
 - f. Reimbursement shall be made as soon as practicable following presentation of official receipts and satisfactory evidence of successful completion of the approved course(s) or program. If grades are received, successful completion shall be defined as a grade of C or passing.
 - g. Effective July 1, 2008, tuition reimbursement shall be limited to a maximum of seven hundred dollars (\$700) for any individual employee during any twelve (12) month period.
 - h. The course(s) or program for which tuition reimbursement is requested shall be completed within the period for which it was approved, or the employee must submit a new request.

Article XX - Tuition Reimbursement

- 1.1 Provisions of this Section shall not apply to any employee eligible for reimbursement by any other governmental agency, organization or association.
- 1.2 An employee who terminates employment with the District within six (6) months of receiving tuition reimbursement pursuant to this Section, shall refund the amount of the reimbursement to the District, or it shall be deducted from the employee's final warrant. This requirement shall be waived in the event of the employee's death or physical or mental disability which precludes the employee from returning to District employment.

ARTICLE XXI

RECLASSIFICATION PROCEDURE

- 1.0 For the purpose of this article, "reclassification" means the upgrading of a position to a higher classification as a result of the gradual increase of duties being performed by the incumbent in such a position.
 [see Education Code §45101 (f).]
- 2.0 <u>Request for Reclassification</u>: An employee may initiate a request for a classification study regarding whether or not to reclassify his/her position by completing the Position Description (80.4) Form which may be obtained from the Personnel Commission office.
- 3.0 Approvals: Provided the description of the duties is accurate, necessary approval signatures shall be accomplished by the unit or section head, Branch head and Division head/Local District Superintendent within sixty (60) calendar days after submission to the employee's immediate administrator. Extension of this time limit shall only be made upon consultation with the employee. The required approvals certify that the duties listed on the 80.4 Form are assigned the subject position and do not necessarily imply that the position is misclassified. The responsible administrator may attach supplemental information to the Form before signing. If, after review by the responsible administrator, the employee does not agree that the approved 80.4 Form adequately reflects the assigned duties of the subject position, the employee may attach supplemental information to the Form detailing concerns. The responsible administrator shall notify the employee upon disposition of the request.
- 4.0 The request shall be forwarded to the Personnel Commission staff for classification study. The employee shall be informed of the disposition of the request by the Personnel Commission.
- 5.0 In conjunction with any request by the District to the Personnel Commission to abolish one or more classifications and replace the abolished classification(s) with one or more new or existing classifications, the District shall concurrently provide the Union with notice of such a request.

ARTICLE XXII

JOB STEWARDS

- 1.0 <u>Job Stewards</u>: At each work location, the Union will have the right to designate, pursuant to its own procedures, one employee per shift to serve as the Job Steward for that work location. The Union shall inform the Office of Staff Relations in writing of each employee so designated. The Job Steward shall have the right to:
 - a. Represent an employee at that work location, upon request, in a grievance meeting as expressly provided for in Article V (Grievance Procedure), Section 2.1 and Article X (Evaluation Procedures) Section 3.0. Upon request of a Job Steward serving as a representative described above, the responsible administrator shall, whenever operationally practical, reschedule a lunch and/or rest period of the employee and Job Steward so that they may timely confer regarding the pending grievance on appeal.
 - b. On his/her own time (outside of duty hours) to coordinate Union meetings, which may be held on the work site during unpaid time for any employee in attendance, subject to availability of facilities and provided that there is no interference with other scheduled duties or events;
 - c. Post, initial, and date official Union notices on a bulletin board space designated by the site administrator for such purposes; and
 - d. Report to the appropriate administrator upon discovery and without delay any unsafe or unsanitary conditions at the work site.
 - e. On his/her own time (outside of duty hours) represent an employee at another work location at Step One of the Grievance Procedure at Article V, Section 8.0. Thereafter, either that Job Steward may continue to represent that employee at Steps Two and Three and at the Arbitration as set forth in Article V, Sections 11.0 thru 13.0 (and pursuant to the provisions of Article V, Section 3.0) or the employee may be represented by another Union representative, who shall provide trial representation without any cost to the District and outside of that representative's District duty time if any.

ARTICLE XXIII

ENTIRE AGREEMENT

- 1.0 The Union agrees that this Agreement is intended to cover all matters relating to wages, hours and all other terms and conditions of employment and that during the term of the Agreement, neither the District nor the Union will be required to meet and negotiate on any further matters affecting these or any other subjects not specifically set forth in this Agreement, even though such subjects or matters may not have been within the knowledge or contemplation of either or both the District or the Union at the time they met and negotiated on and executed this Agreement, or even though such subjects or matters were proposed and later withdrawn. Nothing herein is intended to prevent the parties from meeting and negotiating during the term of this Agreement pursuant to mutual consent.
- 2.0 To the extent any term of this Agreement conflicts with any valid provision of the California Education Code, the Education Code provision shall prevail.
- 3.0 Notwithstanding the foregoing, if the subject matter in a Personnel Commission Rule falls within the scope of representation, as defined in California Government Code Section 3543.2, that rule shall be applicable as follows:
 - a. If this Agreement deals with the subject matter of a provision of a Personnel Commission rule, then that provision shall not apply to employees in this unit;
 - b. If the Personnel Commission rule provides for a nonprocedural benefit, and this Agreement does not provide for the benefit, the benefit shall not be available to employees in this unit, unless required by law;
 - c. If the Personnel Commission rule prescribes a procedure and this Agreement does not, the rule shall apply to employees in this unit;
 - d. Where a conflict exists between a Personnel Commission rule and an express provision of this Agreement, then this Agreement shall prevail; or
 - e. In the event that a Personnel Commission rule specifically cited in this Agreement is modified by the Personnel Commission during the term of this Agreement, the applicability of such modification to the unit will be subject to agreement between the Union and the District. If an agreement is not reached, the language of the unmodified Personnel Commission Rule shall remain in effect for the term of this Agreement.

ARTICLE XXIV

TERM OF AGREEMENT

- 1.0 <u>Term</u>: This Agreement shall become effective upon adoption by the Board of Education and shall remain in full force and effect, pursuant to its terms, to and including June 30, 201<u>7</u>, and thereafter extended on a day-to-day basis until canceled by either party upon ten (10) days' written notice.
- 2.0 <u>Negotiations for Successor Agreement</u>: Negotiations for a successor Agreement shall commence upon request of either the District or the Union at any time after January 1, 2017. Such negotiations shall commence as soon as reasonably practicable after a request to commence negotiations is made, but in no event more than ten (10) days from the date of the request, absent mutual agreement to extend this time.

APPENDIX A

UNIT S MEMORANDUM OF UNDERSTANDING 2014-2017

This Memorandum of Understanding (MOU) is made and entered into this 3rd day of December, 2014 by and between the Board of Education of the Los Angeles Unified School District ("District") and the Teamsters Local 572 ("Teamsters") for employees in Unit S (Classified Supervisors).

Pursuant to Section D of the parties' 2014-2017 Agreement previously adopted by the Board on July 1, 2014, the District and Teamsters have met and negotiated in good faith and have completed their negotiations on items previously sunshined. This MOU is a supplement to that Agreement. The term of that Agreement and this supplement shall cover a period through June 30, 2017 (and continued thereafter on a day-to-day basis until such time as it may be terminated by either party upon 10 days' notice).

- A. The parties' 2014-2017 Agreement shall be further modified as follows:
 - 1. Article VII Non-Discrimination 1.0
 - 2. Article X Evaluation Procedures (New 1.1)
 - 3. Article XIV Wages and Salaries, Pay Allowances, Differentials and Special Salary Practices 2.2
 - 4. Article XXIV Respectful Treatment (New)
 - 5. Appendix F Food Services Addendum
 - 6. Payroll Optimization The Union agrees to the implementation of a semi-monthly pay system, effective July 1, 2017.
- B. The parties agree to continue discussions on the topic of the 2013-2014 Lump-Sum Payments previously made to bargaining unit members. Additionally, the District acknowledges the Teamsters concern regarding clerical supports provided to schools.

//	
//	
//	
//	
//	
//	
//	
//	

LAUSD Board of Education.	
Date of agreement: 121314	
Los Angeles Unified School District	Teamsters Local 572
By: <u>Junan</u> <u>Gudy</u> an Vivian Ekchian Office of Labor Relations	By: R. E. Middleton Secretary/Treasurer
	By: Adriana Salazar Avila Business Representative
Adopted and approved by the Board of Education on	
By: Richard Vladovic, Ed. D. President	

The above is subject to ratification by the Teamsters membership of Unit S and to final approval by the

APPENDIX B

TRANSPORTATION ADDENDUM

- Bus Supervisors to Drive a School Bus -- While it is the District's general policy that Area Bus Supervisors and Assistant Area Bus Supervisors complete renewal training in their assigned regions, exceptions may be granted based upon the work load of the Driver Trainers in the regions, feasibility, space availability and efficiency to the District. Requests for exceptions must be approved by the Regional Transportation Managers, or appropriate designees, of both of the affected regions. Such requests must be submitted with reasonable notice, not fewer than ten (10) working days prior to the first scheduled day of the training. No such request shall be denied based upon an arbitrary or capricious reason. In accordance with current practice, renewing of documents shall be during scheduled classes and within assigned hours, but in no event shall classes or testing be the cause of overtime to the employee. Nor shall such classes or testing cause the splitting of assigned hours beyond a one (1) hour duty-free unpaid period.
- 2.0 <u>Internal Transfers</u> -- When the District makes determinations regarding transfer requests within the Transportation Branch, seniority will be considered. For the purposes of identification for transfer, "seniority" means length of service in the classification. All other factors being equal, seniority will prevail.

Transfer requests, including inter-regional transfer requests, shall not be denied for arbitrary or capricious reasons. When the District denies any internal transfer request within the Transportation Branch, the employee shall upon written request be informed in writing by the manager or supervisor making the decision of the reason for the denial.

- 3.0 <u>Work Schedules</u> -- Area Bus Supervisors may submit a request to the Regional Transportation Manager, or appropriate designee, for a change in work schedule. No such request shall be denied based upon an arbitrary or capricious basis.
- 4.0 <u>Office Space</u> -- The District will make reasonable effort to provide adequate office space and materials when available.
- 5.0 <u>New Technology, Equipment, and Vehicles</u> -- New equipment and technology shall be deployed by the branch based on the funding source and operational needs of the branch and the community. When all other factors are equal with respect to such a deployment, seniority will be considered.
- 6.0 <u>Overtime</u> --Upon request, transportation supervisors may be provided with the most recent copy of an official payroll printout showing hours and overtime by region and by job classifications. Supervisors may be provided with the printout for their own classification. Overtime shall be assigned based upon the operational needs of the region and the area, as determined by the Regional Transportation Manager, or appropriate designee. Upon request, the Union shall be provided with the most recent copy of an official payroll printout showing

Appendix B - Transportation Addendum

hours and overtime for a region or the branch.

- 7.0 <u>Attendance Policy</u> -- The Transportation Branch shall institute the same attendance policy for all branch supervisors.
- 8.0 A bargaining unit employee within the Transportation Branch may submit a single transfer request for one or more identified locations, or regions or for all openings in the relevant classification within the branch.
- 9.0 <u>Assistant Area Bus Supervisor Bus Assignment Bids</u> Any employee newly assigned as an Assistant Area Bus Supervisor must bid an appropriate bus at the next annual bid conducted for Bus Drivers.

All non-newly assigned Assistant Area Bus Supervisors shall be permitted to re-bid for bus assignment once every three years, beginning Summer 2002, at a time designated by the Branch. Such bidding shall be conducted based upon seniority within the Assistant Area Bus Supervisor job classification. A bus appropriate to the employee's assignment, from a pool of buses established by the Transportation Branch, shall be bid. An employee who previously participated in a bid who wishes to re-bid in a subsequent bid shall notify the Branch in writing per procedures established by the Branch.

Notwithstanding the foregoing, adjustments or assignment changes may be made by the Director of Transportation when necessary for the best interests of the District.

APPENDIX C

FLEET MAINTENANCE ADDENDUM

- 1.0 When the District makes determinations regarding transfer requests within the Automotive Maintenance Section, seniority will be considered. For the purposes of identification for transfer, seniority means length of service in the classification. All other factors being equal, seniority will prevail.
- 1.1 Transfer requests shall not be restricted by geographic areas or zones or regions.
- 1.2 If a transfer is denied, the employee may request the reason for denial in writing.

APPENDIX D

SCHOOL ADMINISTRATIVE ASSISTANT ADDENDUM

- 1.0 The direct supervisor of School Administrative Assistants shall be the Principal or his/her designee.
- 2.0 School Administrative Assistants shall be entitled to receive overtime compensation in accordance with Article IX, (Hours of Work). Administrators shall not discipline School Administrative Assistants for requesting compensation for overtime worked either prior to working the overtime or after having received authority to work such overtime.
- 3.0 The District recognizes the importance of communication between the School Administrative Assistants and their direct supervisors. The District shall emphasize to those supervisors the importance of providing School Administrative Assistants with pertinent information to help accomplish their job duties. School Administrative Assistants may also request permission from their direct supervisor to attend a relevant school staff meeting and/or to provide input to their direct supervisor regarding topics to be discussed at such meetings.
- 4.0 School Administrative Assistant's job duties may include overseeing students waiting in the office of the school. School Administrative Assistants who, in conjunction with overseeing students, are placed in an unsafe situation shall be entitled to request assistance from their direct supervisor and/or school security. Requests for such assistance shall not be unreasonably denied.
- 4.1 School Administrative Assistants will provide assistance beyond basic first aid only to the extent that they have voluntarily attended the District's medication administration training or received other appropriate training.
- 5.0 It is the intent of the District that all its employees, including School Administrative Assistants, treat their co-employees and be treated by their co-employees with respect and courtesy in accordance with the Board of Education's unanimous resolution of October 10, 1988.

The Personnel Commission shall write a Memorandum to the Union informing them of the duties of School Administrative Assistants and that supervision by a School Administrative Assistant is not a necessary component of the job classification.

- 6.0 Off-basis Summer Assignments: Summer Assignments for School Administrative Assistants shall be made in the following order:
 - a. Off-basis regular School Administrative Assistants who are selected by the assigned summer school principal;
 - b. Off basis regular School Administrative Assistants who have

Appendix D - School Administrative Assistant Addendum

an application on file with Classified Personnel Assignments Branch. Assignments from this group shall be based on District seniority.

- c. Available persons on the School Administrative Assistant eligibility lists;
- d. Previously assigned substitutes who have qualified by examination; and
 - e. Persons who have never qualified by examination.
- 6.1 When an employee accepts a summer assignment, he/she must complete that assignment for its entire summer program period and should not request vacation or to be changed from one assignment to another. Exceptions may be made at sole discretion of the District.

APPENDIX E

PLANT MANAGERS ADDENDUM

- 1.0 It is the intent of the District to provide reasonable and necessary training which will assist Plant Managers in maintaining the ability to perform their work duties. The District shall also make available to Plant Managers information regarding staff development classes which will assist Plant Managers in the performance of their work duties and/or promotional opportunities.
- 2.0 The site administrators shall, upon request, meet quarterly with the Plant Managers to review pertinent budget items.
- 3.0 Plant Managers shall receive in-service training on the safe transport of gasoline and other potentially hazardous materials. Plant Managers shall not be required to transport more than five (5) gallons of gasoline in their personal vehicles. Plant Managers may only transport gasoline in District-provided and approved containers. Plant Managers shall receive reimbursement promptly for gasoline purchased for District purposes when the purchase has been pre-approved by the site Administrator or designee.
- 4.0 Plant Managers shall not be required to perform herbicide/pesticide spraying which requires special certification or licensing unless they possess said certification or licensing.
- 5.0 Plant Managers shall be entitled to receive overtime compensation in accordance with Article IX (Hours of Work) of this Agreement. Administrators shall not discipline Plant Managers for requesting compensation for overtime worked either prior to working the overtime or after having received authorization to work such overtime.
- 6.0 The Union may initiate a request to the Personnel Commission to study the qualifications required by the District on the class description for the classification of Cluster Facilities Specialist.
- 7.0 Plant Managers shall act as the direct supervisor and evaluator of the Building and Grounds Workers at their sites. The site administrator or designee may act as the reviewer of and provide input to the evaluation of such Building and Grounds Workers.
- 8.0 When the Plant Manager is assigned by the site administrator as the designee to act as the direct supervisor of a Senior Gardener, then that Senior Gardener's position shall be included in the Plant Manager's manpower allotment.
- 9.0 Plant Managers assigned to local Maintenance and Operations (M & O) offices in "Pod" assignments will be allowed to request their "Pod" assignment subject to availability.

Assignments will be based upon a combination of factors:

- 1. Work performance (eligible employees have no Below Standards rating on the current evaluation). In the event there is no current evaluation, it will be assumed the Plant Managers are eligible for the assignment.
- 2. Principal request.
- 3. All factors being equal, seniority will be the tie-breaker.

For "Pod" Assignments:

- 1. Each "Pod" will be clearly defined by"
 - a. Plant Manager I and II assignment
 - b. School site(s)
 - c. Area Operations Supervisor and Operations Program Manager with oversight
 - d. Working Hours

Transfers—All transfers will follow transfer procedures in Article XI. Note that transfers refer to movement between M & O area in contrast to the assignment requests that refer to Pods.

APPENDIX F

FOOD SERVICES ADDENDUM

- 1.0 <u>Training</u> -- It is the intent of the Food Services Division to provide reasonable and necessary training to Food Services supervisors which will assist them in maintaining the ability to perform their duties. The District shall also make available to Food Services supervisors, information regarding staff development classes which will assist Food Services supervisors in the performance of their work duties.
- 2.0 <u>Review of Division Files</u> -- Upon written request to the Director of Food Services, a Food Services supervisor shall be afforded the opportunity to review his or her own division file at a scheduled appointment time.
- 3.0 <u>Input on Hiring</u> Food Service Managers shall, upon request, be entitled to provide input to the relevant decision-makers regarding the hiring of Food Services Division site staff at the site to which the Food Service Manager is assigned.
- 4.0 <u>In-Service</u> -- Food Service Managers who, after completing eight hours of regular duty, are then assigned to conduct formal training on entry-level skills, shall be compensated at the rate of time and a half for the hours worked in such training. Personnel Commission Rule 597 shall continue to apply to Food Service Managers in its current form unless the Personnel Commission changes Rule 597, in which case the parties shall meet and negotiate regarding the effect of such a change.
- 5.0 <u>Transfer Notification</u> -- In the event a Food Services employee is administratively transferred because his/her presence poses a substantiated threat to the health and safety of students, staff or others, the Food Service Manager at the receiving site shall be informed of the reasons for the transfer.
- 6.0 <u>Record Keeping</u> -- Food Service Managers should not be negatively evaluated based solely upon inadequate record-keeping completed by replacement employees while the Food Service Managers were on <u>leave</u>.
- 7.0 Rotation of Staff -- When the District makes determinations regarding the rotation of staff within the Food Services Division, the affected employees' requests shall be considered. When experience and qualifications are equal, seniority, among other factors, will be considered. In the event that an employee's request concerning rotation is denied, upon the employee's request, the District shall state the reasons for the denial in writing.
- 8.0 <u>Off-basis Assignments:</u> Off-basis work assignments (i.e., during summer, winter, and spring breaks) for Food Service Managers shall be made by the Area Food Services Supervisors. To the extent practicable, Food Service Managers shall be considered first for and offered off-basis assignment bargaining unit work, if work is available, based upon the following:

Appendix F - Food Services Addendum

- a. The employee shall have filed with the Division an application for <u>assignments</u> or an availability form by the deadline established by the Division. Late applications or <u>an</u> availability forms may be accepted at the sole discretion of the Division.
- b. An employee who has received a Notice of Unsatisfactory Service or Act shall be precluded from working an off-basis assignment for a period of one year from the date the Notice was issued, unless the employee has consent from his/her Area Food Services Supervisor.
- c. The availability of Food Service Manager assignments during the off-basis period.
- d. If the regularly-assigned Food Service Manager is not available during the assignment in question, the Division will first offer the assignment to an employee based on seniority in the same classification as the regularly assigned Food Service Manager and based on the employee's availability to work as stated on the application or availability form. If no employee in the same classification is available for the assignment, the assignment will be made by the Division at its discretion.
- e. The assignment decision of the Division is final. An employee need not be offered more than one (1) assignment, provided such offer is in one of the area(s) requested on the employee's application or availability form.
- <u>f.</u> Employees who accept an off-basis assignment shall complete that assignment for the entire off-basis period. Failure to begin on the first day or otherwise complete an assignment other than for verifiable illness, bereavement or personal necessity absence may result in separation from the assignment. It may also preclude an employee for consideration for the next off-basis assignment. Exceptions may be made at the sole discretion of the Division. Notification requirements per Article XIII, Section 5.0 of this Agreement will be in effect during the assignment.
- 8.1 Pay Rate during Off-Basis Assignments: Any manager who accepts and works in an off-basis assignment shall be paid their regular rate of pay for the duration of the assignment.
- 8.2 Hours per day: Each off-basis assignment shall be a minimum of three (3) hours per day.
- 9.0 The Director of Food Services shall disseminate information to the appropriate classifications throughout the Division regarding seminars or conferences to which the District intends to send a designated number of Food Services Division supervisors in that classification. Supervisors who have applied for and received pre-

Appendix F - Food Services Addendum

approval for reimbursement for the cost of attending such a seminar or conference shall, in fact, receive that reimbursement.

- 10.0 Food Services Managers assigned to year round schools and who also work a "C" Basis assignment plus a "Z" Basis assignment for the remainder of the school year shall accrue and be entitled to take vacation and illness time pursuant to the terms of Article XVII (Vacation) and Article XIII (Leaves) during either their "C" Basis or "Z" Basis assignment.
- 11.0 Food Services Training Specialists assigned to a "C" Basis assignment and who also work a "Z" Basis assignment for the remainder of the school year shall accrue and be entitled to take vacation and illness time pursuant to terms of Article XVII (Vacation) and Article XIII (Leaves) during their "C" Basis or "Z" Basis assignment.
- 12.0 Food Services Division supervisors who work a "Z" Basis Assignment shall be eligible to take holidays during that time pursuant to Article XVI (Holidays), Section 1.0 (b).

APPENDIX G

CONSTRUCTION INSPECTORS ADDENDUM

1.0 Personnel Commission Rule 589 shall continue to apply to Construction Inspectors in whatever form promulgated by the Personnel Commission at the time of its application.

APPENDIX H

SENIOR GARDENERS ADDENDUM

- 1.0 It is the intent of the District to provide reasonable and necessary training which will assist Senior Gardeners in maintaining the ability to perform their duties. The District shall also make available to Senior Gardeners information regarding staff development classes, including planning classes, which will assist Senior Gardeners in the performance of their work duties and/or promotional opportunities, including Principal Gardener.
- 2.0 The District shall provide quarterly safety meetings for Senior Gardeners.
- 3.0 The Principal Gardener and/or Landscape Supervisor shall be responsible for providing technical supervision for all Senior Gardeners. Not-withstanding the foregoing, for those Senior Gardeners who are assigned to the central crew and who report directly to the Principal Gardener, the Principal Gardener shall provide both technical and general supervision for such Senior Gardeners.
- 4.0 The Principal Gardener and/or Landscape Supervisor shall provide semi-annual inspections of work performed by Senior Gardeners.
- 5.0 The Senior Gardener may request that the site administrator or designee obtain technical input from the Principal Gardener and/or Landscape Supervisor on the evaluations of Senior Gardeners. Notwithstanding the foregoing, for those Senior Gardeners who are assigned to the central crew and who report directly to the Principal Gardener, the Principal Gardeners shall provide both technical and general input into the Senior Gardeners' evaluation.
- 6.0 Senior Gardeners shall make suggestions regarding developing and maintaining work schedules. The Senior Gardener may include in these suggestions information regarding the length of time necessary to complete various tasks as a Senior Gardener. Implementation of these suggestions shall be subject to the approval of the site administrator or designee, who may receive input from the Landscape Supervisor and/or Principal Gardener regarding that schedule.
- 7.0 The District shall continue to provide information to Senior Gardeners regarding both District training and opportunities outside the District for training related to spraying of pesticides and herbicides. The District shall explore training opportunities for Senior Gardeners that are available at no cost through vendors, that are related to fertilization, weed abatement, pruning, turf renovation, minor equipment repair and the spraying of pesticides and/or herbicides. Nothing herein shall be construed as requiring the District to provide such training.
- 8.0 Senior Gardeners shall be entitled to receive overtime compensation in accordance with Article IX (Hours of Work) of this agreement.

Appendix H – Senior Gardeners Addendum

- 9.0 Maintenance and Operations shall make reasonable efforts to assist in the transportation of oversized gardening equipment for repair.
- 10.0 The District and Union recognize that the District does not provide substitutes for Senior Gardeners who are absent for extended periods due to vacation, illness, leaves or jury duty. The District encourages school sites to work internally and with each other in cooperation to help reduce the impact.

APPENDIX I

SUPERVISING SPECIAL EDUCATION ASSISTANTS ADDENDUM

EXTENDED SCHOOL YEAR (ESY) ASSIGNMENTS

- 1.0 Note less than ten (10) working days prior to the deadline date for applications, the District shall establish and distribute a list of schools for each Local District. Supervising Special Education Assistants (SSEAs) must apply on-line for ESY.
 - 1.1 SSEAs who have not received a Notice of Unsatisfactory Service in the last one (1) school year may apply for any school that has been established as hosting ESY.
- 2.0 SSEAs shall be assigned at each school on the basis of priority followed by seniority in the job classification.
 - 2.1 For the purpose of 2.0 above, priority will be defined to mean the hosting school's SSEA will have priority to that school site, regardless of enrollment.
 - 2.1.1 If there is an assignment at a non-school site, the regularly assigned SSEA shall have priority to the assignment.
 - 2.2 Should it become necessary to "share" or otherwise reassign an SSEA, the District will choose from available applicants who have the most seniority in the job class and have indicated the availability to work.
 - 2.3 Section 2.2 shall apply if the current SSEA position is vacant or otherwise unfilled.
- 3.0 SSEAs shall be notified in writing as to assignment to ESY. To the extent possible, notice shall be e-mailed to the applicant's LAUSD e-mail address ten (10) days prior to the end of the current school year.
 - 3.1 SSEAs shall also be notified by e-email if they are not receiving an assignment and informed their name will be added to the ESY substitute pool.

APPENDIX J

UNIT S SALARY RATES 2014-2017 Salary Schedule Effective July 1, 2014

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
2375	Absence Coordinator	SS	\$38.56025	\$40.77449	\$43.10786	\$45.51132	\$47.98483
1161	Accountant	SS	\$21.83422	\$23.06747	\$24.37080	\$25.76522	\$27.23672
1101	Accounting Analyst	SS	\$28.73625	\$30.36890	\$32.02259	\$33.88648	\$35.79242
1127	Accounting System Specialist	SS	\$29.52804	\$31.20976	\$32.95454	\$34.82544	\$36.82948
2170	Administrative Assistant, Building Program	SS	\$34.88851	\$36.83649	\$38.94563	\$41.13185	\$43.44421
2406	Administrative Secretary I	SS	\$23.40382	\$24.69313	\$26.12258	\$27.58707	\$29.17769
2150	Adult Education Administrative Specialist	SS	\$7,354.68	\$7,765.02	\$8,187.69	\$8,635.03	\$9,112.64
1444	Architect	SS	\$37.90157	\$40.00371	\$42.26700	\$44.61440	\$47.04587
1438	Architectural Associate	SS	\$33.01760	\$34.88851	\$36.83649	\$38.94563	\$41.13185
1460	Architectural Project Facilitator	SS	\$32.12770	\$33.94955	\$35.83446	\$37.90157	\$40.00371
5804	Area Bus Supervisor	SS	\$27.29978	\$28.85536	\$30.47402	\$32.19777	\$34.00561
3271	Area Carpentry Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$37.53019
3206	Area Electrical Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$44.22201
4314	Area Food Services Supervisor	SS	\$5,351.21	\$5,653.91	\$5,972.31	\$6,309.78	\$6,665.18
3211	Area Heating and Air Conditioning Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$46.24005
3236	Area Metal Work Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$43.51428
4016	Area Operations Supervisor	SS	\$4,953.20	\$5,224.51	\$5,522.74	\$5,837.78	\$6,166.27
3276	Area Painting Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$35.57521
3216	Area Plumbing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$45.62343
3165	Asbestos Abatement Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$42.04278
5806	Assistant Area Bus Supervisor	SS	\$25.88434	\$27.29978	\$28.85536	\$30.47402	\$32.19777
1541	Assistant Civil Engineer	SS	\$31.25180	\$33.01760	\$34.88851	\$36.83649	\$38.94563
2057	Assistant Contracts Supervisor	SS	\$32.02259	\$33.88648	\$35.79242	\$37.79646	\$39.96867
4329	Assistant Food Production Manager	SS	\$29.07959	\$30.74029	\$32.47806	\$34.30691	\$36.24789
5755	Assistant Garage Supervisor	SS	\$33.45905	\$35.37199	\$37.36903	\$39.48518	\$41.73446
4173	Assistant Gardening Supervisor	SS	\$24.63007	\$26.01748	\$27.51701	\$29.06558	\$30.70524
3258	Assistant Hardware Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$33.88648
5140	Assistant Logistics Supervisor	SS	\$26.08049	\$27.55924	\$29.12174	\$30.75405	\$32.47818
2218	Assistant Mail Supervisor	SS	\$26.01046	\$27.44693	\$28.99550	\$30.62817	\$32.37294
4574	Assistant Outdoor-Education Center Manager	SS	\$23.52994	\$24.88232	\$26.31178	\$27.76225	\$29.33185
4061	Assistant Plant Manager I	SS	\$14.79905	\$15.59087	\$16.50179	\$17.44776	\$18.38671
4048	Assistant Plant Manager II	SS	\$16.50179	\$17.44776	\$18.38671	\$19.45881	\$20.56593
1791	Assistant Projects Manager	SS	\$6,842.33	\$7,230.24	\$7,624.87	\$8,049.79	\$8,489.28
1632	Assistant Supervising Building/Construction Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$51.19549
1531	Associate Civil Engineer	SS	\$34.88851	\$36.83649	\$38.94563	\$41.13185	\$43.44421
1709	Associate Electrical Engineer	SS	\$33.01760	\$34.88851	\$36.83649	\$38.94563	\$41.13185
1763	Associate Mechanical Engineer	SS	\$33.01760	\$34.88851	\$36.83649	\$38.94563	\$41.13185
1471	Building Project Estimator	SS	\$31.25180	\$33.01760	\$34.88851	\$36.83649	\$38.94563
1651	Building/Construction Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$46.47129
1650	Building/Construction Inspector (Class 1)	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$47.73958

APPENDIX J

UNIT S SALARY RATES 2014-2017 Salary Schedule Effective July 1, 2014

5731 Bus Dispatcher SS \$28,85536 \$30,47402 \$32,19777 \$34,00561 \$35,92556 3192 Carpentry Technical Supervisor SS \$0,00000 \$0,00000 \$0,00000 \$37,53019 2620 Cataloging Supervisor SS \$6,842,33 \$7,230,24 \$76,2487 \$8,049,79 \$8,489,28 2620 Cataloging Supervisor SS \$20,26462 \$21,44182 \$22,64704 \$23,89431 \$25,24670 3160 Central Shops Supervisor SS \$7,828,93 \$7,695,51 \$8,000.00 \$0,00000 \$30,0000 \$29,07769 2110 Chief Clerk SS \$3,40382 \$24,69313 \$26,12258 \$27,58707 \$29,17769 5151 Chief Inventory Analyst SS \$30,0890 \$42,37913 \$44,69848 \$47,17901 \$49,7643 5152 Chief Inventory Analyst SS \$30,04277 \$31,91349 \$33,7636 \$35,79242 \$37,79646 5150 Chief Englinedro SS \$30,04277 \$31,91349 \$33,74303 <	Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
3192 Carpentry Supervisor (Building Moving) SS \$0,00000 \$0,00000 \$0,00000 \$0,00000 \$37,580119 620 Cataloging Supervisor SS \$20,26462 \$21,44182 \$22,6470 \$23,89431 \$25,24670 3160 Central Shops Supervisor SS \$0,00000 \$0,00000 \$0,00000 \$30,0000 \$20,0000 \$		Bus Dispatcher	SS	\$28.85536	\$30.47402	\$32.19777	\$34.00561	\$35.92556
3148 Carpentry Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 2620 Cataloging Supervisor SS \$20,26460 \$21,44182 \$22,64704 \$23,89431 \$25,24670 3100 Central Shops Supervisor SS \$0,00000 \$0,00000 \$30,0000 \$0,00000 \$38,88125 2511 Chief Ernel Gliment Analysis Coordinator SS \$7,892.93 \$7,695.51 \$8,108.09 \$8,511.03 \$9,028.55 5192 Chief Inventory Analyst SS \$6,051.92 \$6,388.26 \$6,753.75 \$71,371.91 \$7,536.31 2045 Claims Processing Supervisor SS \$30,36890 \$32,0259 \$33,86848 \$35,79242 \$37,70946 2056 Communications Production Facilitator SS \$30,26250 \$33,8648 \$35,79242 \$37,70946 2056 Communications Production Facilitator SS \$32,62520 \$33,8648 \$37,7933 \$37,67033 2072 Construction Safety Specialist SS \$32,62520 \$34,46107		·						
2620 Calaloging Supervisor SS \$20.26462 \$21.44182 \$22.6470 \$23.89431 \$25.24670 3160 Central Shops Supervisor SS \$3.0000 \$0.0000 \$0.0000 \$0.0000 \$38.58125 2110 Chief Broadcasting Engineer SS \$7.282.93 \$7.695.51 \$8.1080.9 \$8.561.03 \$9.028.55 2511 Chief Ehrollment Analysis Coordinator SS \$40.09480 \$42.37913 \$44.69848 \$47.77901 \$49.76463 5192 Chief Inventory Analyst SS \$6.051.22 \$6.882.26 \$6.75.35 \$7.137.19 \$7.536.31 1528 Civil Engineer SS \$38.65834 \$40.80253 \$43.11487 \$45.50431 \$47.97081 2045 Claims Processing Supervisor SS \$32.62250 \$33.86848 \$35.79424 \$37.79646 2056 Computer Applications Support Supervisor SS \$32.62520 \$34.46107 \$36.4160 \$38.48316 \$40.6032 1792 Construction Salety Specialist SS \$32.62520 \$34.46107		1 3 1 , 0 6,			•	•		
3160 Central Snops Supervisor SS \$0,00000 \$0,00000 \$0,00000 \$38,58125 2110 Chief Broadcasting Engineer SS \$7,282.93 \$7,695.51 \$8,108.09 \$8,561.03 \$9,028.55 511 Chief Clerk SS \$32,40838 \$24,69313 \$26,12258 \$27,58707 \$29,1776 1501 Chief Enrollment Analysis Coordinator SS \$40,09480 \$42,37913 \$44,69848 \$47,17901 \$49,76463 5192 Chief Inventory Analyst SS \$6,051.92 \$6,388.26 \$6,753.75 \$71,371.91 \$7,536.31 2015 Claims Processing Supervisor SS \$30,68680 \$32,02259 \$33,88648 \$35,79242 \$37,7033 4817 Communications Production Facilitator SS \$32,62520 \$34,46107 \$36,4160 \$38,48316 \$40,60632 1792 Constructions Safety Specialist SS \$32,213160 \$22,11582 \$23,70513 \$25,06451 \$24,2389 1125 Contract Assistant SS \$21,2160 \$22,11582		·						
2110 Chief Broadcasting Engineer SS \$7,282.93 \$7,695.15 \$8,108.09 \$8,561.03 \$9,028.55 2511 Chief Clerk SS \$23,40382 \$24,69313 \$26,12258 \$27,58707 \$29,17769 510 Chief Inventory Analyst SS \$6,051.92 \$6,388.26 \$6,753.75 \$7,137.19 \$7,536.31 1528 Civil Engineer SS \$30,36890 \$32,02259 \$33,8848 \$35,79242 \$37,79646 2056 Communications Production Facilitator SS \$30,36890 \$32,02259 \$33,8848 \$35,79324 \$37,79646 2056 Communications Support Supervisor SS \$32,62520 \$34,46107 \$36,41606 \$38,48316 \$40,60632 4817 Contract Assistant SS \$21,23160 \$22,41582 \$23,70513 \$25,06451 \$26,64538 1725 Contract Payments Supervisor SS \$32,76513 \$25,06451 \$26,42389 \$27,9444 \$29,52804 2136 Coordinator of Telecommunication Services SS \$32,70513								
2511 Chief Clerk SS \$23.40828 \$24.69313 \$26.12258 \$27.58707 \$29.17769 1501 Chief Inventory Analyst SS \$40.09480 \$42.37913 \$44.69848 \$47.17901 \$49.70661 1502 Chief Inventory Analyst SS \$6.051.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.051.92 \$6.051.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92 \$6.060.92				\$7,282.93	\$7,695.51	\$8,108.09	\$8,561.03	
5128 Chief Inventory Analyst SS \$6,051.92 \$6,388.26 \$6,753.75 \$7,137.19 \$7,536.31 1528 Civil Engineer SS \$38,65834 \$40,80253 \$43,11487 \$45,50431 \$47,97081 2045 Claims Processing Supervisor SS \$30,36890 \$33,76036 \$35,67330 \$37,79046 2056 Communications Production Facilitator SS \$30,24277 \$31,93149 \$33,76036 \$35,67330 \$37,67033 4817 Computer Applications Support Supervisor SS \$32,26520 \$34,46107 \$36,41606 \$38,48316 \$40,60632 2531 Contract Payments Supervisor SS \$21,23160 \$22,41582 \$37,9137.9 \$30,38999 2534 Coordinating Contract Assistant SS \$23,05131 \$25,06451 \$26,42389 \$27,94444 \$29,52804 1125 Contract Payments Supervisor SS \$30,12365 \$31,78441 \$33,62020 \$35,51213 \$37,50216 2236 Coordinator of Telecommunication Services SS \$30,12365 \$31,784	2511	5 5		\$23.40382	\$24.69313	\$26.12258	\$27.58707	\$29.17769
5192 Chief Inventory Analyst SS \$6,051.92 \$6,388.26 \$6,753.75 \$7,137.19 \$7,536.31 1528 Civil Engineer SS \$30,36890 \$32.02259 \$43.11487 \$45.0431 \$47,79081 2045 Claims Processing Supervisor SS \$30,36890 \$32.02259 \$33.8648 \$35,79242 \$37.79464 2056 Commulcat Applications Support Supervisor SS \$32,62520 \$34.4107 \$36.41606 \$38.48316 \$40.60632 27792 Contract Applications Support Supervisor SS \$32.21301 \$22.41582 \$37.19348 \$39.31701 \$41.51725 2531 Contract Assistant SS \$27.23160 \$22.41582 \$23.70513 \$25.06451 \$26.42389 1125 Contract Payments Supervisor SS \$23.70513 \$25.06451 \$26.42389 \$27.94444 \$29.52804 1125 Contract Payments Supervisor SS \$23.05255 \$34.46107 \$36.41606 \$38.4310 \$20.5280 125 Contract Payments Supervisor SS \$23.0123	1501	Chief Enrollment Analysis Coordinator	SS	\$40.09480	\$42.37913	\$44.69848	\$47.17901	\$49.76463
2045 Claims Processing Supervisor SS \$30.36890 \$32.02259 \$33.8648 \$35.79242 \$37.79646 2056 Communications Production Facilitator SS \$30.24277 \$31.93149 \$33.76036 \$35.67330 \$37.707033 4817 Computer Applications Support Supervisor SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1792 Construction Safety Specialist SS \$33.31891 \$35.22485 \$37.19384 \$39.31701 \$41.51725 2531 Contract Payments Supervisor SS \$21.23160 \$22.41582 \$23.70513 \$25.06451 \$26.42389 2534 Coordinating Contract Assistant SS \$32.6520 \$34.41607 \$36.41606 \$38.48316 \$40.60632 1790 Cost Recovery Claims Processing Supervisor SS \$32.62520 \$34.41607 \$36.41606 \$38.48316 \$40.60632 1790 Cost Recovery Claims Processing Supervisor SS \$32.389431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 1301 Customer Se	5192		SS	\$6,051.92	\$6,388.26	\$6,753.75	\$7,137.19	\$7,536.31
2056 Communications Production Facilitator SS \$30.24277 \$31.93149 \$33.76036 \$35.67330 \$37.67033 4817 Computer Applications Support Supervisor SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1792 Construction Safety Specialist SS \$33.31891 \$35.22485 \$37.19384 \$39.31701 \$41.51725 2531 Contract Payments Supervisor SS \$21.23160 \$22.41852 \$23.70513 \$25.06451 \$26.42389 125 Coordinating Contract Assistant SS \$32.70513 \$25.06451 \$26.42389 \$27.94444 \$29.52804 2138 Coordinating Contract Assistant SS \$32.62520 \$34.46107 \$36.41606 \$38.848316 \$40.60632 2138 Coordinator of Telecommunication Services SS \$32.05250 \$34.46107 \$36.41606 \$38.48316 \$40.60632 21990 Cost Recovery Claims Processing Supervisor SS \$32.389431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 2301 Databil	1528	Civil Engineer	SS	\$38.65834	\$40.80253	\$43.11487	\$45.50431	\$47.97081
4817 Computer Applications Support Supervisor SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1792 Construction Safety Specialist SS \$33.31891 \$35.22485 \$37.19384 \$39.31701 \$41.51725 2531 Contract Assistant SS \$21.23160 \$22.41582 \$32.70513 \$25.06451 \$26.42389 2534 Coordinating Contract Assistant SS \$23.70513 \$25.06451 \$26.42389 \$37.9444 \$29.52804 2138 Coordinator of Telecommunication Services SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1990 Cost Recovery Claims Processing Supervisor SS \$30.12365 \$31.78434 \$33.60200 \$35.51213 \$37.50216 2236 Credentials and Contract Supervisor SS \$18.78991 \$19.85277 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$0.0000 \$7,896.19 \$8,330.08 \$87,91.98 4881 Design Network Engineer	2045	Claims Processing Supervisor	SS	\$30.36890	\$32.02259	\$33.88648	\$35.79242	\$37.79646
1792 Construction Safety Specialist SS \$33.31891 \$35.22485 \$37.19384 \$39.31701 \$41.51725 2531 Contract Assistant SS \$21.23160 \$22.41582 \$23.70513 \$25.06451 \$26.42389 2534 Coordinating Contract Assistant SS \$27.06131 \$25.06451 \$26.42389 \$27.94444 \$29.52804 2138 Coordinator of Telecommunication Services SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1990 Cost Recovery Claims Processing Supervisor SS \$30.12365 \$31.78434 \$33.62020 \$35.51213 \$37.50216 2236 Credentials and Contract Supervisor SS \$23.89431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 2301 Customer Services Center Supervisor SS \$18.78901 \$19.86277 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$0.0000 \$0.0000 \$7.906-19 \$8,330.8 \$8.791.98 4624 Displitity Suppor	2056	Communications Production Facilitator	SS	\$30.24277	\$31.93149	\$33.76036	\$35.67330	\$37.67033
2531 Contract Assistant SS \$21.23160 \$22.41582 \$23.70513 \$25.06451 \$26.42389 1125 Contract Payments Supervisor SS \$27.64314 \$29.17769 \$30.81735 \$32.51308 \$34.39099 2534 Coordinating Contract Assistant SS \$23.70513 \$25.06451 \$26.42389 \$27.94444 \$29.52804 2138 Coordinator of Telecommunication Services SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1990 Cost Recovery Claims Processing Supervisor SS \$30.2035 \$31.78434 \$33.6020 \$35.51213 \$37.50216 2236 Credentials and Contract Supervisor SS \$23.89431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 1301 Customer Services Center Supervisor SS \$18.78901 \$19.85277 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$0.00000 \$7.896.19 \$8,330.88 \$87919 4881 Design Network Engineer SS <td>4817</td> <td>Computer Applications Support Supervisor</td> <td>SS</td> <td>\$32.62520</td> <td>\$34.46107</td> <td>\$36.41606</td> <td>\$38.48316</td> <td>\$40.60632</td>	4817	Computer Applications Support Supervisor	SS	\$32.62520	\$34.46107	\$36.41606	\$38.48316	\$40.60632
1125 Contract Payments Supervisor SS \$27.64314 \$29.17769 \$30.81735 \$32.51308 \$34.39099 2534 Coordinating Contract Assistant SS \$23.70513 \$25.06451 \$26.42389 \$27.94444 \$29.52804 2138 Coordinator of Telecommunication Services SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1990 Cost Recovery Claims Processing Supervisor SS \$31.2365 \$31.78434 \$33.62020 \$35.51213 \$37.50216 2236 Credentials and Contract Supervisor SS \$23.89431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 1301 Customer Services Center Supervisor SS \$18.78901 \$19.85277 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$7.896.19 \$8.330.08 \$8.791.98 4881 Design Network Engineer SS \$41.52426 \$43.87165 \$46.26808 \$48.81167 \$51.52343 4600 Disability Support Services Program Assistant S	1792	Construction Safety Specialist	SS	\$33.31891	\$35.22485	\$37.19384	\$39.31701	\$41.51725
2534 Coordinating Contract Assistant SS \$23.70513 \$25.06451 \$26.42389 \$27.94444 \$29.52804 2138 Coordinator of Telecommunication Services SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1990 Cost Recovery Claims Processing Supervisor SS \$30.12365 \$31.78434 \$33.62020 \$35.51213 \$37.50216 2236 Credentials and Contract Supervisor SS \$23.89431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 1301 Customer Services Center Supervisor SS \$18.78901 \$19.85277 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$7.896.19 \$8,330.08 \$8,791.98 4881 Design Network Engineer SS \$41.52426 \$43.87165 \$46.26808 \$48.81167 \$51.52343 4600 Display and Graphic Designer SS \$27.94444 \$29.52804 \$31.20976 \$32.95545 \$34.82544 5739 Driver Trainer (Truck) SS \$28	2531	Contract Assistant	SS	\$21.23160	\$22.41582	\$23.70513	\$25.06451	\$26.42389
2138 Coordinator of Telecommunication Services SS \$32.62520 \$34.46107 \$36.41606 \$38.48316 \$40.60632 1990 Cost Recovery Claims Processing Supervisor SS \$30.12365 \$31.78434 \$33.62020 \$35.51213 \$37.50216 2236 Credentials and Contract Supervisor SS \$23.89431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 1301 Customer Services Center Supervisor SS \$18.78901 \$19.82577 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$7.896.19 \$8,330.08 \$8,791.98 4881 Design Network Engineer SS \$41.52426 \$43.87165 \$44.26808 \$48.81167 \$51.52343 4600 Disablity Support Services Program Assistant SS \$22.51688 \$21.67306 \$22.91331 \$24.19562 \$25.52698 4624 Display and Graphic Designer SS \$27.94444 \$29.52804 \$31.20976 \$32.95454 \$34.82544 5739 Driver Trainer (Bus-Truck) SS <td>1125</td> <td>Contract Payments Supervisor</td> <td></td> <td></td> <td>\$29.17769</td> <td>\$30.81735</td> <td>\$32.51308</td> <td>\$34.39099</td>	1125	Contract Payments Supervisor			\$29.17769	\$30.81735	\$32.51308	\$34.39099
1990 Cost Recovery Claims Processing Supervisor SS \$30.12365 \$31.78434 \$33.62020 \$35.51213 \$37.50216 2236 Credentials and Contract Supervisor SS \$23.89431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 1301 Customer Services Center Supervisor SS \$18.78901 \$19.85277 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$7.896.19 \$8,830.08 \$8,791.98 4881 Design Network Engineer SS \$41.52426 \$43.87165 \$46.26808 \$48.81167 \$51.52343 4600 Disability Support Services Program Assistant SS \$20.51688 \$21.67306 \$22.91331 \$24.19562 \$25.52698 4624 Display and Graphic Designer SS \$27.94444 \$29.52804 \$31.20976 \$32.95454 \$34.82544 5739 Driver Trainer (Fuck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 5734 Driver-Trainer (Bus-Truck) SS \$6.84		· · · · · · · · · · · · · · · · · · ·			\$25.06451	\$26.42389		\$29.52804
2236 Credentials and Contract Supervisor SS \$23.89431 \$25.24670 \$26.70417 \$28.21071 \$29.79432 1301 Customer Services Center Supervisor SS \$18.78901 \$19.85277 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$0.00000 \$7,896.19 \$8,330.08 \$8,791.98 4881 Design Network Engineer SS \$41.52426 \$43.87165 \$46.26808 \$48.81167 \$51.52343 4600 Disability Support Services Program Assistant SS \$20.51688 \$21.67306 \$22.91331 \$24.19562 \$25.52698 4624 Display and Graphic Designer SS \$27.94444 \$29.52804 \$31.20976 \$32.95454 \$34.82544 5739 Driver Trainer (Bus-Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 3126 Electrical Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 5034 Employment Services Manager SS </td <td></td> <td></td> <td></td> <td></td> <td>\$34.46107</td> <td></td> <td></td> <td>\$40.60632</td>					\$34.46107			\$40.60632
1301 Customer Services Center Supervisor SS \$18.78901 \$19.85277 \$20.96516 \$22.18088 \$23.43308 1073 Database Specialist SS \$0.00000 \$0.00000 \$7,896.19 \$8,330.08 \$8,791.98 4881 Design Network Engineer SS \$41.52426 \$43.87165 \$46.26808 \$48.81167 \$51.52343 4600 Disability Support Services Program Assistant SS \$20.51688 \$21.67306 \$22.91331 \$24.19562 \$25.52698 4624 Display and Graphic Designer SS \$27.94444 \$29.52804 \$31.20976 \$32.95454 \$34.82544 5739 Driver Trainer (Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 5734 Driver-Trainer (Bus-Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 5734 Electrical Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 3522 Electrical Technical Supervisor SS								
1073 Database Specialist SS \$0.00000 \$0.00000 \$7,896.19 \$8,330.08 \$8,791.98 4881 Design Network Engineer SS \$41.52426 \$43.87165 \$46.26808 \$48.81167 \$51.52343 4600 Disability Support Services Program Assistant SS \$20.51688 \$21.67306 \$22.91331 \$24.19562 \$25.52698 4624 Display and Graphic Designer SS \$27.94444 \$29.52804 \$31.20976 \$32.95454 \$34.82544 5739 Driver Trainer (Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 3126 Electrical Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 3522 Electronics Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 5034 Employment Services Manager SS \$33.76036 \$35.55299 \$37.0538 \$39.84956 \$42.03578 1778 Environmental Assessment Coordinator SS		·						
4881 Design Network Engineer SS \$41.52426 \$43.87165 \$46.26808 \$48.81167 \$51.52343 4600 Disability Support Services Program Assistant SS \$20.51688 \$21.67306 \$22.91331 \$24.19562 \$25.52698 4624 Display and Graphic Designer SS \$27.94444 \$29.52804 \$31.20976 \$32.95454 \$34.82544 5739 Driver Trainer (Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 5734 Driver-Trainer (Bus-Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 3126 Electrical Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 3522 Electronics Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 5034 Employment Services Manager SS \$33.76036 \$35.565929 \$37.70538 \$39.84956 \$42.03578 1778 Environmental Assessment Coordinator		•						
4600Disability Support Services Program AssistantSS\$20.51688\$21.67306\$22.91331\$24.19562\$25.526984624Display and Graphic DesignerSS\$27.94444\$29.52804\$31.20976\$32.95454\$34.825445739Driver Trainer (Truck)SS\$28.85536\$30.47402\$32.19777\$34.00561\$35.925565734Driver-Trainer (Bus-Truck)SS\$28.85536\$30.47402\$32.19777\$34.00561\$35.925563126Electrical Technical SupervisorSS\$6,842.33\$7,230.24\$7,624.87\$8,049.79\$8,489.283522Electronics Technical SupervisorSS\$6,842.33\$7,230.24\$7,624.87\$8,049.79\$8,489.285034Employment Services ManagerSS\$33.76036\$35.65929\$37.70538\$39.84956\$42.035781778Environmental Assessment CoordinatorSS\$33.31891\$35.22485\$37.19384\$39.31701\$41.517253015Environmental Health SupervisorSS\$37.21332\$39.33476\$41.53521\$43.88764\$46.361642143Escrow and Title SupervisorSS\$32.75834\$34.62223\$36.54919\$38.63031\$40.809531963Facilities Cost AnalystSS\$39.30299\$41.53127\$43.80158\$46.24005\$48.769621813Facilities Services CoordinatorSS\$31.48304\$33.27687\$35.14076\$37.13079\$39.225915748Fleet Technical Support SpecialistSS\$35.37199\$3		•						
4624Display and Graphic DesignerSS\$27.94444\$29.52804\$31.20976\$32.95454\$34.825445739Driver Trainer (Truck)SS\$28.85536\$30.47402\$32.19777\$34.00561\$35.925565734Driver-Trainer (Bus-Truck)SS\$28.85536\$30.47402\$32.19777\$34.00561\$35.925563126Electrical Technical SupervisorSS\$6,842.33\$7,230.24\$7,624.87\$8,049.79\$8,489.283522Electronics Technical SupervisorSS\$6,842.33\$7,230.24\$7,624.87\$8,049.79\$8,489.285034Employment Services ManagerSS\$33.76036\$35.65929\$37.70538\$39.84956\$42.035781778Environmental Assessment CoordinatorSS\$33.31891\$35.22485\$37.19384\$39.31701\$41.517253015Environmental Compliance CoordinatorSS\$28.24574\$29.85739\$31.53208\$33.31891\$35.224851801Environmental Health SupervisorSS\$37.21332\$39.33476\$41.53521\$43.88764\$46.361642143Escrow and Title SupervisorSS\$41.06880\$43.37414\$45.79860\$48.26511\$50.941831998Ethics AdviserSS\$32.02259\$33.88648\$35.79242\$37.79646\$39.968671435Facilities Partnership and Development CoordinatorSS\$39.30299\$41.53127\$43.80158\$46.24005\$48.769621813Facilities Services CoordinatorSS\$31.48304\$33.		0						
5739 Driver Trainer (Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 5734 Driver-Trainer (Bus-Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 3126 Electrical Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 3522 Electronics Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 5034 Employment Services Manager SS \$33.76036 \$35.65929 \$37.70538 \$39.84956 \$42.03578 1778 Environmental Assessment Coordinator SS \$33.31891 \$35.22485 \$37.19384 \$39.31701 \$41.51725 3015 Environmental Health Supervisor SS \$37.21332 \$39.33476 \$41.53521 \$43.88764 \$46.36164 2143 Escrow and Title Supervisor SS \$32.75834 \$34.62223 \$36.54919 \$38.63031 \$40.80953 1963 Facilities Cost Analyst SS		3 11						
5734 Driver-Trainer (Bus-Truck) SS \$28.85536 \$30.47402 \$32.19777 \$34.00561 \$35.92556 3126 Electrical Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 3522 Electronics Technical Supervisor SS \$6,842.33 \$7,230.24 \$7,624.87 \$8,049.79 \$8,489.28 5034 Employment Services Manager SS \$33.76036 \$35.65929 \$37.70538 \$39.84956 \$42.03578 1778 Environmental Assessment Coordinator SS \$33.31891 \$35.22485 \$37.19384 \$39.31701 \$41.51725 3015 Environmental Compliance Coordinator SS \$28.24574 \$29.85739 \$31.53208 \$33.31891 \$35.22485 1801 Environmental Health Supervisor SS \$37.21332 \$39.33476 \$41.53521 \$43.88764 \$46.36164 2143 Escrow and Title Supervisor SS \$32.75834 \$34.62223 \$36.54919 \$38.63031 \$40.80953 1963 Facilities Cost Analyst SS								
3126Electrical Technical SupervisorSS\$6,842.33\$7,230.24\$7,624.87\$8,049.79\$8,489.283522Electronics Technical SupervisorSS\$6,842.33\$7,230.24\$7,624.87\$8,049.79\$8,489.285034Employment Services ManagerSS\$33.76036\$35.65929\$37.70538\$39.84956\$42.035781778Environmental Assessment CoordinatorSS\$33.31891\$35.22485\$37.19384\$39.31701\$41.517253015Environmental Compliance CoordinatorSS\$28.24574\$29.85739\$31.53208\$33.31891\$35.224851801Environmental Health SupervisorSS\$37.21332\$39.33476\$41.53521\$43.88764\$46.361642143Escrow and Title SupervisorSS\$41.06880\$43.37414\$45.79860\$48.26511\$50.941831998Ethics AdviserSS\$32.75834\$34.62223\$36.54919\$38.63031\$40.809531963Facilities Cost AnalystSS\$32.02259\$33.88648\$35.79242\$37.79646\$39.968671435Facilities Partnership and Development CoordinatorSS\$39.30299\$41.53127\$43.80158\$46.24005\$48.769621813Facilities Services CoordinatorSS\$31.48304\$33.27687\$35.14076\$37.13079\$39.225915748Fleet Technical Support SpecialistSS\$35.37199\$37.36903\$39.48518\$41.73446\$43.99777		` ,						
3522Electronics Technical SupervisorSS\$6,842.33\$7,230.24\$7,624.87\$8,049.79\$8,489.285034Employment Services ManagerSS\$33.76036\$35.65929\$37.70538\$39.84956\$42.035781778Environmental Assessment CoordinatorSS\$33.31891\$35.22485\$37.19384\$39.31701\$41.517253015Environmental Compliance CoordinatorSS\$28.24574\$29.85739\$31.53208\$33.31891\$35.224851801Environmental Health SupervisorSS\$37.21332\$39.33476\$41.53521\$43.88764\$46.361642143Escrow and Title SupervisorSS\$41.06880\$43.37414\$45.79860\$48.26511\$50.941831998Ethics AdviserSS\$32.75834\$34.62223\$36.54919\$38.63031\$40.809531963Facilities Cost AnalystSS\$32.02259\$33.88648\$35.79242\$37.79646\$39.968671435Facilities Partnership and Development CoordinatorSS\$39.30299\$41.53127\$43.80158\$46.24005\$48.769621813Facilities Services CoordinatorSS\$31.48304\$33.27687\$35.14076\$37.13079\$39.225915748Fleet Technical Support SpecialistSS\$35.37199\$37.36903\$39.48518\$41.73446\$43.99777		,						
5034Employment Services ManagerSS\$33.76036\$35.65929\$37.70538\$39.84956\$42.035781778Environmental Assessment CoordinatorSS\$33.31891\$35.22485\$37.19384\$39.31701\$41.517253015Environmental Compliance CoordinatorSS\$28.24574\$29.85739\$31.53208\$33.31891\$35.224851801Environmental Health SupervisorSS\$37.21332\$39.33476\$41.53521\$43.88764\$46.361642143Escrow and Title SupervisorSS\$41.06880\$43.37414\$45.79860\$48.26511\$50.941831998Ethics AdviserSS\$32.75834\$34.62223\$36.54919\$38.63031\$40.809531963Facilities Cost AnalystSS\$32.02259\$33.88648\$35.79242\$37.79646\$39.968671435Facilities Partnership and Development CoordinatorSS\$39.30299\$41.53127\$43.80158\$46.24005\$48.769621813Facilities Services CoordinatorSS\$31.48304\$33.27687\$35.14076\$37.13079\$39.225915748Fleet Technical Support SpecialistSS\$35.37199\$37.36903\$39.48518\$41.73446\$43.99777		·						
1778 Environmental Assessment Coordinator SS \$33.31891 \$35.22485 \$37.19384 \$39.31701 \$41.51725 3015 Environmental Compliance Coordinator SS \$28.24574 \$29.85739 \$31.53208 \$33.31891 \$35.22485 1801 Environmental Health Supervisor SS \$37.21332 \$39.33476 \$41.53521 \$43.88764 \$46.36164 2143 Escrow and Title Supervisor SS \$41.06880 \$43.37414 \$45.79860 \$48.26511 \$50.94183 1998 Ethics Adviser SS \$32.75834 \$34.62223 \$36.54919 \$38.63031 \$40.80953 1963 Facilities Cost Analyst SS \$32.02259 \$33.88648 \$35.79242 \$37.79646 \$39.96867 1435 Facilities Partnership and Development Coordinator SS \$39.30299 \$41.53127 \$43.80158 \$46.24005 \$48.76962 1813 Facilities Services Coordinator SS \$31.48304 \$33.27687 \$35.14076 \$37.13079 \$39.22591 5748 Fleet Technical Support Specialist		•						
3015Environmental Compliance CoordinatorSS\$28.24574\$29.85739\$31.53208\$33.31891\$35.224851801Environmental Health SupervisorSS\$37.21332\$39.33476\$41.53521\$43.88764\$46.361642143Escrow and Title SupervisorSS\$41.06880\$43.37414\$45.79860\$48.26511\$50.941831998Ethics AdviserSS\$32.75834\$34.62223\$36.54919\$38.63031\$40.809531963Facilities Cost AnalystSS\$32.02259\$33.88648\$35.79242\$37.79646\$39.968671435Facilities Partnership and Development CoordinatorSS\$39.30299\$41.53127\$43.80158\$46.24005\$48.769621813Facilities Services CoordinatorSS\$31.48304\$33.27687\$35.14076\$37.13079\$39.225915748Fleet Technical Support SpecialistSS\$35.37199\$37.36903\$39.48518\$41.73446\$43.99777								
1801 Environmental Health Supervisor SS \$37.21332 \$39.33476 \$41.53521 \$43.88764 \$46.36164 2143 Escrow and Title Supervisor SS \$41.06880 \$43.37414 \$45.79860 \$48.26511 \$50.94183 1998 Ethics Adviser SS \$32.75834 \$34.62223 \$36.54919 \$38.63031 \$40.80953 1963 Facilities Cost Analyst SS \$32.02259 \$33.88648 \$35.79242 \$37.79646 \$39.96867 1435 Facilities Partnership and Development Coordinator SS \$39.30299 \$41.53127 \$43.80158 \$46.24005 \$48.76962 1813 Facilities Services Coordinator SS \$31.48304 \$33.27687 \$35.14076 \$37.13079 \$39.22591 5748 Fleet Technical Support Specialist SS \$35.37199 \$37.36903 \$39.48518 \$41.73446 \$43.99777								
2143 Escrow and Title Supervisor SS \$41.06880 \$43.37414 \$45.79860 \$48.26511 \$50.94183 1998 Ethics Adviser SS \$32.75834 \$34.62223 \$36.54919 \$38.63031 \$40.80953 1963 Facilities Cost Analyst SS \$32.02259 \$33.88648 \$35.79242 \$37.79646 \$39.96867 1435 Facilities Partnership and Development Coordinator SS \$39.30299 \$41.53127 \$43.80158 \$46.24005 \$48.76962 1813 Facilities Services Coordinator SS \$31.48304 \$33.27687 \$35.14076 \$37.13079 \$39.22591 5748 Fleet Technical Support Specialist SS \$35.37199 \$37.36903 \$39.48518 \$41.73446 \$43.99777		•						
1998 Ethics Adviser SS \$32.75834 \$34.6223 \$36.54919 \$38.63031 \$40.80953 1963 Facilities Cost Analyst SS \$32.02259 \$33.88648 \$35.79242 \$37.79646 \$39.96867 1435 Facilities Partnership and Development Coordinator SS \$39.30299 \$41.53127 \$43.80158 \$46.24005 \$48.76962 1813 Facilities Services Coordinator SS \$31.48304 \$33.27687 \$35.14076 \$37.13079 \$39.22591 5748 Fleet Technical Support Specialist SS \$35.37199 \$37.36903 \$39.48518 \$41.73446 \$43.99777		•						
1963 Facilities Cost Analyst SS \$32.02259 \$33.88648 \$35.79242 \$37.79646 \$39.96867 1435 Facilities Partnership and Development Coordinator SS \$39.30299 \$41.53127 \$43.80158 \$46.24005 \$48.76962 1813 Facilities Services Coordinator SS \$31.48304 \$33.27687 \$35.14076 \$37.13079 \$39.22591 5748 Fleet Technical Support Specialist SS \$35.37199 \$37.36903 \$39.48518 \$41.73446 \$43.99777		•						
1435 Facilities Partnership and Development Coordinator SS \$39.30299 \$41.53127 \$43.80158 \$46.24005 \$48.76962 1813 Facilities Services Coordinator SS \$31.48304 \$33.27687 \$35.14076 \$37.13079 \$39.22591 5748 Fleet Technical Support Specialist SS \$35.37199 \$37.36903 \$39.48518 \$41.73446 \$43.99777								
Coordinator SS \$39.30299 \$41.53127 \$43.80158 \$40.24005 \$48.70902 \$1813 Facilities Services Coordinator SS \$31.48304 \$33.27687 \$35.14076 \$37.13079 \$39.22591 \$5748 Fleet Technical Support Specialist SS \$35.37199 \$37.36903 \$39.48518 \$41.73446 \$43.99777	1963	<u> </u>	55	\$32.02259	\$33.88648	\$35.79242	\$37.79646	\$39.96867
5748 Fleet Technical Support Specialist SS \$35.37199 \$37.36903 \$39.48518 \$41.73446 \$43.99777	1435	•	SS	\$39.30299	\$41.53127	\$43.80158	\$46.24005	\$48.76962
	1813	Facilities Services Coordinator	SS	\$31.48304	\$33.27687	\$35.14076	\$37.13079	\$39.22591
3251 Floor Covering Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$37.22187	5748	Fleet Technical Support Specialist	SS	\$35.37199	\$37.36903	\$39.48518	\$41.73446	\$43.99777
	3251	Floor Covering Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$37.22187

APPENDIX J

UNIT S SALARY RATES 2014-2017 Salary Schedule Effective July 1, 2014

Class							
Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
3128	Floor Covering Technical Supervisor	SS	\$39.32260	\$41.55346	\$43.82077	\$46.26438	\$48.78701
4330	Food Production Supervisor	SS	\$18.64598	\$19.71106	\$20.86023	\$22.01640	\$23.23564
3121	Food Service Equipment Specialist	SS	\$37.79646	\$39.96867	\$42.24600	\$44.60739	\$47.03186
4291	Food Service Manager I	SS	\$16.71201	\$17.65096	\$18.64598	\$19.71106	\$20.86023
4292	Food Service Manager II	SS	\$17.65096	\$18.64598	\$19.71106	\$20.86023	\$22.01640
4293	Food Service Manager III	SS	\$18.64598	\$19.71106	\$20.86023	\$22.01640	\$23.23564
4296	Food Service Manager IV	SS	\$19.71106	\$20.86023	\$22.01640	\$23.23564	\$24.54599
4297	Food Service Manager V	SS	\$20.86023	\$22.01640	\$23.23564	\$24.54599	\$25.93340
4294	Food Service Manager VI	SS	\$22.01640	\$23.23564	\$24.54599	\$25.93340	\$27.40489
4295	Food Service Manager VII	SS	\$23.23564	\$24.54599	\$25.93340	\$27.40489	\$28.93945
4312	Food Service Training Specialist	SS	\$23.64907	\$24.98043	\$26.37484	\$27.87436	\$29.42294
1071	Food Services Cost Analyst	SS	\$32.02259	\$33.88648	\$35.79242	\$37.79646	\$39.96867
5756	Garage Supervisor	SS	\$35.37199	\$37.36903	\$39.48518	\$41.73446	\$43.99777
3266	Glazing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$39.33102
3051	Graphic Arts Machinist Supervisor	SS	\$34.40501	\$36.36000	\$38.40609	\$40.58529	\$42.86962
3256	Hardware Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$37.83150
5201	Head Stock Clerk	SS	\$21.32270	\$22.52793	\$23.80323	\$25.14158	\$26.54300
3137	Heating and Air Conditioning Technical	SS	\$6,842.33	\$7,230.24	\$7,624.87	\$8,049.79	\$8,489.28
3137	Supervisor	33	Φ 0,042.33	\$1,230.24	\$7,024.07	\$0,047.77	Φ0,407.20
3132	HVAC Technical Supervisor (High-Rise)	SS	\$39.30299	\$41.53127	\$43.80158	\$46.24005	\$48.76962
3134	HVAC Test and Inspection Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$46.24005
1782	Industrial Hygienist	SS	\$33.31891	\$35.22485	\$37.19363	\$39.31701	\$41.51725
4620	Instructional Media Technician-Adviser	SS	\$30.54408	\$32.29587	\$34.11071	\$36.03767	\$38.09777
4156	Integrated Pest Management Program	SS	\$31.48304	\$33.27687	\$35.14076	\$37.13079	\$39.22591
	Coordinator						
1151	Intermediate Accountant	SS	\$24.37080	\$25.76522	\$27.23672	\$28.73625	\$30.36890
3031	IT Electronics Communications Supervisor	SS	\$33.45905	\$35.35097	\$37.37603	\$39.46415	\$41.68543
3543	IT Field Services Supervisor	SS	\$29.85739	\$31.54610	\$33.35394	\$35.22485	\$37.23589
4880	IT Infrastructure Project Network Engineer	SS	\$33.63422	\$35.53316	\$37.53720	\$39.66036	\$41.90264
1193	IT Project Manager	SS	\$6,712.27	\$7,097.94	\$7,503.80	\$7,923.10	\$8,353.62
4812	IT Training Project Manager	SS	\$38.56025	\$40.77449	\$43.10786	\$45.51132	\$47.98483
4863	IT Training Supervisor	SS	\$33.88648	\$35.79242	\$37.79646	\$39.96867	\$42.24600
2052	Labor Compliance Officer	SS	\$30.19030	\$31.90557	\$33.64292	\$35.60112	\$37.60351
4116	Landscaping Supervisor	SS	\$31.72829	\$33.53612	\$35.42105	\$37.42509	\$39.53423
5135	Logistics Supervisor	SS	\$29.12174	\$30.75405	\$32.47818	\$34.34890	\$36.25504
3056	Machinist Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$38.68637
2214	Mail Supervisor	SS	\$28.99550	\$30.62817	\$32.37294	\$34.19479	\$36.13578
3158	Maintenance Materiel Coordinator	SS	\$27.94444	\$29.52804	\$31.20976	\$32.95454	\$34.82544
3020	Maintenance Planner	SS	\$39.30299	\$41.53127	\$43.80158	\$46.24005	\$48.76962
2183	Marketing Representative	SS	\$28.85536	\$30.47402	\$32.19777	\$34.00561	\$35.92556
1959	Masterplan Coordinator	SS	\$39.30299	\$41.53127	\$43.80158	\$46.24005	\$48.76962

Class	O		0. 4	0. 0	.	Q1 4	0. 5
Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
1757	Mechanical Engineer	SS	\$38.65834	\$40.80253	\$43.11487	\$45.50431	\$47.97081
2385	Medical Administrative Assistant	SS	\$26.15762	\$27.61511	\$29.20572	\$30.85940	\$32.61820
3140	Metal Technical Supervisor	SS	\$6,842.33	\$7,230.24	\$7,624.87	\$8,049.79	\$8,489.28
2561	Microfilm Supervisor	SS	\$19.19953	\$20.26462	\$21.44182	\$22.64704	\$23.89431
4844	Mobile Device Management Specialist	SS	\$0.00000	\$0.00000	\$7,896.19	\$8,330.08	\$8,791.98
3081	Musical Instrument Repair Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$38.32200
3588	Network Operations Supervisor	SS	\$35.37199	\$37.36903	\$39.48518	\$41.73446	\$43.99777
1178	Network Security Administrator	SS	\$36.45110	\$38.51119	\$40.70442	\$43.00276	\$45.35716
4317	Nutrition Specialist	SS	\$30.74029	\$32.47806	\$34.30691	\$36.24789	\$38.28696
5387	Operations Compliance Manager (LA's BEST)	SS	\$39.56226	\$41.73446	\$44.02580	\$46.45727	\$49.01488
5387A	Operations Compliance Manager (LA's BEST)	SS	\$50.44434	\$51.85978	\$0.00000	\$0.00000	\$0.00000
4009	Operations Training Specialist	SS	\$6,670.79	\$7,047.49	\$7,443.25	\$7,854.71	\$8,280.75
4591	Outdoor Education Center Manager	SS	\$4,579.86	\$4,833.23	\$5,105.67	\$5,393.80	\$5,699.88
5307	Outdoor Education Programs Coordinator	SS	\$21.09146	\$22.29668	\$23.52994	\$24.88232	\$26.31178
5370	Out-of-School Program Coordinator	SS	\$21.32270	\$22.52793	\$23.80323	\$25.14158	\$26.54300
5305	Out-of-School Regional Director	SS	\$28.99550	\$30.62182	\$32.37294	\$34.19479	\$36.13578
5303	Out-of-School Senior Director	SS	\$30.74029	\$32.48017	\$34.30691	\$36.24789	\$38.28696
3147	Painting Technical Supervisor	SS	\$6,842.33	\$7,230.24	\$7,624.87	\$8,049.79	\$8,489.28
3114	Paving Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$44.06084
1338	Payroll Distribution Supervisor	SS	\$19.19953	\$20.26462	\$21.44182	\$22.64704	\$23.89431
4137	Pest Management Supervisor	SS	\$26.54300	\$28.04254	\$29.66118	\$31.32187	\$33.05963
4054	Plant Manager I	SS	\$16.50179	\$17.44776	\$18.38671	\$19.45881	\$20.56593
4047	Plant Manager II	SS	\$19.45881	\$20.56593	\$21.72912	\$22.94134	\$24.23066
4037	Plant Manager III	SS	\$21.72912	\$22.94134	\$24.23066	\$25.63909	\$27.02650
4021	Plant Manager IV	SS	\$24.23066	\$25.63909	\$27.02650	\$28.57507	\$30.20073
3152	Plumbing Technical Supervisor	SS	\$6,842.33	\$7,230.24	\$7,624.87	\$8,049.79	\$8,489.28
1121	Principal Accountant	SS	\$32.02259	\$33.88648	\$35.79242	\$37.79646	\$39.96867
5101	Principal Administrative Analyst	SS	\$6,357.99	\$6,712.27	\$7,097.94	\$7,503.80	\$7,923.10
2716	Principal Assignment Technician	SS	\$22.61201	\$23.88731	\$25.20465	\$26.66214	\$28.17568
1220	Principal Auditor, Inspector General's Office	SS	\$38.84341	\$41.03829	\$43.35678	\$45.80646	\$48.39443
2551	Principal Clerk	SS	\$20.26462	\$21.44182	\$22.64704	\$23.89431	\$25.24670
1116	Principal Financial Analyst	SS	\$6,357.99	\$6,712.27	\$7,097.94	\$7,503.80	\$7,923.10
4118	Principal Gardener	SS	\$21.21059	\$22.38077	\$23.60702	\$25.00144	\$26.40987
2251	Principal Personnel Clerk	SS	\$21.44182	\$22.64704	\$23.89431	\$25.24670	\$26.70417
2141	Principal Realty Agent	SS	\$7,149.51	\$7,550.88	\$7,972.43	\$8,401.83	\$8,868.23
5288	Principal Stock Clerk	SS	\$20.18053	\$21.32270	\$22.52793	\$23.80323	\$25.14158
5199	Procurement Customer Services Supervisor	SS	\$23.40382	\$24.69313	\$26.12258	\$27.58707	\$29.17769
1425	Program and Project Controls Manager	SS	\$8,699.53	\$9,177.66	\$9,684.42	\$10,211.35	\$10,779.77
5090	Program Evaluation and Research Coordinator	SS	\$6,665.18	\$7,056.46	\$7,433.16	\$7,842.38	\$8,278.50
1381	Quality Control and Assurance Audit Specialist	SS	\$38.84341	\$41.03829	\$43.35678	\$45.80646	\$48.39446

Class	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
Code	Class Tille		-	•	•	•	•
3523	Radio Communications Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$39.31701
5211	Receiving Inspector	SS	\$21.32270	\$22.52793	\$23.80323	\$25.14158	\$26.54300
2198	Records Manager	SS	\$6,051.92	\$6,388.26	\$6,753.75	\$7,137.19	\$7,536.31
3226	Refrigeration Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$47.29813
1663	Relocatable Housing Manufacturing Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$51.10300
4759	Reprographic Services Coordinator	SS	\$23.55798	\$24.90334	\$26.31178	\$27.81131	\$29.38790
1048	Retirement Systems Supervisor	SS	\$34.55898	\$36.52136	\$38.56025	\$40.77438	\$43.10805
3280	Roofing Inspection Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$41.33506
3281	Roofing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$41.33506
3482	Roofing Technical Supervisor	SS	\$6,842.33	\$7,230.24	\$7,624.87	\$8,049.79	\$8,489.28
4889	SAP Workflow Analyst	SS	\$39.02271	\$41.24397	\$43.58435	\$46.03685	\$48.57343
4325	Satellite Food Service Manager	SS	\$14.56081	\$15.40868	\$16.29158	\$17.19550	\$18.15548
2500	School Administrative Assistant	SS	\$18.15548	\$19.19953	\$20.26462	\$21.44182	\$22.64704
2500A	School Administrative Assistant	SS	\$23.89431	\$25.24670	\$0.00000	\$0.00000	\$0.00000
1775	School Traffic Safety Coordinator	SS	\$26.20667	\$27.68517	\$29.24776	\$30.68422	\$32.65322
1100	Senior Accounting Analyst	SS	\$32.02259	\$33.88648	\$35.79242	\$37.79646	\$39.96867
5070	Senior Administrative Analyst	SS	\$33.88648	\$35.79242	\$37.79646	\$39.96867	\$42.24600
2062	Senior Administrative Assistant	SS	\$28.73625	\$30.36890	\$32.02259	\$33.88648	\$35.79242
2544	Senior Budget Technician	SS	\$20.18053	\$21.36474	\$22.55596	\$23.81023	\$25.18362
1472	Senior Building Project Estimator	SS	\$36.64029	\$38.72140	\$40.92164	\$43.21298	\$45.66547
5721	Senior Bus Dispatcher	SS	\$32.19777	\$34.00561	\$35.92556	\$37.97164	\$40.15085
2192	Senior Community Outreach Organizer	SS	\$33.88648	\$35.79242	\$37.79646	\$39.96867	\$42.24600
1147	Senior Data Center Equipment Operator	SS	\$21.83422	\$23.06747	\$24.37080	\$25.76522	\$27.23672
2/15	Senior Early Childhood Education Division Fiscal	CC	¢10.0404E	¢10.00/00	<u> ተ</u> ገ1 በ4ገገገ	¢22.22/02	¢ጋጋ 40E10
2615	Technician	SS	\$18.84945	\$19.88629	\$21.04233	\$22.22683	\$23.49510
1665	Senior Electrical Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$41.58732
5033	Senior Employment Services Manager	SS	\$6,520.48	\$6,887.23	\$7,282.41	\$7,696.78	\$8,118.79
1773	Senior Energy Specialist	SS	\$33.45065	\$35.34110	\$37.33494	\$39.44416	\$41.67503
1780	Senior Environmental Health and Safety Officer	SS	\$0.00000	\$0.00000	\$33.44504	\$35.35097	\$37.36202
1114	Senior Financial Analyst	SS	\$33.88648	\$35.79242	\$37.79646	\$39.96867	\$42.24600
1128	Senior Fiscal Specialist	SS	\$33.88648	\$35.79242	\$37.79646	\$39.96867	\$42.24600
4321	Senior Food Service Training Specialist	SS	\$26.37484	\$27.87436	\$29.42294	\$31.09064	\$32.85644
4174	Senior Gardener	SS	\$19.08742	\$20.18053	\$21.32270	\$22.52793	\$23.80323
4998	Senior Human Resources Specialist	SS	\$37.79646	\$39.96867	\$42.24600	\$44.60739	\$47.03186
3860	Senior IT Customer Services Support	SS	\$24.16058	\$25.56902	\$26.98446	\$28.54004	\$30.10263
12/5	Representative	CC	¢22/2002	¢25 50505	¢27 E0707	¢20.70574	¢41.00007
1365	Senior Labor Compliance Officer	SS	\$33.63982	\$35.59505	\$37.59707	\$39.70574	\$41.99096
2182	Senior Marketing Representative	SS	\$33.88648	\$35.79242	\$37.79646	\$39.96867	\$42.24600
2383	Senior Medical Administrative Assistant	SS	\$33.76036	\$35.66629	\$37.62829	\$39.80050	\$42.03578
4309	Senior Nutrition Specialist	SS	\$34.30691	\$36.24789	\$38.28696	\$40.46618	\$42.70145
1686	Senior Painting Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$37.65632

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
4143	Senior Pest Management Technician	SS	\$23.40382	\$24.69313	\$26.12258	\$27.58707	\$29.17769
5779	Senior Power Equipment Mechanic	SS	\$26.86533	\$28.39289	\$29.98351	\$31.67223	\$33.45204
5198	Senior Procurement Customer Services Supervisor	SS	\$28.85536	\$30.36890	\$32.02259	\$33.88648	\$35.79242
3809	Senior Programmer Analyst (COBOL)	SS	\$39.02271	\$41.24397	\$43.58435	\$46.03685	\$48.57343
3810	Senior Programmer Analyst (Oracle)	SS	\$39.02271	\$41.24397	\$43.58435	\$46.03685	\$48.57343
4892	Senior Programmer Analyst (SAP)	SS	\$39.02271	\$41.24397	\$43.58435	\$46.03685	\$48.57343
3811	Senior Programmer Analyst (Visual Basic)	SS	\$39.02271	\$41.24397	\$43.58435	\$46.03685	\$48.57343
3819	Senior Programmer Analyst, Filenet	SS	\$39.02271	\$41.24397	\$43.58435	\$46.03685	\$48.57343
3807	Senior Programmer Analyst, JAVA	SS	\$39.02271	\$41.24397	\$43.58435	\$46.03685	\$48.57343
4436	Senior Radiologic Technologist	SS	\$21.23160	\$22.41582	\$23.70513	\$25.06451	\$26.42389
2144	Senior Realty Agent	SS	\$36.17080	\$38.25193	\$40.33304	\$42.57532	\$44.92972
5164	Senior Stores Supervisor	SS	\$31.32187	\$33.05963	\$34.94456	\$36.93459	\$39.01570
1806	Senior Sustainability Specialist	SS	\$41.84948	\$44.20129	\$46.60466	\$49.20133	\$51.90753
3336	Senior Tractor and Roller Operator	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$35.93957
4131	Senior Tree Surgeon	SS	\$23.80323	\$25.14158	\$26.54300	\$28.04254	\$29.66118
4045	Senior Window/Wall Washer	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$21.82722
1126	Special Education Fiscal Analyst	SS	\$6,357.99	\$6,712.27	\$7,097.94	\$7,503.80	\$7,923.09
5175	Stores Supervisor	SS	\$28.04254	\$29.66118	\$31.32187	\$33.05963	\$34.94456
4974	Strategic Data Analyst	SS	\$6,357.99	\$6,712.27	\$7,097.94	\$7,503.80	\$7,923.09
1836	Structural Engineer	SS	\$38.65834	\$40.80253	\$43.11487	\$45.50431	\$47.97081
4736	Student Records and Data Supervisor	SS	\$20.26462	\$21.44182	\$22.64704	\$23.89431	\$25.24670
2356	Student Support Services Program Coordinator	SS	\$24.30072	\$25.68814	\$27.08257	\$28.64515	\$30.26380
4369	Summer Food Service Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$15.38205
1129	Supervising Accountant	SS	\$27.23672	\$28.73625	\$30.36890	\$32.02259	\$33.88648
1316	Supervising Accounting Technician	SS	\$18.84918	\$19.88624	\$21.04241	\$22.22662	\$23.49491
1440	Supervising Architect	SS	\$42.00073	\$44.38316	\$46.84267	\$49.40027	\$52.13305
2720	Supervising Assignment Technician	SS	\$21.36474	\$22.55596	\$23.81023	\$25.18362	\$26.60608
5758	Supervising Automotive Body Mechanic	SS	\$31.27983	\$33.12971	\$34.95858	\$36.96262	\$39.02271
1631	Supervising Building/Construction Inspector	SS	\$7,781.35	\$8,221.35	\$8,685.66	\$9,176.43	\$9,694.69
1527	Supervising Civil Engineer	SS	\$43.04481	\$45.47628	\$48.00585	\$50.61950	\$53.42938
5648	Supervising Classified Training Representative	SS	\$30.12365	\$31.78434	\$33.62020	\$35.51213	\$37.50216
2576	Supervising Clerk	SS	\$17.20250	\$18.15548	\$19.19953	\$20.26462	\$21.44182
2575	Supervising Clerk (Non-Typing)	SS	\$17.20250	\$18.15548	\$19.19953	\$20.26462	\$21.44182
4818	Supervising Data Center Technician	SS	\$27.23672	\$28.73625	\$30.36890	\$32.02259	\$33.88648
1704	Supervising Electrical Engineer	SS	\$43.04481	\$45.47628	\$48.00585	\$50.61950	\$53.42938
1664	Supervising Electrical Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$44.22201
5214	Supervising Fleet-Parts Storekeeper	SS	\$25.43588	\$26.87235	\$28.42793	\$29.99052	\$31.69325
2773	Supervising Insurance Technician	SS	\$17.96628	\$18.98232	\$20.05440	\$21.21059	\$22.38077
2367	Supervising Job Order Contracting Specialist	SS	\$38.74242	\$41.11784	\$43.30406	\$45.72153	\$48.20905
3778	Supervising Maintenance Worker	SS	\$24.62305	\$26.02448	\$27.51701	\$29.06558	\$30.71226

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
1754	Supervising Mechanical Engineer	SS	\$43.04481	\$45.47628	\$48.00585	\$50.61950	\$53.42938
2211	Supervising Microcomputer Support Assistant	SS	\$29.85739	\$31.54610	\$33.35394	\$35.22485	\$37.23589
1303	Supervising Payroll Technician	SS	\$19.88624	\$21.04241	\$22.22662	\$23.49491	\$24.79824
2261	Supervising Personnel Clerk	SS	\$19.19953	\$20.26462	\$21.44182	\$22.64704	\$23.89431
1675	Supervising Plumbing Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$42.69444
5783	Supervising Power Equipment Mechanic	SS	\$30.36890	\$32.02259	\$33.88648	\$35.79242	\$37.79646
5144	Supervising Purchasing Services Coordinator	SS	\$6,378.16	\$6,740.30	\$7,123.73	\$7,522.85	\$7,948.90
4560	Supervising Special Education Assistant	SS	\$20.51688	\$21.67306	\$22.91331	\$24.19562	\$25.52698
5395	Supervising Staff Development Coordinator (LA's BEST)	SS	\$38.74242	\$41.11784	\$43.30406	\$45.72153	\$48.20905
1832	Supervising Structural Engineer	SS	\$43.04481	\$45.47628	\$48.00585	\$50.61950	\$53.42938
1584	Supervisor, Plots Plans and Allotments	SS	\$24.43386	\$25.95441	\$27.39787	\$28.96747	\$30.62817
1864	Survey Party Chief	SS	\$34.82849	\$36.77235	\$38.84685	\$41.04429	\$43.34160
3155	Swimming Pool Technical Supervisor	SS	\$38.34301	\$40.51523	\$42.73649	\$45.11190	\$47.57841
1053	Telecommunications Billing Supervisor	SS	\$32.36593	\$34.18780	\$36.12876	\$38.16083	\$40.31202
2952	Test Services Coordinator	SS	\$26.15762	\$27.61511	\$29.19872	\$30.85940	\$32.62520
5099	Testing Operations Manager	SS	\$33.76036	\$35.66629	\$37.62829	\$39.80050	\$42.03578
1120	Transportation Cost Analyst	SS	\$28.42092	\$30.00453	\$31.67223	\$33.49408	\$35.40002
5741	Transportation Inspector	SS	\$27.29978	\$28.85536	\$30.47402	\$32.19777	\$34.00561
5708	Transportation Planner	SS	\$32.49906	\$34.34895	\$36.25489	\$38.32200	\$40.50121
5728	Transportation Routing Center Supervisor	SS	\$32.49906	\$34.34895	\$36.25489	\$38.32200	\$40.50121
4127	Tree Maintenance Supervisor	SS	\$26.54300	\$28.04254	\$29.66118	\$31.32187	\$33.05963
5736	Truck Dispatcher	SS	\$29.12163	\$30.75430	\$32.47806	\$34.34895	\$36.25489
1180	Web Architect	SS	\$7,661.88	\$8,085.67	\$8,525.15	\$8,994.91	\$9,491.58
2047	Workers' Compensation Claims Processing Supervisor	SS	\$30.12365	\$31.78434	\$33.62020	\$35.51213	\$37.50216

Class	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
Code			•	-	•	-	•
2375	Absence Coordinator	SS	\$39.33146	\$41.58998	\$43.97002	\$46.42155	\$48.94453
1161	Accountant	SS	\$22.27090	\$23.52882	\$24.85822	\$26.28052	\$27.78145
1101	Accounting Analyst	SS	\$29.31098	\$30.97628	\$32.66304	\$34.56421	\$36.50827
1127	Accounting System Specialist	SS	\$30.11860	\$31.83396	\$33.61363	\$35.52195	\$37.56607
2170	Administrative Assistant, Building Program	SS	\$35.58628	\$37.57322	\$39.72454	\$41.95449	\$44.31309
2406	Administrative Secretary I	SS	\$23.87190	\$25.18699	\$26.64503	\$28.13881	\$29.76124
2150	Adult Education Administrative Specialist	SS	\$7,501.77	\$7,920.32	\$8,351.44	\$8,807.73	\$9,294.89
1444	Architect	SS	\$38.65960	\$40.80378	\$43.11234	\$45.50669	\$47.98679
1438	Architectural Associate	SS	\$33.67795	\$35.58628	\$37.57322	\$39.72454	\$41.95449
1460	Architectural Project Facilitator	SS	\$32.77025	\$34.62854	\$36.55115	\$38.65960	\$40.80378
5804	Area Bus Supervisor	SS	\$27.84578	\$29.43247	\$31.08350	\$32.84173	\$34.68572
3271	Area Carpentry Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$38.28079
3206	Area Electrical Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$45.10645
4314	Area Food Services Supervisor	SS	\$5,458.23	\$5,766.99	\$6,091.76	\$6,435.98	\$6,798.48
3211	Area Heating and Air Conditioning Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$47.16485
3236	Area Metal Work Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$44.38457
4016	Area Operations Supervisor	SS	\$5,052.26	\$5,329.00	\$5,633.19	\$5,954.54	\$6,289.60
3276	Area Painting Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$36.28671
3216	Area Plumbing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$46.53590
3165	Asbestos Abatement Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$42.88364
5806	Assistant Area Bus Supervisor	SS	\$26.40203	\$27.84578	\$29.43247	\$31.08350	\$32.84173
1541	Assistant Civil Engineer	SS	\$31.87684	\$33.67795	\$35.58628	\$37.57322	\$39.72454
2057	Assistant Contracts Supervisor	SS	\$32.66304	\$34.56421	\$36.50827	\$38.55239	\$40.76804
4329	Assistant Food Production Manager	SS	\$29.66118	\$31.35510	\$33.12762	\$34.99305	\$36.97285
5755	Assistant Garage Supervisor	SS	\$34.12823	\$36.07943	\$38.11641	\$40.27488	\$42.56915
4173	Assistant Gardening Supervisor	SS	\$25.12267	\$26.53783	\$28.06735	\$29.64689	\$31.31934
3258	Assistant Hardware Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$34.56421
5140	Assistant Logistics Supervisor	SS	\$26.60210	\$28.11042	\$29.70417	\$31.36913	\$33.12774
2218	Assistant Mail Supervisor	SS	\$26.53067	\$27.99587	\$29.57541	\$31.24073	\$33.02040
4574	Assistant Outdoor-Education Center Manager	SS	\$24.00054	\$25.37997	\$26.83802	\$28.31750	\$29.91849
4061	Assistant Plant Manager I	SS	\$15.09503	\$15.90269	\$16.83183	\$17.79672	\$18.75444
4048	Assistant Plant Manager II	SS	\$16.83183	\$17.79672	\$18.75444	\$19.84799	\$20.97725
1791	Assistant Projects Manager	SS	\$6,979.18	\$7,374.84	\$7,777.37	\$8,210.79	\$8,659.07
1632	Assistant Supervising Building/Construction Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$52.21940
1531	Associate Civil Engineer	SS	\$35.58628	\$37.57322	\$39.72454	\$41.95449	\$44.31309
1709	Associate Electrical Engineer	SS	\$33.67795	\$35.58628	\$37.57322	\$39.72454	\$41.95449
1763	Associate Mechanical Engineer	SS	\$33.67795	\$35.58628	\$37.57322	\$39.72454	\$41.95449
1471	Building Project Estimator	SS	\$31.87684	\$33.67795	\$35.58628	\$37.57322	\$39.72454
1651	Building/Construction Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$47.40072
1650	Building/Construction Inspector (Class 1)	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$48.69437
5731	Bus Dispatcher	SS	\$29.43247	\$31.08350	\$32.84173	\$34.68572	\$36.64407
	1						

3192 Carpentry Supervisor (Building Moving) SS \$0.00000 \$0.00000 \$0.00000 \$38.28079 3148 Carpentry Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 2620 Cataloging Supervisor SS \$20.66991 \$21.87066 \$23.09998 \$24.37220 \$25.75163 3160 Central Shops Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$39.35288 2110 Chief Broadcasting Engineer SS \$7,428.59 \$7,849.42 \$8,270.25 \$8,732.25 \$9,209.12 2511 Chief Clerk SS \$23.87190 \$25.18699 \$26.64503 \$28.13881 \$29.76124 1501 Chief Enrollment Analysis Coordinator SS \$40.89670 \$43.22671 \$45.59245 \$48.12259 \$50.75992 5192 Chief Inventory Analyst SS \$39.43151 \$41.61858 \$43.97717 \$46.41440 \$48.93023	Class	Class Title	Unit	Step 1	Stop 2	Stop 2	Stop /	Step 5
3148Carpentry Technical SupervisorSS\$6,979.18\$7,374.84\$7,777.37\$8,210.79\$8,659.072620Cataloging SupervisorSS\$20.66991\$21.87066\$23.09998\$24.37220\$25.751633160Central Shops SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$39.352882110Chief Broadcasting EngineerSS\$7,428.59\$7,849.42\$8,270.25\$8,732.25\$9,209.122511Chief ClerkSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241501Chief Enrollment Analysis CoordinatorSS\$40.89670\$43.22671\$45.59245\$48.12259\$50.759925192Chief Inventory AnalystSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.041528Civil EngineerSS\$39.43151\$41.61858\$43.97717\$46.41440\$48.930232045Claims Processing SupervisorSS\$30.97628\$32.66304\$34.56421\$36.50827\$38.552392056Communications Production FacilitatorSS\$30.84763\$32.57012\$34.43557\$36.38677\$38.423744817Computer Applications Support SupervisorSS\$33.98529\$35.92935\$37.93772\$40.10335\$42.347602531Contract AssistantSS\$21.65623\$22.86414\$24.17923\$25.56580\$26.952371125Contract Payments SupervisorSS\$28.19600\$29.76124\$31.43370\$33.16334\$35.078		Class Title		-	-	•	•	-
2620Cataloging SupervisorSS\$20.66991\$21.87066\$23.09998\$24.37220\$25.751633160Central Shops SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$39.352882110Chief Broadcasting EngineerSS\$7,428.59\$7,849.42\$8,270.25\$8,732.25\$9,209.122511Chief ClerkSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241501Chief Enrollment Analysis CoordinatorSS\$40.89670\$43.22671\$45.59245\$48.12259\$50.759925192Chief Inventory AnalystSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.041528Civil EngineerSS\$39.43151\$41.61858\$43.97717\$46.41440\$48.930232045Claims Processing SupervisorSS\$30.97628\$32.66304\$34.56421\$36.50827\$38.552392056Communications Production FacilitatorSS\$30.84763\$32.57012\$34.43557\$36.38677\$38.423744817Computer Applications Support SupervisorSS\$33.98529\$35.92935\$37.14438\$39.25282\$41.418451792Construction Safety SpecialistSS\$21.65623\$22.86414\$24.17923\$25.56580\$26.952372531Contract AssistantSS\$28.19600\$29.76124\$31.43370\$33.16334\$35.078812534Coordinating Contract AssistantSS\$24.17923\$25.56580\$26.95237\$28.50333\$30.								
3160Central Shops SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$39.352882110Chief Broadcasting EngineerSS\$7,428.59\$7,849.42\$8,270.25\$8,732.25\$9,209.122511Chief ClerkSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241501Chief Enrollment Analysis CoordinatorSS\$40.89670\$43.22671\$45.59245\$48.12259\$50.759925192Chief Inventory AnalystSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.041528Civil EngineerSS\$39.43151\$41.61858\$43.97717\$46.41440\$48.930232045Claims Processing SupervisorSS\$30.97628\$32.66304\$34.56421\$36.50827\$38.552392056Communications Production FacilitatorSS\$30.84763\$32.57012\$34.43557\$36.38677\$38.423744817Computer Applications Support SupervisorSS\$33.27770\$35.15029\$37.14438\$39.25282\$41.418451792Construction Safety SpecialistSS\$33.98529\$35.92935\$37.93772\$40.10335\$42.347602531Contract AssistantSS\$21.65623\$22.86414\$24.17923\$25.56580\$26.95237125Contract Payments SupervisorSS\$28.19600\$29.76124\$31.43370\$33.16334\$35.078812534Coordinating Contract AssistantSS\$24.17923\$25.56580\$26.95237\$28.50333 <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>								
2110Chief Broadcasting EngineerSS\$7,428.59\$7,849.42\$8,270.25\$8,732.25\$9,209.122511Chief ClerkSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241501Chief Enrollment Analysis CoordinatorSS\$40.89670\$43.22671\$45.59245\$48.12259\$50.759925192Chief Inventory AnalystSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.041528Civil EngineerSS\$39.43151\$41.61858\$43.97717\$46.41440\$48.930232045Claims Processing SupervisorSS\$30.97628\$32.66304\$34.56421\$36.50827\$38.552392056Communications Production FacilitatorSS\$30.84763\$32.57012\$34.43557\$36.38677\$38.423744817Computer Applications Support SupervisorSS\$33.27770\$35.15029\$37.14438\$39.25282\$41.418451792Construction Safety SpecialistSS\$33.98529\$35.92935\$37.93772\$40.10335\$42.347602531Contract AssistantSS\$21.65623\$22.86414\$24.17923\$25.56580\$25.56580\$26.952371125Contract Payments SupervisorSS\$28.19600\$29.76124\$31.43370\$33.16334\$35.078812534Coordinating Contract AssistantSS\$24.17923\$25.56580\$26.95237\$28.50333\$30.11860		0 0 1						
2511Chief ClerkSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241501Chief Enrollment Analysis CoordinatorSS\$40.89670\$43.22671\$45.59245\$48.12259\$50.759925192Chief Inventory AnalystSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.041528Civil EngineerSS\$39.43151\$41.61858\$43.97717\$46.41440\$48.930232045Claims Processing SupervisorSS\$30.97628\$32.66304\$34.56421\$36.50827\$38.552392056Communications Production FacilitatorSS\$30.84763\$32.57012\$34.43557\$36.38677\$38.423744817Computer Applications Support SupervisorSS\$33.27770\$35.15029\$37.14438\$39.25282\$41.418451792Construction Safety SpecialistSS\$33.98529\$35.92935\$37.93772\$40.10335\$42.347602531Contract AssistantSS\$21.65623\$22.86414\$24.17923\$25.56580\$26.952371125Contract Payments SupervisorSS\$28.19600\$29.76124\$31.43370\$33.16334\$35.078812534Coordinating Contract AssistantSS\$24.17923\$25.56580\$26.95237\$28.50333\$30.11860								
1501Chief Enrollment Analysis CoordinatorSS\$40.89670\$43.22671\$45.59245\$48.12259\$50.759925192Chief Inventory AnalystSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.041528Civil EngineerSS\$39.43151\$41.61858\$43.97717\$46.41440\$48.930232045Claims Processing SupervisorSS\$30.97628\$32.66304\$34.56421\$36.50827\$38.552392056Communications Production FacilitatorSS\$30.84763\$32.57012\$34.43557\$36.38677\$38.423744817Computer Applications Support SupervisorSS\$33.27770\$35.15029\$37.14438\$39.25282\$41.418451792Construction Safety SpecialistSS\$33.98529\$35.92935\$37.93772\$40.10335\$42.347602531Contract AssistantSS\$21.65623\$22.86414\$24.17923\$25.56580\$26.952371125Contract Payments SupervisorSS\$28.19600\$29.76124\$31.43370\$33.16334\$35.078812534Coordinating Contract AssistantSS\$24.17923\$25.56580\$26.95237\$28.50333\$30.11860		3 3						
5192Chief Inventory AnalystSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.041528Civil EngineerSS\$39.43151\$41.61858\$43.97717\$46.41440\$48.930232045Claims Processing SupervisorSS\$30.97628\$32.66304\$34.56421\$36.50827\$38.552392056Communications Production FacilitatorSS\$30.84763\$32.57012\$34.43557\$36.38677\$38.423744817Computer Applications Support SupervisorSS\$33.27770\$35.15029\$37.14438\$39.25282\$41.418451792Construction Safety SpecialistSS\$33.98529\$35.92935\$37.93772\$40.10335\$42.347602531Contract AssistantSS\$21.65623\$22.86414\$24.17923\$25.56580\$26.952371125Contract Payments SupervisorSS\$28.19600\$29.76124\$31.43370\$33.16334\$35.078812534Coordinating Contract AssistantSS\$24.17923\$25.56580\$26.95237\$28.50333\$30.11860								
1528 Civil Engineer SS \$39.43151 \$41.61858 \$43.97717 \$46.41440 \$48.93023 2045 Claims Processing Supervisor SS \$30.97628 \$32.66304 \$34.56421 \$36.50827 \$38.55239 2056 Communications Production Facilitator SS \$30.84763 \$32.57012 \$34.43557 \$36.38677 \$38.42374 4817 Computer Applications Support Supervisor SS \$33.27770 \$35.15029 \$37.14438 \$39.25282 \$41.41845 1792 Construction Safety Specialist SS \$33.98529 \$35.92935 \$37.93772 \$40.10335 \$42.34760 2531 Contract Assistant SS \$21.65623 \$22.86414 \$24.17923 \$25.56580 \$26.95237 1125 Contract Payments Supervisor SS \$28.19600 \$29.76124 \$31.43370 \$33.16334 \$35.07881 2534 Coordinating Contract Assistant SS \$24.17923 \$25.56580 \$26.95237 \$28.50333 \$30.11860		,					·	
2045 Claims Processing Supervisor SS \$30.97628 \$32.66304 \$34.56421 \$36.50827 \$38.55239 2056 Communications Production Facilitator SS \$30.84763 \$32.57012 \$34.43557 \$36.38677 \$38.42374 4817 Computer Applications Support Supervisor SS \$33.27770 \$35.15029 \$37.14438 \$39.25282 \$41.41845 1792 Construction Safety Specialist SS \$33.98529 \$35.92935 \$37.93772 \$40.10335 \$42.34760 2531 Contract Assistant SS \$21.65623 \$22.86414 \$24.17923 \$25.56580 \$26.95237 1125 Contract Payments Supervisor SS \$28.19600 \$29.76124 \$31.43370 \$33.16334 \$35.07881 2534 Coordinating Contract Assistant SS \$24.17923 \$25.56580 \$26.95237 \$28.50333 \$30.11860		, ,						
2056 Communications Production Facilitator SS \$30.84763 \$32.57012 \$34.43557 \$36.38677 \$38.42374 4817 Computer Applications Support Supervisor SS \$33.27770 \$35.15029 \$37.14438 \$39.25282 \$41.41845 1792 Construction Safety Specialist SS \$33.98529 \$35.92935 \$37.93772 \$40.10335 \$42.34760 2531 Contract Assistant SS \$21.65623 \$22.86414 \$24.17923 \$25.56580 \$26.95237 1125 Contract Payments Supervisor SS \$28.19600 \$29.76124 \$31.43370 \$33.16334 \$35.07881 2534 Coordinating Contract Assistant SS \$24.17923 \$25.56580 \$26.95237 \$28.50333 \$30.11860		•						
4817 Computer Applications Support Supervisor SS \$33.27770 \$35.15029 \$37.14438 \$39.25282 \$41.41845 1792 Construction Safety Specialist SS \$33.98529 \$35.92935 \$37.93772 \$40.10335 \$42.34760 2531 Contract Assistant SS \$21.65623 \$22.86414 \$24.17923 \$25.56580 \$26.95237 1125 Contract Payments Supervisor SS \$28.19600 \$29.76124 \$31.43370 \$33.16334 \$35.07881 2534 Coordinating Contract Assistant SS \$24.17923 \$25.56580 \$26.95237 \$28.50333 \$30.11860		U		\$30.97628		\$34.56421		\$38.55239
1792 Construction Safety Specialist SS \$33.98529 \$35.92935 \$37.93772 \$40.10335 \$42.34760 2531 Contract Assistant SS \$21.65623 \$22.86414 \$24.17923 \$25.56580 \$26.95237 1125 Contract Payments Supervisor SS \$28.19600 \$29.76124 \$31.43370 \$33.16334 \$35.07881 2534 Coordinating Contract Assistant SS \$24.17923 \$25.56580 \$26.95237 \$28.50333 \$30.11860	2056	Communications Production Facilitator		\$30.84763	\$32.57012	\$34.43557	\$36.38677	\$38.42374
2531 Contract Assistant SS \$21.65623 \$22.86414 \$24.17923 \$25.56580 \$26.95237 1125 Contract Payments Supervisor SS \$28.19600 \$29.76124 \$31.43370 \$33.16334 \$35.07881 2534 Coordinating Contract Assistant SS \$24.17923 \$25.56580 \$26.95237 \$28.50333 \$30.11860		Computer Applications Support Supervisor		\$33.27770	\$35.15029		\$39.25282	\$41.41845
1125 Contract Payments Supervisor SS \$28.19600 \$29.76124 \$31.43370 \$33.16334 \$35.07881 2534 Coordinating Contract Assistant SS \$24.17923 \$25.56580 \$26.95237 \$28.50333 \$30.11860		Construction Safety Specialist		\$33.98529	\$35.92935	\$37.93772	\$40.10335	\$42.34760
2534 Coordinating Contract Assistant SS \$24.17923 \$25.56580 \$26.95237 \$28.50333 \$30.11860	2531	Contract Assistant		\$21.65623	\$22.86414	\$24.17923	\$25.56580	\$26.95237
	1125	Contract Payments Supervisor		\$28.19600	\$29.76124	\$31.43370	\$33.16334	\$35.07881
2138 Coordinator of Telecommunication Services SS \$33.27770 \$35.15029 \$37.14438 \$39.25282 \$41.41845	2534			\$24.17923	\$25.56580	\$26.95237	\$28.50333	\$30.11860
	2138	Coordinator of Telecommunication Services		\$33.27770		\$37.14438	\$39.25282	\$41.41845
1990 Cost Recovery Claims Processing Supervisor SS \$30.72612 \$32.42003 \$34.29260 \$36.22237 \$38.25220	1990	Cost Recovery Claims Processing Supervisor	SS	\$30.72612	\$32.42003	\$34.29260	\$36.22237	\$38.25220
2236 Credentials and Contract Supervisor SS \$24.37220 \$25.75163 \$27.23825 \$28.77492 \$30.39021	2236	Credentials and Contract Supervisor	SS	\$24.37220	\$25.75163	\$27.23825	\$28.77492	\$30.39021
1301 Customer Services Center Supervisor SS \$19.16479 \$20.24983 \$21.38446 \$22.62450 \$23.90174	1301	Customer Services Center Supervisor	SS	\$19.16479	\$20.24983	\$21.38446	\$22.62450	\$23.90174
1073 Database Specialist SS \$0.00000 \$0.00000 \$8,054.11 \$8,496.68 \$8,967.82	1073	Database Specialist		\$0.00000	\$0.00000	\$8,054.11	\$8,496.68	\$8,967.82
4881 Design Network Engineer SS \$42.35475 \$44.74908 \$47.19344 \$49.78790 \$52.55390	4881	Design Network Engineer		\$42.35475	\$44.74908	\$47.19344	\$49.78790	\$52.55390
4600 Disability Support Services Program Assistant SS \$20.92722 \$22.10652 \$23.37158 \$24.67953 \$26.03752		Disability Support Services Program Assistant		\$20.92722	\$22.10652	\$23.37158	\$24.67953	\$26.03752
4624 Display and Graphic Designer SS \$28.50333 \$30.11860 \$31.83396 \$33.61363 \$35.52195		Display and Graphic Designer			\$30.11860	\$31.83396	\$33.61363	\$35.52195
5739 Driver Trainer (Truck) SS \$29.43247 \$31.08350 \$32.84173 \$34.68572 \$36.64407		Driver Trainer (Truck)			\$31.08350	\$32.84173	\$34.68572	\$36.64407
5734 Driver-Trainer (Bus-Truck) SS \$29.43247 \$31.08350 \$32.84173 \$34.68572 \$36.64407	5734	Driver-Trainer (Bus-Truck)		\$29.43247	\$31.08350	\$32.84173	\$34.68572	\$36.64407
3126 Electrical Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07		Electrical Technical Supervisor		\$6,979.18	\$7,374.84		\$8,210.79	\$8,659.07
3522 Electronics Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07		Electronics Technical Supervisor		\$6,979.18	\$7,374.84	\$7,777.37	\$8,210.79	\$8,659.07
1 3	5034	Employment Services Manager		\$34.43557	\$36.37248	\$38.45949	\$40.64655	\$42.87650
1778 Environmental Assessment Coordinator SS \$33.98529 \$35.92935 \$37.93772 \$40.10335 \$42.34760	1778			\$33.98529	\$35.92935	\$37.93772		\$42.34760
,	3015	Environmental Compliance Coordinator			\$30.45454	\$32.16272		\$35.92935
1801 Environmental Health Supervisor SS \$37.95759 \$40.12146 \$42.36591 \$44.76539 \$47.28887		Environmental Health Supervisor		\$37.95759		\$42.36591	\$44.76539	\$47.28887
2143 Escrow and Title Supervisor SS \$41.89018 \$44.24162 \$46.71457 \$49.23041 \$51.96067		Escrow and Title Supervisor		\$41.89018			\$49.23041	\$51.96067
		Ethics Adviser			\$35.31467	\$37.28017		\$41.62572
1963 Facilities Cost Analyst SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804	1963	3	SS	\$32.66304	\$34.56421	\$36.50827	\$38.55239	\$40.76804
Facilities Partnership and Development SS \$40.08905 \$42.36190 \$44.67761 \$47.16485 \$49.74501 Coordinator	1435	· · · · · · · · · · · · · · · · · · ·	SS	\$40.08905	\$42.36190	\$44.67761	\$47.16485	\$49.74501
	1813		SS	\$32.11270	\$33.94241	\$35.84358	\$37.87341	\$40.01043
								\$44.87773
3251 Floor Covering Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$37.96631		· · · · · · · · · · · · · · · · · · ·						
0 1		U .						\$49.76275
								\$23.70035

Class							
Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
3121	Food Service Equipment Specialist	SS	\$38.55239	\$40.76804	\$43.09092	\$45.49954	\$47.97250
4291	Food Service Manager I	SS	\$17.04625	\$18.00398	\$19.01890	\$20.10528	\$21.27743
4292	Food Service Manager II	SS	\$18.00398	\$19.01890	\$20.10528	\$21.27743	\$22.45673
4293	Food Service Manager III	SS	\$19.01890	\$20.10528	\$21.27743	\$22.45673	\$23.70035
4296	Food Service Manager IV	SS	\$20.10528	\$21.27743	\$22.45673	\$23.70035	\$25.03691
4297	Food Service Manager V	SS	\$21.27743	\$22.45673	\$23.70035	\$25.03691	\$26.45207
4294	Food Service Manager VI	SS	\$22.45673	\$23.70035	\$25.03691	\$26.45207	\$27.95299
4295	Food Service Manager VII	SS	\$23.70035	\$25.03691	\$26.45207	\$27.95299	\$29.51824
4312	Food Service Training Specialist	SS	\$24.12205	\$25.48004	\$26.90234	\$28.43185	\$30.01140
1071	Food Services Cost Analyst	SS	\$32.66304	\$34.56421	\$36.50827	\$38.55239	\$40.76804
5756	Garage Supervisor	SS	\$36.07943	\$38.11641	\$40.27488	\$42.56915	\$44.87773
3266	Glazing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	40.11764
3051	Graphic Arts Machinist Supervisor	SS	\$35.09311	\$37.08720	\$39.17421	\$41.39700	\$43.72701
3256	Hardware Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	38.58813
5201	Head Stock Clerk	SS	\$21.74915	\$22.97849	\$24.27929	\$25.64441	\$27.07386
3137	Heating and Air Conditioning Technical Supervisor	SS	\$6,979.18	\$7,374.84	\$7,777.37	\$8,210.79	\$8,659.07
3132	HVAC Technical Supervisor (High-Rise)	SS	\$40.08905	\$42.36190	\$44.67761	\$47.16485	\$49.74501
3134	HVAC Test and Inspection Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	47.16485
1782	Industrial Hygienist	SS	\$33.98529	\$35.92935	\$37.93750	\$40.10335	\$42.34760
4620	Instructional Media Technician-Adviser	SS	\$31.15496	\$32.94179	\$34.79292	\$36.75842	\$38.85973
4156	Integrated Pest Management Program	SS	\$32.11270	\$33.94241	\$35.84358	\$37.87341	\$40.01043
	Coordinator						
1151	Intermediate Accountant	SS	\$24.85822	\$26.28052	\$27.78145	\$29.31098	\$30.97628
3031	IT Electronics Communications Supervisor	SS	\$34.12823	\$36.05799	\$38.12355	\$40.25343	\$42.51914
3543	IT Field Services Supervisor	SS	\$30.45454	\$32.17702	\$34.02102	\$35.92935	\$37.98061
4880	IT Infrastructure Project Network Engineer	SS	\$34.30690	\$36.24382	\$38.28794	\$40.45357	\$42.74069
1193	IT Project Manager	SS	\$6,846.52	\$7,239.90	\$7,653.88	\$8,081.56	\$8,520.69
4812	IT Training Project Manager	SS	\$39.33146	\$41.58998	\$43.97002	\$46.42155	\$48.94453
4863	IT Training Supervisor	SS	\$34.56421	\$36.50827	\$38.55239	\$40.76804	\$43.09092
2052	Labor Compliance Officer	SS	\$30.79411	\$32.54368	\$34.31578	\$36.31314	\$38.35558
4116	Landscaping Supervisor	SS	\$32.36286	\$34.20684	\$36.12947	\$38.17359	\$40.32491
5135	Logistics Supervisor	SS	\$29.70417	\$31.36913	\$33.12774	\$35.03588	\$36.98014
3056	Machinist Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$39.4601
2214	Mail Supervisor	SS	\$29.57541	\$31.24073	\$33.02040	\$34.87869	\$36.85850
3158	Maintenance Materiel Coordinator	SS	\$28.50333	\$30.11860	\$31.83396	\$33.61363	\$35.52195
3020	Maintenance Planner	SS	\$40.08905	\$42.36190	\$44.67761	\$47.16485	\$49.74501
2183	Marketing Representative	SS	\$29.43247	\$31.08350	\$32.84173	\$34.68572	\$36.64407
1959	Masterplan Coordinator	SS	\$40.08905	\$42.36190	\$44.67761	\$47.16485	\$49.74501
1757	Mechanical Engineer	SS	\$39.43151	\$41.61858	\$43.97717	\$46.41440	\$48.93023
2385	Medical Administrative Assistant	SS	\$26.68077	\$28.16741	\$29.78983	\$31.47659	\$33.27056
3140	Metal Technical Supervisor	SS	\$6,979.18	\$7,374.84	\$7,777.37	\$8,210.79	\$8,659.07

2561 Microfilm Supervisor SS \$19.58352 \$20.66991 \$21.87066 \$23.09998 \$24.37220	Class	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
4844 Mobile Device Management Specialist SS \$0,00000 \$0,00000 \$0,00000 \$3,00000 \$3,00000 \$3,00000 \$3,00000 \$3,00000 \$3,00000 \$3,00000 \$3,00000 \$3,00000 \$3,00000 \$3,000000 \$3,000000 \$3,000000 \$3,000000 \$3,000000 \$3,000000 \$3,000000 \$3,000000 \$3,000000 \$3,000000 \$3,000000 \$44,87773 1,178 Network Operations Compliance Manager (LA'S BEST) SS \$3,13,5510 \$33,12762 \$34,99305 \$36,97285 \$39,02270 5387 Operations Compliance Manager (LA'S BEST) SS \$51,45232 \$52,9698 \$0,00000	Code			-	-	•	-	•
3081 Musical Instrument Répair Supérvisor SS \$0,00000 \$0,00000 \$0,00000 \$3,08844 3588 Network Operations Supervisor SS \$36,07943 \$38,11641 \$40,27488 \$42,56915 \$44,87773 17178 Nutrition Specialist SS \$37,18012 \$39,28141 \$41,51851 \$43,8622 \$46,26430 5387 Operations Compliance Manager (LA's BEST) SS \$51,35323 \$42,56915 \$44,90632 \$47,38642 \$49,99518 5387 Operations Compliance Manager (LA's BEST) SS \$51,45523 \$52,89698 \$0,00000 \$0,00000 \$0,00000 4009 Operations Training Specialist SS \$6,804.21 \$7,188.44 \$7,592.12 \$8,011.80 \$8,46.37 5307 Outdoor Education Center Manager SS \$21,51329 \$22,7426t \$25,207.78 \$5,611.68 \$5,818.88 5307 Out-of-School Programs Coordinator SS \$21,4915 \$22,97849 \$24,27292 \$25,64441 \$27,07386 5305 Out-of-School Regional Director SS		•						
3588 Network Operations Supervisor SS \$36,07943 \$38,11641 \$40,27488 \$42,5915 \$44,8773 1178 Network Security Administrator SS \$37,18012 \$39,28141 \$41,51851 \$43,86282 \$46,26430 5387 Operations Compliance Manager (LA'S BEST) SS \$40,35351 \$32,50752 \$44,90632 \$47,38642 \$49,99518 5387A Operations Compliance Manager (LA'S BEST) SS \$6,80421 \$7,1884 \$47,592,12 \$8,011.00 \$0,00000 4009 Operations Training Specialist SS \$6,804.21 \$7,1884 \$7,592,12 \$8,011.00 \$8,445.37 5307 Outdoor Education Center Manager SS \$4,671.46 \$4,929.89 \$5,207.78 \$5,501.68 \$5,813.88 5307 Out-of-School Program Coordinator SS \$21,51329 \$22,74261 \$24,00054 \$23,3997 \$26,83802 5303 Out-of-School Regional Director SS \$31,35510 \$31,312977 \$34,99305 \$36,85936 5303 Out-of-School Regional Director SS <td></td> <td>·</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>		·						
1178 Network Security Administrator SS \$37.18012 \$39.28141 \$41.51851 \$43.86282 \$46.26430 4317 Nutrition Specialist SS \$31.35510 \$33.12762 \$34.99052 \$36.97285 \$39.05270 5387 Operations Compliance Manager (LA's BEST) SS \$40.35351 \$42.56915 \$44.90632 \$47.38642 \$49.99518 5387A Operations Training Specialist SS \$51.45232 \$52.89698 \$0.0000 \$0.00000 4009 Operations Training Specialist SS \$6.804.21 \$7.188.44 \$7.592.12 \$8.011.80 \$8.46.264 5307 Out-of-School Program Coordinator SS \$21.51329 \$22.74261 \$24.00054 \$25.3797 \$26.83802 5305 Out-of-School Regional Director SS \$29.57541 \$31.23426 \$33.00040 \$34.9686 \$39.05270 3147 Painting Technical Supervisor SS \$6.979.18 \$7.374.84 \$7.777.37 \$8.210.79 \$8.659.07 1338 Payoril Distribution Supervisor SS \$19.881		·						
4317 Nutrition Specialist SS 331,35510 333,12762 334,99305 336,67285 339,05270 5387A Operations Compliance Manager (LA's BEST) SS \$40,35351 \$42,56915 \$44,90632 247,38642 \$49,99518 3377A Operations Compliance Manager (LA's BEST) SS \$51,45323 \$52,89698 \$0,00000 \$0,00000 4099 Operations Training Specialist SS \$6,804.21 \$7,188.44 \$7,592.12 \$8,011.80 \$8,463.74 5307 Outdoor Education Center Manager SS \$4,671.46 \$4,929.89 \$5,207.78 \$5,501.68 \$5,813.88 5307 Out-of-School Program Coordinator SS \$221,73491 \$22,97849 \$24,27929 \$25,64441 \$27,07386 5303 Out-of-School Program Coordinator SS \$29,57541 \$31,23470 \$34,99305 \$34,679369 \$36,85850 5303 Out-of-School Program Coordinator SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$36,85850 5303 Out-of-School Program Coordinator								
5387 Operation's Compliance Manager (LA'S BEST) SS \$40,35351 \$42,56915 \$44,90632 \$47,38642 \$49,99518 5387A Operation's Compliance Manager (LA'S BEST) SS \$6,804.21 \$7,188.44 \$7,592.12 \$8,011.80 \$8,466.37 4591 Outdoor Education Center Manager SS \$4,671.46 \$4,929.89 \$5,207.78 \$5,501.68 \$5,813.88 5307 Outdoor Education Programs Coordinator SS \$21,51329 \$22,74261 \$24,00054 \$25,3797 \$26,83802 5307 Out-of-School Program Coordinator SS \$29,57541 \$31,23426 \$33,00040 \$34,87869 \$30,0000 5303 Out-of-School Senior Director SS \$31,35510 \$33,12977 \$34,9905 \$36,9728 \$39,05270 3141 Painting Technical Supervisor SS \$6,079,18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 1338 Payroll Distribution Supervisor SS \$19,58352 \$20,66991 \$21,87066 \$33,97282 \$39,05204 4137 Peat Manager II<								
5387A Operations Compilance Manager (LA's BEST) SS \$51,45323 \$52,89698 30,00000 30,00000 50,00000 4009 Operations Training Specialist SS \$6,804.21 \$7,818.44 \$7,592.12 \$8,511.80 \$8,446.37 4591 Outdoor Education Programs Coordinator SS \$21,14915 \$22,74261 \$24,00054 \$25,3797 \$26,83802 5307 Out-of-School Program Coordinator SS \$21,14915 \$22,74261 \$24,00054 \$25,3797 \$26,83802 5305 Out-of-School Senior Director SS \$29,57541 \$31,32510 \$33,12977 \$34,99305 \$36,97285 \$39,05270 3147 Painting Technical Supervisor SS \$0,00000 \$0,00000 \$0,00000 \$44,94206 3138 Payroll Distribution Supervisor SS \$19,58382 \$20,66901 \$21,87066 \$23,09998 \$24,37220 4037 Plant Manager II SS \$16,83183 \$17,777,77 \$12,76703 \$21,4657 \$20,46991 \$1,87464 \$19,48799 \$20,97725 \$21,56703<		•						
4009 Operations Training Specialist SS \$6,804.21 \$7,188.44 \$7,592.12 \$8,011.80 \$8,446.37 4591 Outdoor Education Center Manager SS \$4,671.46 \$4,929.89 \$5,207.83 \$5,50.368 \$5,831.88 5307 Out-of-School Program Coordinator SS \$21,51329 \$22,47641 \$24,00054 \$34,87869 \$36,8882 5305 Out-of-School Regional Director SS \$29,57541 \$31,23246 \$33,02040 \$34,87869 \$36,85850 5303 Out-of-School Senior Director SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,201.79 \$36,979.85 \$30,02040 \$34,97305 \$36,97285 \$39,05270 3114 Painting Technical Supervisor SS \$0,00000 \$0,00000 \$0,00000 \$44,9206 1338 Payroll Distribution Supervisor SS \$19,88325 \$20,66991 \$21,8706 \$23,30998 \$24,37220 4054 Plant Manager I SS \$16,83183 \$17,97672 \$21,8706 \$23,0017 \$24,71527 4037 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>								
4591 Outdoor Education Center Manager SS \$4,671.46 \$4,929.89 \$5,207.78 \$5,501.68 \$5,813.88 5307 Outdoor Education Programs Coordinator SS \$21,51329 \$22,274261 \$24,00054 \$25,37979 \$26,83802 5307 Out-of-School Regional Director SS \$21,74915 \$22,97849 \$24,27929 \$25,64441 \$27,07366 5303 Out-of-School Senior Director SS \$31,35510 \$33,12977 \$34,99305 \$36,97285 \$39,05270 3147 Painting Technical Supervisor SS \$6,079.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 3134 Payroll Distribution Supervisor SS \$19,58352 \$20,66991 \$21,87066 \$23,09998 \$24,37220 4137 Pest Management Supervisor SS \$27,07386 \$28,60339 \$30,25440 \$31,94831 \$33,72082 4054 Plant Manager II SS \$16,83183 \$17,777.73 \$21,4570 \$23,40017 \$22,16370 \$23,40017 \$24,1527 \$26,15187 \$27,56703		1 ,						
5307 Outdoor Education Programs Coordinator SS \$21.51329 \$22.74261 \$24.00054 \$25.37977 \$26.83802 5307 Out-of-School Program Coordinator SS \$21.74915 \$22.97849 \$24.27929 \$25.64441 \$27.07386 5305 Out-of-School Senior Director SS \$29.57541 \$31.23426 \$33.02040 \$34.87869 \$36.85850 5303 Out-of-School Senior Director SS \$30.0000 \$33.12977 \$34.99305 \$36.97285 \$39.05270 3114 Painting Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,699.07 3114 Paint Manager Il SS \$19,58352 \$20.60909 \$21.87060 \$23.09998 \$24.37220 4074 Plant Manager II SS \$19,84799 \$20.97725 \$22.16370 \$22.16370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.10370 \$22.								
5370 Out-of-School Program Coordinator SS \$21.74915 \$22.97849 \$24.27929 \$25.64441 \$27.07386 5303 Out-of-School Regional Director SS \$29.57541 \$31.23426 \$33.02040 \$34.87869 \$36.85850 5303 Out-of-School Senior Director SS \$31.35510 \$33.12977 \$34.99305 \$36.97285 \$39.05270 3114 Painting Technical Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$44.94206 1338 Payroll Distribution Supervisor SS \$19.58352 \$20.66991 \$21.87066 \$23.09998 \$24.37220 4137 Pest Management Supervisor SS \$19.58352 \$20.66991 \$21.87066 \$23.09998 \$24.37220 4054 Plant Manager II SS \$19.84799 \$20.97725 \$22.16370 \$23.40017 \$24.71527 4074 Plant Manager III SS \$22.16370 \$24.71527 \$26.15187 \$27.56703 4021 Plant Manager II SS \$24.471527 \$26.15187 \$27.56703		· · · · · · · · · · · · · · · · · · ·						
5305 Out-of-School Regional Director SS \$29.57541 \$31.23426 \$33.02040 \$34.87869 \$36.85850 5303 Out-of-School Senior Director SS \$31.35510 \$33.12977 \$34,99305 \$36.97285 \$39.05270 3147 Painting Technical Supervisor SS \$6,079.18 \$7,777.37 \$8,210.79 \$8,659.07 3138 Payroll Distribution Supervisor SS \$19.58352 \$20.66991 \$21.87066 \$23.09998 \$24.37220 4054 Plant Manager I SS \$27.07386 \$28.60339 \$30.25440 \$31.94831 \$33.7082 4054 Plant Manager II SS \$16.83183 \$17.79672 \$18.75444 \$19.84799 \$20.97725 4047 Plant Manager III SS \$22.16370 \$23.40017 \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3152 Plumbing Technical Supervisor SS \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3101 Principal Administrative Analyst SS		· · · · · · · · · · · · · · · · · · ·						
5303 Out-of-School Senior Director SS \$31.35510 \$33.12977 \$34.99305 \$36.97285 \$39.05270 3147 Painting Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 3114 Paving Supervisor SS \$10,0000 \$0,00000 \$0,00000 \$44.94206 4137 Pest Management Supervisor SS \$19,58352 \$20,66991 \$21,87066 \$23.09998 \$24.37220 4054 Plant Manager II SS \$16,83183 \$17.79672 \$18,75444 \$19,84799 \$20,97725 \$22,16370 \$23,40017 \$24,71527 \$26,15187 \$27,56703 \$29,14657 \$20,97725 \$20,97725 \$22,16370 \$22,16370 \$23,40017 \$24,71527 \$26,15187 \$27,56703 \$29,14657 \$20,097725 \$20,00772 \$21,77373 \$8,210.79 \$27,56703 \$29,14657 \$20,097725 \$20,00772 \$22,17527 \$26,15187 \$27,56703 \$29,14657 \$20,009725 \$20,00772 \$26,15187 \$27,56703 \$29,14657 \$20,009725		•						
3147 Painting Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 3114 Paving Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$44,94206 1338 Payroll Distribution Supervisor SS \$19.58352 \$20.66991 \$21.87066 \$23.09998 \$24.37220 4137 Pest Management Supervisor SS \$16.88188 \$17.79672 \$18.75444 \$19.84799 \$20.97725 4047 Plant Manager II SS \$19.84799 \$20.97725 \$22.16370 \$23.40017 \$24.71527 4037 Plant Manager III SS \$21.6370 \$23.40017 \$24.71527 \$26.15187 \$27.56703 4021 Plant Manager IIV SS \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3152 Plumbing Technical Supervisor SS \$6,4979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8.659.07 1121 Principal Accountant SS \$326.6304 \$34.55421 \$36.50827 <t< td=""><td></td><td>•</td><td></td><td>\$29.57541</td><td></td><td>\$33.02040</td><td></td><td>\$36.85850</td></t<>		•		\$29.57541		\$33.02040		\$36.85850
3114 Paving Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$44.94206 1338 Payroll Distribution Supervisor SS \$19.58352 \$20.66991 \$21.87066 \$23.09998 \$24.37220 4137 Pest Managernent Supervisor SS \$16.83183 \$17.79672 \$18.75444 \$19.84799 \$20.97725 4047 Plant Manager II SS \$19.84799 \$22.97725 \$22.16370 \$23.40017 \$24.71527 \$26.15187 \$27.56703 4021 Plant Manager III SS \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 4152 Plumbing Technical Supervisor SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Accountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Assignment Technician SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$8.87319 2551 Principal Auditor, Inspector General's Office SS	5303	Out-of-School Senior Director		\$31.35510	\$33.12977	\$34.99305	\$36.97285	\$39.05270
1338 Payroll Distribution Supervisor SS \$19.58352 \$20.66991 \$21.87066 \$23.09998 \$24.37220 4137 Pest Management Supervisor SS \$27.07386 \$28.60339 \$30.25440 \$31,94831 \$33.72082 4047 Plant Manager II SS \$16.83183 \$17.79672 \$18.75444 \$19.84799 \$20.97725 4047 Plant Manager III SS \$19.84799 \$20.216370 \$23.40017 \$24.71527 \$26.15187 \$27.56703 4021 Plant Manager IV SS \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3152 Plumbing Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 1121 Principal Accountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Assignment Technician SS \$23.06425 \$24.3650 \$27.7039.0 \$7,653.88 \$8,081.56 2716 Principal Assignment Technician SS \$39.62	3147	Painting Technical Supervisor	SS	\$6,979.18	\$7,374.84	\$7,777.37	\$8,210.79	\$8,659.07
4137 Pest Management Supervisor SS \$27.07386 \$28.60339 \$30.25440 \$31.94831 \$33.72082 4054 Plant Manager II SS \$16.83183 \$17.79672 \$18.75444 \$19.84799 \$20.97725 4037 Plant Manager III SS \$19.84799 \$22.16370 \$223.40017 \$224.71527 \$26.15187 \$27.56703 4021 Plant Manager IV SS \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3152 Plumbing Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 1121 Principal Accountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Administrative Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 2716 Principal Accountant SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$28.73939 \$46.72259 \$49.36232 2716 Principal Acditin, Inspector G	3114			\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$44.94206
4054 Plant Manager I SS \$16.83183 \$17.79672 \$18.75444 \$19.84799 \$20.97725 4047 Plant Manager II SS \$19.84799 \$20.97725 \$22.16370 \$23.40017 \$24.71527 4037 Plant Manager III SS \$22.16370 \$23.40017 \$24.71527 \$26.15187 \$27.56703 4021 Plant Manager IV SS \$22.16370 \$23.40017 \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3152 Plumbing Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 5101 Principal Accountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Administrative Analyst SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$28.73919 1220 Principal Administrative Analyst SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$28.73919 1220 Principal Clerk SS \$20.6	1338	Payroll Distribution Supervisor	SS	\$19.58352	\$20.66991	\$21.87066	\$23.09998	\$24.37220
4047 Plant Manager II SS \$19.84799 \$20.97725 \$22.16370 \$23.40017 \$24.71527 4037 Plant Manager III SS \$22.16370 \$23.40017 \$24.71527 \$26.15187 \$27.56703 4021 Plant Manager IV SS \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3152 Plumbing Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$2,210.79 \$8,659.07 1121 Principal Adcountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Additor, Inspector General's Office SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$28.73919 1220 Principal Auditor, Inspector General's Office SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36232 2551 Principal Financial Analyst SS \$21.63480 \$22.82839 \$24.07916 \$25.50147 \$26.93807 2251 Principal Personnel Clerk SS \$21.87066 </td <td>4137</td> <td>Pest Management Supervisor</td> <td>SS</td> <td>\$27.07386</td> <td>\$28.60339</td> <td>\$30.25440</td> <td>\$31.94831</td> <td>\$33.72082</td>	4137	Pest Management Supervisor	SS	\$27.07386	\$28.60339	\$30.25440	\$31.94831	\$33.72082
4037 Plant Manager III SS \$22.16370 \$23.40017 \$24.71527 \$26.15187 \$27.56703 4021 Plant Manager IV SS \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3152 Plumbing Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 1121 Principal Accountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Administrative Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$80.81.56 2716 Principal Assignment Technician SS \$23.06425 \$24.36506 \$257.0874 \$27.19538 \$28.73919 1220 Principal Auditor, Inspector General's Office SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36232 2551 Principal Financial Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 4118 Principal Gardener SS \$21.63480	4054	Plant Manager I	SS	\$16.83183	\$17.79672	\$18.75444	\$19.84799	\$20.97725
4021 Plant Manager IV SS \$24.71527 \$26.15187 \$27.56703 \$29.14657 \$30.80474 3152 Plumbing Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 1121 Principal Accountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Administrative Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 2716 Principal Assignment Technician SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$28.73919 1220 Principal Auditor, Inspector General's Office SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36232 2551 Principal Clerk SS \$20.66991 \$21.87066 \$23.09998 \$24.37220 \$25.75163 4118 Principal Gardener SS \$21.63480 \$22.82839 \$24.07916 \$25.50147 \$26.93807 2251 Principal Realty Agent SS \$7,292.50	4047	Plant Manager II	SS	\$19.84799	\$20.97725	\$22.16370	\$23.40017	\$24.71527
3152 Plumbing Technical Supervisor SS \$6,979.18 \$7,374.84 \$7,777.37 \$8,210.79 \$8,659.07 1121 Principal Accountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Administrative Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 2716 Principal Assignment Technician SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$28.73919 1220 Principal Auditor, Inspector General's Office SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36232 2551 Principal Clerk SS \$20.66991 \$21.87066 \$23.09998 \$24.37220 \$25.75163 1116 Principal Financial Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 4118 Principal Gardener SS \$21.63480 \$22.82893 \$24.07916 \$25.50147 \$26.93807 2251 Principal Realty Agent SS \$7,292.50 <td>4037</td> <td>Plant Manager III</td> <td>SS</td> <td>\$22.16370</td> <td>\$23.40017</td> <td>\$24.71527</td> <td>\$26.15187</td> <td>\$27.56703</td>	4037	Plant Manager III	SS	\$22.16370	\$23.40017	\$24.71527	\$26.15187	\$27.56703
1121 Principal Accountant SS \$32.66304 \$34.56421 \$36.50827 \$38.55239 \$40.76804 5101 Principal Administrative Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 2716 Principal Assignment Technician SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$28.73919 1220 Principal Auditor, Inspector General's Office SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36232 2551 Principal Clerk SS \$20.66991 \$21.87066 \$23.09998 \$24.37220 \$25.75163 1116 Principal Financial Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 4118 Principal Gardener SS \$21.63480 \$22.282839 \$24.07916 \$25.50147 \$26.93807 2251 Principal Realty Agent SS \$7,292.50 \$7,701.90 \$8,131.88 \$8,569.87 \$9,045.59 5288 Principal Stock Clerk SS \$23.87190	4021	Plant Manager IV	SS	\$24.71527	\$26.15187	\$27.56703	\$29.14657	\$30.80474
5101 Principal Administrative Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 2716 Principal Assignment Technician SS \$23.06425 \$24.36506 \$25.70874 \$27.19538 \$28.73919 1220 Principal Auditor, Inspector General's Office SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36232 2551 Principal Clerk SS \$20.66991 \$21.87066 \$23.09998 \$24.37220 \$25.75163 1116 Principal Financial Analyst SS \$6,485.15 \$6,846.52 \$7,239.90 \$7,653.88 \$8,081.56 4118 Principal Gardener SS \$21.63480 \$22.82839 \$24.07916 \$25.50147 \$26.93807 2251 Principal Personnel Clerk SS \$21.87066 \$23.09998 \$24.37220 \$25.50147 \$26.93807 2251 Principal Realty Agent SS \$7,292.50 \$7,701.90 \$8,131.88 \$8,569.87 \$9,045.59 5288 Principal Realty Agent SS \$23.87190	3152	Plumbing Technical Supervisor	SS	\$6,979.18	\$7,374.84	\$7,777.37	\$8,210.79	\$8,659.07
2716Principal Assignment TechnicianSS\$23.06425\$24.36506\$25.70874\$27.19538\$28.739191220Principal Auditor, Inspector General's OfficeSS\$39.62028\$41.85906\$44.22392\$46.72259\$49.362322551Principal ClerkSS\$20.66991\$21.87066\$23.09998\$24.37220\$25.751631116Principal Financial AnalystSS\$6,485.15\$6,846.52\$7,239.90\$7,653.88\$8,081.564118Principal GardenerSS\$21.63480\$22.82839\$24.07916\$25.50147\$26.938072251Principal Personnel ClerkSS\$21.87066\$23.09998\$24.37220\$25.75163\$27.238252141Principal Realty AgentSS\$7,292.50\$7,701.90\$8,131.88\$8,569.87\$9,045.595288Principal Stock ClerkSS\$20.58414\$21.74915\$22.97849\$24.27929\$25.644415199Procurement Customer Services SupervisorSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241425Program and Project Controls ManagerSS\$8,873.52\$9,361.21\$9,878.11\$10,415.58\$10,995.375090Program Evaluation and Research CoordinatorSS\$6,798.48\$7,197.59\$7,581.82\$7,999.23\$8,444.071381Quality Control and Assurance Audit SpecialistSS\$30.00000\$0.00000\$0.00000\$0.00000\$40.703355211Receiving InspectorSS\$21.74915\$22.9784	1121	Principal Accountant	SS	\$32.66304	\$34.56421	\$36.50827	\$38.55239	\$40.76804
1220Principal Auditor, Inspector General's OfficeSS\$39.62028\$41.85906\$44.22392\$46.72259\$49.362322551Principal ClerkSS\$20.66991\$21.87066\$23.09998\$24.37220\$25.751631116Principal Financial AnalystSS\$6,485.15\$6,846.52\$7,239.90\$7,653.88\$8,081.564118Principal GardenerSS\$21.63480\$22.82839\$24.07916\$25.50147\$26.938072251Principal Personnel ClerkSS\$21.87066\$23.09998\$24.37220\$25.75163\$27.238252141Principal Realty AgentSS\$7,292.50\$7,701.90\$8,131.88\$8,569.87\$9,045.595288Principal Stock ClerkSS\$20.58414\$21.74915\$22.97849\$24.27929\$25.644415199Procurement Customer Services SupervisorSS\$3.87190\$25.18699\$26.64503\$28.13881\$29.761241425Program and Project Controls ManagerSS\$8,873.52\$9,361.21\$9,878.11\$10,415.58\$10,995.375090Program Evaluation and Research CoordinatorSS\$6,798.48\$7,197.59\$7,581.82\$7,999.23\$8,444.071381Quality Control and Assurance Audit SpecialistSS\$0.00000\$0.00000\$0.00000\$40.103355211Receiving InspectorSS\$21.74915\$22.97849\$24.27929\$25.64441\$27.073862198Records ManagerSS\$6,172.96\$6,516.03\$6,888.83\$7,279.	5101	Principal Administrative Analyst	SS	\$6,485.15	\$6,846.52	\$7,239.90	\$7,653.88	\$8,081.56
2551Principal ClerkSS\$20.66991\$21.87066\$23.09998\$24.37220\$25.751631116Principal Financial AnalystSS\$6,485.15\$6,846.52\$7,239.90\$7,653.88\$8,081.564118Principal GardenerSS\$21.63480\$22.82839\$24.07916\$25.50147\$26.938072251Principal Personnel ClerkSS\$21.87066\$23.09998\$24.37220\$25.75163\$27.238252141Principal Realty AgentSS\$7,292.50\$7,701.90\$8,131.88\$8,569.87\$9,045.595288Principal Stock ClerkSS\$20.58414\$21.74915\$22.97849\$24.27929\$25.644415199Procurement Customer Services SupervisorSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241425Program and Project Controls ManagerSS\$8,873.52\$9,361.21\$9,878.11\$10,415.58\$10,995.375090Program Evaluation and Research CoordinatorSS\$6,798.48\$7,197.59\$7,581.82\$7,999.23\$8,444.071381Quality Control and Assurance Audit SpecialistSS\$39.62028\$41.85906\$44.22392\$46.72259\$49.362353523Radio Communications SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$0.00000\$25.64441\$27.073862198Receiving InspectorSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.04	2716	Principal Assignment Technician	SS	\$23.06425	\$24.36506	\$25.70874	\$27.19538	\$28.73919
1116Principal Financial AnalystSS\$6,485.15\$6,846.52\$7,239.90\$7,653.88\$8,081.564118Principal GardenerSS\$21.63480\$22.82839\$24.07916\$25.50147\$26.938072251Principal Personnel ClerkSS\$21.87066\$23.09998\$24.37220\$25.75163\$27.238252141Principal Realty AgentSS\$7,292.50\$7,701.90\$8,131.88\$8,569.87\$9,045.595288Principal Stock ClerkSS\$20.58414\$21.74915\$22.97849\$24.27929\$25.644415199Procurement Customer Services SupervisorSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241425Program and Project Controls ManagerSS\$8,873.52\$9,361.21\$9,878.11\$10,415.58\$10,995.375090Program Evaluation and Research CoordinatorSS\$6,798.48\$7,197.59\$7,581.82\$7,999.23\$8,444.071381Quality Control and Assurance Audit SpecialistSS\$39.62028\$41.85906\$44.22392\$46.72259\$49.362353523Radio Communications SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$40.103355211Receiving InspectorSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.042198Records ManagerSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.04	1220	Principal Auditor, Inspector General's Office	SS	\$39.62028	\$41.85906	\$44.22392	\$46.72259	\$49.36232
4118Principal GardenerSS\$21.63480\$22.82839\$24.07916\$25.50147\$26.938072251Principal Personnel ClerkSS\$21.87066\$23.09998\$24.37220\$25.75163\$27.238252141Principal Realty AgentSS\$7,292.50\$7,701.90\$8,131.88\$8,569.87\$9,045.595288Principal Stock ClerkSS\$20.58414\$21.74915\$22.97849\$24.27929\$25.644415199Procurement Customer Services SupervisorSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241425Program and Project Controls ManagerSS\$8,873.52\$9,361.21\$9,878.11\$10,415.58\$10,995.375090Program Evaluation and Research CoordinatorSS\$6,798.48\$7,197.59\$7,581.82\$7,999.23\$8,444.071381Quality Control and Assurance Audit SpecialistSS\$39.62028\$41.85906\$44.22392\$46.72259\$49.362353523Radio Communications SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$40.103355211Receiving InspectorSS\$21.74915\$22.97849\$24.27929\$25.64441\$27.073862198Records ManagerSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.04	2551	Principal Clerk	SS	\$20.66991	\$21.87066	\$23.09998	\$24.37220	\$25.75163
2251Principal Personnel ClerkSS\$21.87066\$23.09998\$24.37220\$25.75163\$27.238252141Principal Realty AgentSS\$7,292.50\$7,701.90\$8,131.88\$8,569.87\$9,045.595288Principal Stock ClerkSS\$20.58414\$21.74915\$22.97849\$24.27929\$25.644415199Procurement Customer Services SupervisorSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241425Program and Project Controls ManagerSS\$8,873.52\$9,361.21\$9,878.11\$10,415.58\$10,995.375090Program Evaluation and Research CoordinatorSS\$6,798.48\$7,197.59\$7,581.82\$7,999.23\$8,444.071381Quality Control and Assurance Audit SpecialistSS\$39.62028\$41.85906\$44.22392\$46.72259\$49.362353523Radio Communications SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$0.00000\$40.103355211Receiving InspectorSS\$21.74915\$22.97849\$24.27929\$25.64441\$27.073862198Records ManagerSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.04	1116	Principal Financial Analyst	SS	\$6,485.15	\$6,846.52	\$7,239.90	\$7,653.88	\$8,081.56
2141Principal Realty AgentSS\$7,292.50\$7,701.90\$8,131.88\$8,569.87\$9,045.595288Principal Stock ClerkSS\$20.58414\$21.74915\$22.97849\$24.27929\$25.644415199Procurement Customer Services SupervisorSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241425Program and Project Controls ManagerSS\$8,873.52\$9,361.21\$9,878.11\$10,415.58\$10,995.375090Program Evaluation and Research CoordinatorSS\$6,798.48\$7,197.59\$7,581.82\$7,999.23\$8,444.071381Quality Control and Assurance Audit SpecialistSS\$39.62028\$41.85906\$44.22392\$46.72259\$49.362353523Radio Communications SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$40.103355211Receiving InspectorSS\$21.74915\$22.97849\$24.27929\$25.64441\$27.073862198Records ManagerSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.04	4118	Principal Gardener	SS	\$21.63480	\$22.82839	\$24.07916	\$25.50147	\$26.93807
5288 Principal Stock Clerk SS \$20.58414 \$21.74915 \$22.97849 \$24.27929 \$25.64441 5199 Procurement Customer Services Supervisor SS \$23.87190 \$25.18699 \$26.64503 \$28.13881 \$29.76124 1425 Program and Project Controls Manager SS \$8,873.52 \$9,361.21 \$9,878.11 \$10,415.58 \$10,995.37 5090 Program Evaluation and Research Coordinator SS \$6,798.48 \$7,197.59 \$7,581.82 \$7,999.23 \$8,444.07 1381 Quality Control and Assurance Audit Specialist SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36235 3523 Radio Communications Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$0.00000 \$40.10335 5211 Receiving Inspector SS \$21.74915 \$22.97849 \$24.27929 \$25.64441 \$27.07386 2198 Records Manager SS \$6,516.03 \$6,888.83 \$7,279.93 \$7,687.04	2251	Principal Personnel Clerk	SS	\$21.87066	\$23.09998	\$24.37220	\$25.75163	\$27.23825
5199Procurement Customer Services SupervisorSS\$23.87190\$25.18699\$26.64503\$28.13881\$29.761241425Program and Project Controls ManagerSS\$8,873.52\$9,361.21\$9,878.11\$10,415.58\$10,995.375090Program Evaluation and Research CoordinatorSS\$6,798.48\$7,197.59\$7,581.82\$7,999.23\$8,444.071381Quality Control and Assurance Audit SpecialistSS\$39.62028\$41.85906\$44.22392\$46.72259\$49.362353523Radio Communications SupervisorSS\$0.00000\$0.00000\$0.00000\$0.00000\$40.103355211Receiving InspectorSS\$21.74915\$22.97849\$24.27929\$25.64441\$27.073862198Records ManagerSS\$6,172.96\$6,516.03\$6,888.83\$7,279.93\$7,687.04	2141	Principal Realty Agent	SS	\$7,292.50	\$7,701.90	\$8,131.88	\$8,569.87	\$9,045.59
1425 Program and Project Controls Manager SS \$8,873.52 \$9,361.21 \$9,878.11 \$10,415.58 \$10,995.37 5090 Program Evaluation and Research Coordinator SS \$6,798.48 \$7,197.59 \$7,581.82 \$7,999.23 \$8,444.07 1381 Quality Control and Assurance Audit Specialist SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36235 3523 Radio Communications Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$0.00000 \$40.10335 5211 Receiving Inspector SS \$21.74915 \$22.97849 \$24.27929 \$25.64441 \$27.07386 2198 Records Manager SS \$6,172.96 \$6,516.03 \$6,888.83 \$7,279.93 \$7,687.04	5288	Principal Stock Clerk	SS	\$20.58414	\$21.74915	\$22.97849	\$24.27929	\$25.64441
5090 Program Evaluation and Research Coordinator SS \$6,798.48 \$7,197.59 \$7,581.82 \$7,999.23 \$8,444.07 1381 Quality Control and Assurance Audit Specialist SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36235 3523 Radio Communications Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$0.00000 \$40.10335 5211 Receiving Inspector SS \$21.74915 \$22.97849 \$24.27929 \$25.64441 \$27.07386 2198 Records Manager SS \$6,172.96 \$6,516.03 \$6,888.83 \$7,279.93 \$7,687.04	5199	Procurement Customer Services Supervisor	SS	\$23.87190	\$25.18699	\$26.64503	\$28.13881	\$29.76124
1381 Quality Control and Assurance Audit Specialist SS \$39.62028 \$41.85906 \$44.22392 \$46.72259 \$49.36235 3523 Radio Communications Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$0.00000 \$40.10335 5211 Receiving Inspector SS \$21.74915 \$22.97849 \$24.27929 \$25.64441 \$27.07386 2198 Records Manager SS \$6,172.96 \$6,516.03 \$6,888.83 \$7,279.93 \$7,687.04	1425	Program and Project Controls Manager	SS	\$8,873.52	\$9,361.21	\$9,878.11	\$10,415.58	\$10,995.37
3523 Radio Communications Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$0.00000 \$40.10335 5211 Receiving Inspector SS \$21.74915 \$22.97849 \$24.27929 \$25.64441 \$27.07386 2198 Records Manager SS \$6,516.03 \$6,888.83 \$7,279.93 \$7,687.04	5090	Program Evaluation and Research Coordinator	SS	\$6,798.48	\$7,197.59	\$7,581.82	\$7,999.23	\$8,444.07
5211 Receiving Inspector SS \$21.74915 \$22.97849 \$24.27929 \$25.64441 \$27.07386 2198 Records Manager SS \$6,172.96 \$6,516.03 \$6,888.83 \$7,279.93 \$7,687.04	1381	Quality Control and Assurance Audit Specialist	SS	\$39.62028	\$41.85906	\$44.22392	\$46.72259	\$49.36235
2198 Records Manager SS \$6,172.96 \$6,516.03 \$6,888.83 \$7,279.93 \$7,687.04	3523	Radio Communications Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$40.10335
	5211	Receiving Inspector	SS	\$21.74915	\$22.97849	\$24.27929	\$25.64441	\$27.07386
	2198	Records Manager	SS	\$6,172.96	\$6,516.03	\$6,888.83	\$7,279.93	\$7,687.04
3226 Refrigeration Supervisor SS \$0.00000 \$0.00000 \$0.00000 \$48.24409	3226	Refrigeration Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$48.24409
1663 Relocatable Housing Manufacturing Inspector SS \$0.00000 \$0.00000 \$0.00000 \$0.00000 \$52.12506	1663	Relocatable Housing Manufacturing Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$52.12506

Class	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
Code			-	•	•	•	•
4759	Reprographic Services Coordinator	SS	\$24.02914	\$25.40141	\$26.83802	\$28.36754	\$29.97566
1048	Retirement Systems Supervisor	SS	\$35.25016	\$37.25179	\$39.33146	\$41.58987	\$43.97021
3280	Roofing Inspection Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$42.16176
3281	Roofing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$42.16176
3482	Roofing Technical Supervisor	SS	\$6,979.18	\$7,374.84	\$7,777.37	\$8,210.79	\$8,659.07
4889	SAP Workflow Analyst	SS	\$39.80316	\$42.06885	\$44.45604	\$46.95759	\$49.54490
4325	Satellite Food Service Manager	SS	\$14.85203	\$15.71685	\$16.61741	\$17.53941	\$18.51859
2500	School Administrative Assistant	SS	\$18.51859	\$19.58352	\$20.66991	\$21.87066	\$23.09998
2500A	School Administrative Assistant	SS	\$24.37220	\$25.75163	\$0.00000	\$0.00000	\$0.00000
1775	School Traffic Safety Coordinator	SS	\$26.73080	\$28.23887	\$29.83272	\$31.29790	\$33.30628
1100	Senior Accounting Analyst	SS	\$32.66304	\$34.56421	\$36.50827	\$38.55239	\$40.76804
5070	Senior Administrative Analyst	SS	\$34.56421	\$36.50827	\$38.55239	\$40.76804	\$43.09092
2062	Senior Administrative Assistant	SS	\$29.31098	\$30.97628	\$32.66304	\$34.56421	\$36.50827
2544	Senior Budget Technician	SS	\$20.58414	\$21.79203	\$23.00708	\$24.28643	\$25.68729
1472	Senior Building Project Estimator	SS	\$37.37310	\$39.49583	\$41.74007	\$44.07724	\$46.57878
5721	Senior Bus Dispatcher	SS	\$32.84173	\$34.68572	\$36.64407	\$38.73107	\$40.95387
2192	Senior Community Outreach Organizer	SS	\$34.56421	\$36.50827	\$38.55239	\$40.76804	\$43.09092
1147	Senior Data Center Equipment Operator	SS	\$22.27090	\$23.52882	\$24.85822	\$26.28052	\$27.78145
2615	Senior Early Childhood Education Division Fiscal Technician	SS	\$19.22644	\$20.28402	\$21.46318	\$22.67137	\$23.96500
1665	Senior Electrical Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$42.41907
5033	Senior Employment Services Manager	SS	\$6,650.89	\$7,024.97	\$7,428.06	\$7,850.72	\$8,281.17
1773	Senior Energy Specialist	SS	\$34.11966	\$36.04792	\$38.08164	\$40.23304	\$42.50853
1780	Senior Environmental Health and Safety Officer	SS	\$0.00000	\$0.00000	\$34.11394	\$36.05799	\$38.10926
1114	Senior Financial Analyst	SS	\$34.56421	\$36.50827	\$38.55239	\$40.76804	\$43.09092
1128	Senior Fiscal Specialist	SS	\$34.56421	\$36.50827	\$38.55239	\$40.76804	\$43.09092
4321	Senior Food Service Training Specialist	SS	\$26.90234	\$28.43185	\$30.01140	\$31.71245	\$33.51357
4174	Senior Gardener	SS	\$19.46917	\$20.58414	\$21.74915	\$22.97849	\$24.27929
4998	Senior Human Resources Specialist	SS	\$38.55239	\$40.76804	\$43.09092	\$45.49954	\$47.97250
3860	Senior IT Customer Services Support Representative	SS	\$24.64379	\$26.08040	\$27.52415	\$29.11084	\$30.70468
1365	Senior Labor Compliance Officer	SS	\$34.31262	\$36.30695	\$38.34901	\$40.49985	\$42.83078
2182	Senior Marketing Representative	SS	\$34.56421	\$36.50827	\$38.55239	\$40.76804	\$43.09092
2383	Senior Medical Administrative Assistant	SS	\$34.43557	\$36.37962	\$38.38086	\$40.59651	\$42.87650
4309	Senior Nutrition Specialist	SS	\$34.99305	\$36.97285	\$39.05270	\$41.27550	\$43.55548
1686	Senior Painting Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$38.40945
4143	Senior Pest Management Technician	SS	\$23.87190	\$25.18699	\$26.64503	\$28.13881	\$29.76124
5779	Senior Power Equipment Mechanic	SS	\$27.40264	\$28.96075	\$30.58318	\$32.30567	\$34.12108
	Senior Procurement Customer Services						
5198	Supervisor	SS	\$29.43247	\$30.97628	\$32.66304	\$34.56421	\$36.50827
3809	Senior Programmer Analyst (COBOL)	SS	\$39.80316	\$42.06885	\$44.45604	\$46.95759	\$49.54490
3810	Senior Programmer Analyst (Oracle)	SS	\$39.80316	\$42.06885	\$44.45604	\$46.95759	\$49.54490
	· · · · · · · · · · · · · · · · · · ·						

Class	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
Code	Soniar Dragrammar Analyst (SAD)	SS	¢20 00214	-	¢11 1E401	•	¢ 40 E 4 400
4892 3811	Senior Programmer Analyst (SAP) Senior Programmer Analyst (Visual Basic)	SS	\$39.80316 \$39.80316	\$42.06885 \$42.06885	\$44.45604 \$44.45604	\$46.95759 \$46.95759	\$49.54490 \$49.54490
3819	Senior Programmer Analyst, Filenet	SS	\$39.80316	\$42.06885	\$44.45604 \$44.45604	\$46.95759	\$49.54490
3807	Senior Programmer Analyst, JAVA	SS	\$39.80316	\$42.06885	\$44.45604	\$46.95759	\$49.54490
4436	Senior Radiologic Technologist	SS	\$21.65623	\$22.86414	\$24.17923	\$25.56580	\$26.95237
2144	Senior Realty Agent	SS	\$36.89422	\$39.01697	\$41.13970	\$43.42683	\$45.82831
5164	Senior Stores Supervisor	SS	\$31.94831	\$33.72082	\$35.64345	\$37.67328	\$39.79601
1806	Senior Sustainability Specialist	SS	\$42.68647	\$45.08532	\$47.53675	\$50.18536	\$52.94568
3336	Senior Tractor and Roller Operator	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$36.65836
4131	Senior Tree Surgeon	SS	\$24.27929	\$25.64441	\$27.07386	\$28.60339	\$30.25440
4045	Senior Window/Wall Washer	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$22.26376
1126	Special Education Fiscal Analyst	SS	\$6,485.15	\$6,846.52	\$7,239.90	\$7,653.88	\$8,081.55
5175	Stores Supervisor	SS	\$28.60339	\$30.25440	\$31.94831	\$33.72082	\$35.64345
4974	Strategic Data Analyst	SS	\$6,485.15	\$6,846.52	\$7,239.90	\$7,653.88	\$8,081.55
1836	Structural Engineer	SS	\$39.43151	\$41.61858	\$43.97717	\$46.41440	\$48.93023
4736	Student Records and Data Supervisor	SS	\$20.66991	\$21.87066	\$23.09998	\$24.37220	\$25.75163
2356	Student Support Services Program Coordinator	SS	\$24.78673	\$26.20190	\$27.62422	\$29.21805	\$30.86908
4369	Summer Food Service Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$15.68969
1129	Supervising Accountant	SS	\$27.78145	\$29.31098	\$30.97628	\$32.66304	\$34.56421
1316	Supervising Accounting Technician	SS	\$19.22616	\$20.28396	\$21.46326	\$22.67115	\$23.96481
1440	Supervising Architect	SS	\$42.84074	\$45.27082	\$47.77952	\$50.38828	\$53.17571
2720	Supervising Assignment Technician	SS	\$21.79203	\$23.00708	\$24.28643	\$25.68729	\$27.13820
5758	Supervising Automotive Body Mechanic	SS	\$31.90543	\$33.79230	\$35.65775	\$37.70187	\$39.80316
1631	Supervising Building/Construction Inspector	SS	\$7,936.98	\$8,385.78	\$8,859.37	\$9,359.96	\$9,888.58
1527	Supervising Civil Engineer	SS	\$43.90571	\$46.38581	\$48.96597	\$51.63189	\$54.49797
5648	Supervising Classified Training Representative	SS	\$30.72612	\$32.42003	\$34.29260	\$36.22237	\$38.25220
2576	Supervising Clerk	SS	\$17.54655	\$18.51859	\$19.58352	\$20.66991	\$21.87066
2575	Supervising Clerk (Non-Typing)	SS	\$17.54655	\$18.51859	\$19.58352	\$20.66991	\$21.87066
4818	Supervising Data Center Technician	SS	\$27.78145	\$29.31098	\$30.97628	\$32.66304	\$34.56421
1704	Supervising Electrical Engineer	SS	\$43.90571	\$46.38581	\$48.96597	\$51.63189	\$54.49797
1664	Supervising Electrical Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$45.10645
5214	Supervising Fleet-Parts Storekeeper	SS	\$25.94460	\$27.40980	\$28.99649	\$30.59033	\$32.32712
2773	Supervising Insurance Technician	SS	\$18.32561	\$19.36197	\$20.45549	\$21.63480	\$22.82839
2367	Supervising Job Order Contracting Specialist	SS	\$39.51727	\$41.94020	\$44.17014	\$46.63596	\$49.17323
3778	Supervising Maintenance Worker	SS	\$25.11551	\$26.54497	\$28.06735	\$29.64689	\$31.32651
1754	Supervising Mechanical Engineer	SS	\$43.90571	\$46.38581	\$48.96597	\$51.63189	\$54.49797
2211	Supervising Microcomputer Support Assistant	SS	\$30.45454	\$32.17702	\$34.02102	\$35.92935	\$37.98061
1303	Supervising Payroll Technician	SS	\$20.28396	\$21.46326	\$22.67115	\$23.96481	\$25.29420
2261	Supervising Personnel Clerk	SS	\$19.58352	\$20.66991	\$21.87066	\$23.09998	\$24.37220
1675	Supervising Plumbing Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$43.54833
5783	Supervising Power Equipment Mechanic	SS	\$30.97628	\$32.66304	\$34.56421	\$36.50827	\$38.55239
5144	Supervising Purchasing Services Coordinator	SS	\$6,505.72	\$6,875.11	\$7,266.20	\$7,673.31	\$8,107.88

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
4560	Supervising Special Education Assistant	SS	\$20.92722	\$22.10652	\$23.37158	\$24.67953	\$26.03752
5395	Supervising Staff Development Coordinator (LA's BEST)	SS	\$39.51727	\$41.94020	\$44.17014	\$46.63596	\$49.17323
1832	Supervising Structural Engineer	SS	\$43.90571	\$46.38581	\$48.96597	\$51.63189	\$54.49797
1584	Supervisor, Plots Plans and Allotments	SS	\$24.92254	\$26.47350	\$27.94583	\$29.54682	\$31.24073
1864	Survey Party Chief	SS	\$35.52506	\$37.50780	\$39.62379	\$41.86518	\$44.20843
3155	Swimming Pool Technical Supervisor	SS	\$39.10987	\$41.32553	\$43.59122	\$46.01414	\$48.52998
1053	Telecommunications Billing Supervisor	SS	\$33.01325	\$34.87156	\$36.85134	\$38.92405	\$41.11826
2952	Test Services Coordinator	SS	\$26.68077	\$28.16741	\$29.78269	\$31.47659	\$33.27770
5099	Testing Operations Manager	SS	\$34.43557	\$36.37962	\$38.38086	\$40.59651	\$42.87650
1120	Transportation Cost Analyst	SS	\$28.98934	\$30.60462	\$32.30567	\$34.16396	\$36.10802
5741	Transportation Inspector	SS	\$27.84578	\$29.43247	\$31.08350	\$32.84173	\$34.68572
5708	Transportation Planner	SS	\$33.14904	\$35.03593	\$36.97999	\$39.08844	\$41.31123
5728	Transportation Routing Center Supervisor	SS	\$33.14904	\$35.03593	\$36.97999	\$39.08844	\$41.31123
4127	Tree Maintenance Supervisor	SS	\$27.07386	\$28.60339	\$30.25440	\$31.94831	\$33.72082
5736	Truck Dispatcher	SS	\$29.70406	\$31.36939	\$33.12762	\$35.03593	\$36.97999
1180	Web Architect	SS	\$7,815.12	\$8,247.38	\$8,695.65	\$9,174.81	\$9,681.41
2047	Workers' Compensation Claims Processing Supervisor	SS	\$30.72612	\$32.42003	\$34.29260	\$36.22237	\$38.25220

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
2375	Absence Coordinator	SS	\$40.11809	\$42.42178	\$44.84942	\$47.34998	\$49.92342
1161	Accountant	SS	\$22.71632	\$23.99940	\$25.35538	\$26.80613	\$28.33708
1101	Accounting Analyst	SS	\$29.89720	\$31.59581	\$33.31630	\$35.25549	\$37.23844
1127	Accounting System Specialist	SS	\$30.72097	\$32.47064	\$34.28590	\$36.23239	\$38.31739
2170	Administrative Assistant, Building Program	SS	\$36.29801	\$38.32468	\$40.51903	\$42.79358	\$45.19935
2406	Administrative Secretary I	SS	\$24.34934	\$25.69073	\$27.17793	\$28.70159	\$30.35646
2150	Adult Education Administrative Specialist	SS	\$7,651.81	\$8,078.73	\$8,518.47	\$8,983.88	\$9,480.79
1444	Architect	SS	\$39.43279	\$41.61986	\$43.97459	\$46.41682	\$48.94653
1438	Architectural Associate	SS	\$34.35151	\$36.29801	\$38.32468	\$40.51903	\$42.79358
1460	Architectural Project Facilitator	SS	\$33.42566	\$35.32111	\$37.28217	\$39.43279	\$41.61986
5804	Area Bus Supervisor	SS	\$28.40270	\$30.02112	\$31.70517	\$33.49856	\$35.37943
3271	Area Carpentry Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$39.04641
3206	Area Electrical Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$46.00858
4314	Area Food Services Supervisor	SS	\$5,567.39	\$5,882.33	\$6,213.60	\$6,564.70	\$6,934.45
3211	Area Heating and Air Conditioning Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$48.10815
3236	Area Metal Work Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$45.27226
4016	Area Operations Supervisor	SS	\$5,153.31	\$5,435.58	\$5,745.85	\$6,073.63	\$6,415.39
3276	Area Painting Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$37.01244
3216	Area Plumbing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$47.46662
3165	Asbestos Abatement Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$43.74131
5806	Assistant Area Bus Supervisor	SS	\$26.93007	\$28.40270	\$30.02112	\$31.70517	\$33.49856
1541	Assistant Civil Engineer	SS	\$32.51438	\$34.35151	\$36.29801	\$38.32468	\$40.51903
2057	Assistant Contracts Supervisor	SS	\$33.31630	\$35.25549	\$37.23844	\$39.32344	\$41.58340
4329	Assistant Food Production Manager	SS	\$30.25440	\$31.98220	\$33.79017	\$35.69291	\$37.71231
5755	Assistant Garage Supervisor	SS	\$34.81079	\$36.80102	\$38.87874	\$41.08038	\$43.42053
4173	Assistant Gardening Supervisor	SS	\$25.62512	\$27.06859	\$28.62870	\$30.23983	\$31.94573
3258	Assistant Hardware Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$35.25549
5140	Assistant Logistics Supervisor	SS	\$27.13414	\$28.67263	\$30.29825	\$31.99651	\$33.79029
2218	Assistant Mail Supervisor	SS	\$27.06128	\$28.55579	\$30.16692	\$31.86554	\$33.68081
4574	Assistant Outdoor-Education Center Manager	SS	\$24.48055	\$25.88757	\$27.37478	\$28.88385	\$30.51686
4061	Assistant Plant Manager I	SS	\$15.39693	\$16.22074	\$17.16847	\$18.15265	\$19.12953
4048	Assistant Plant Manager II	SS	\$17.16847	\$18.15265	\$19.12953	\$20.24495	\$21.39680
1791	Assistant Projects Manager	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
1632	Assistant Supervising Building/Construction Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$53.26379
1531	Associate Civil Engineer	SS	\$36.29801	\$38.32468	\$40.51903	\$42.79358	\$45.19935

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
1709	Associate Electrical Engineer	SS	\$34.35151	\$36.29801	\$38.32468	\$40.51903	\$42.79358
1763	Associate Mechanical Engineer	SS	\$34.35151	\$36.29801	\$38.32468	\$40.51903	\$42.79358
1471	Building Project Estimator	SS	\$32.51438	\$34.35151	\$36.29801	\$38.32468	\$40.51903
1651	Building/Construction Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$48.34873
1650	Building/Construction Inspector (Class 1)	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$49.66826
5731	Bus Dispatcher	SS	\$30.02112	\$31.70517	\$33.49856	\$35.37943	\$37.37695
3192	Carpentry Supervisor (Building Moving)	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$39.04641
3148	Carpentry Technical Supervisor	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
2620	Cataloging Supervisor	SS	\$21.08331	\$22.30807	\$23.56198	\$24.85964	\$26.26666
3160	Central Shops Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$40.13994
2110	Chief Broadcasting Engineer	SS	\$7,577.16	\$8,006.41	\$8,435.66	\$8,906.90	\$9,393.30
2511	Chief Clerk	SS	\$24.34934	\$25.69073	\$27.17793	\$28.70159	\$30.35646
1501	Chief Enrollment Analysis Coordinator	SS	\$41.71463	\$44.09124	\$46.50430	\$49.08504	\$51.77512
5192	Chief Inventory Analyst	SS	\$6,296.42	\$6,646.35	\$7,026.61	\$7,425.53	\$7,840.78
1528	Civil Engineer	SS	\$40.22014	\$42.45095	\$44.85671	\$47.34269	\$49.90883
2045	Claims Processing Supervisor	SS	\$31.59581	\$33.31630	\$35.25549	\$37.23844	\$39.32344
2056	Communications Production Facilitator	SS	\$31.46458	\$33.22152	\$35.12428	\$37.11451	\$39.19221
4817	Computer Applications Support Supervisor	SS	\$33.94325	\$35.85330	\$37.88727	\$40.03788	\$42.24682
1792	Construction Safety Specialist	SS	\$34.66500	\$36.64794	\$38.69647	\$40.90542	\$43.19455
2531	Contract Assistant	SS	\$22.08935	\$23.32142	\$24.66281	\$26.07712	\$27.49142
1125	Contract Payments Supervisor	SS	\$28.75992	\$30.35646	\$32.06237	\$33.82661	\$35.78039
2534	Coordinating Contract Assistant	SS	\$24.66281	\$26.07712	\$27.49142	\$29.07340	\$30.72097
2138	Coordinator of Telecommunication Services	SS	\$33.94325	\$35.85330	\$37.88727	\$40.03788	\$42.24682
1990	Cost Recovery Claims Processing Supervisor	SS	\$31.34064	\$33.06843	\$34.97845	\$36.94682	\$39.01724
2236	Credentials and Contract Supervisor	SS	\$24.85964	\$26.26666	\$27.78302	\$29.35042	\$30.99801
1301	Customer Services Center Supervisor	SS	\$19.54809	\$20.65483	\$21.81215	\$23.07699	\$24.37977
1073	Database Specialist	SS	\$0.00000	\$0.00000	\$8,215.19	\$8,666.61	\$9,147.18
4881	Design Network Engineer	SS	\$43.20185	\$45.64406	\$48.13731	\$50.78366	\$53.60498
4600	Disability Support Services Program Assistant	SS	\$21.34576	\$22.54865	\$23.83901	\$25.17312	\$26.55827
4624	Display and Graphic Designer	SS	\$29.07340	\$30.72097	\$32.47064	\$34.28590	\$36.23239
5739	Driver Trainer (Truck)	SS	\$30.02112	\$31.70517	\$33.49856	\$35.37943	\$37.37695
5734	Driver-Trainer (Bus-Truck)	SS	\$30.02112	\$31.70517	\$33.49856	\$35.37943	\$37.37695
3126	Electrical Technical Supervisor	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
3522	Electronics Technical Supervisor	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
5034	Employment Services Manager	SS	\$35.12428	\$37.09993	\$39.22868	\$41.45948	\$43.73403
1778	Environmental Assessment Coordinator	SS	\$34.66500	\$36.64794	\$38.69647	\$40.90542	\$43.19455

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
3015	Environmental Compliance Coordinator	SS	\$29.38686	\$31.06363	\$32.80597	\$34.66500	\$36.64794
1801	Environmental Health Supervisor	SS	\$38.71674	\$40.92389	\$43.21323	\$45.66070	\$48.23465
2143	Escrow and Title Supervisor	SS	\$42.72798	\$45.12645	\$47.64886	\$50.21502	\$52.99988
1998	Ethics Adviser	SS	\$34.08178	\$36.02096	\$38.02577	\$40.19098	\$42.45823
1963	Facilities Cost Analyst	SS	\$33.31630	\$35.25549	\$37.23844	\$39.32344	\$41.58340
1435	Facilities Partnership and Development Coordinator	SS	\$40.89083	\$43.20914	\$45.57116	\$48.10815	\$50.73991
1813	Facilities Services Coordinator	SS	\$32.75495	\$34.62126	\$36.56045	\$38.63088	\$40.81064
5748	Fleet Technical Support Specialist	SS	\$36.80102	\$38.87874	\$41.08038	\$43.42053	\$45.77528
3251	Floor Covering Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$38.72564
3128	Floor Covering Technical Supervisor	SS	\$40.91123	\$43.23222	\$45.59113	\$48.13346	\$50.75801
4330	Food Production Supervisor	SS	\$19.39928	\$20.50739	\$21.70298	\$22.90586	\$24.17436
3121	Food Service Equipment Specialist	SS	\$39.32344	\$41.58340	\$43.95274	\$46.40953	\$48.93195
4291	Food Service Manager I	SS	\$17.38718	\$18.36406	\$19.39928	\$20.50739	\$21.70298
4292	Food Service Manager II	SS	\$18.36406	\$19.39928	\$20.50739	\$21.70298	\$22.90586
4293	Food Service Manager III	SS	\$19.39928	\$20.50739	\$21.70298	\$22.90586	\$24.17436
4296	Food Service Manager IV	SS	\$20.50739	\$21.70298	\$22.90586	\$24.17436	\$25.53765
4297	Food Service Manager V	SS	\$21.70298	\$22.90586	\$24.17436	\$25.53765	\$26.98111
4294	Food Service Manager VI	SS	\$22.90586	\$24.17436	\$25.53765	\$26.98111	\$28.51205
4295	Food Service Manager VII	SS	\$24.17436	\$25.53765	\$26.98111	\$28.51205	\$30.10860
4312	Food Service Training Specialist	SS	\$24.60449	\$25.98964	\$27.44039	\$29.00049	\$30.61163
1071	Food Services Cost Analyst	SS	\$33.31630	\$35.25549	\$37.23844	\$39.32344	\$41.58340
5756	Garage Supervisor	SS	\$36.80102	\$38.87874	\$41.08038	\$43.42053	\$45.77528
3266	Glazing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$40.91999
3051	Graphic Arts Machinist Supervisor	SS	\$35.79497	\$37.82894	\$39.95769	\$42.22494	\$44.60155
3256	Hardware Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$39.35989
5201	Head Stock Clerk	SS	\$22.18413	\$23.43806	\$24.76488	\$26.15730	\$27.61534
3137	Heating and Air Conditioning Technical Supervisor	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
3132	HVAC Technical Supervisor (High-Rise)	SS	\$40.89083	\$43.20914	\$45.57116	\$48.10815	\$50.73991
3134	HVAC Test and Inspection Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$48.10815
1782	Industrial Hygienist	SS	\$34.66500	\$36.64794	\$38.69625	\$40.90542	\$43.19455
4620	Instructional Media Technician-Adviser	SS	\$31.77806	\$33.60063	\$35.48878	\$37.49359	\$39.63692
4156	Integrated Pest Management Program Coordinator	SS	\$32.75495	\$34.62126	\$36.56045	\$38.63088	\$40.81064
1151	Intermediate Accountant	SS	\$25.35538	\$26.80613	\$28.33708	\$29.89720	\$31.59581
3031	IT Electronics Communications Supervisor	SS	\$34.81079	\$36.77915	\$38.88602	\$41.05850	\$43.36952

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
3543	IT Field Services Supervisor	SS	\$31.06363	\$32.82056	\$34.70144	\$36.64794	\$38.74022
4880	IT Infrastructure Project Network Engineer	SS	\$34.99304	\$36.96870	\$39.05370	\$41.26264	\$43.59550
1193	IT Project Manager	SS	\$6,983.45	\$7,384.70	\$7,806.96	\$8,243.19	\$8,691.10
4812	IT Training Project Manager	SS	\$40.11809	\$42.42178	\$44.84942	\$47.34998	\$49.92342
4863	IT Training Supervisor	SS	\$35.25549	\$37.23844	\$39.32344	\$41.58340	\$43.95274
2052	Labor Compliance Officer	SS	\$31.40999	\$33.19455	\$35.00210	\$37.03940	\$39.12269
4116	Landscaping Supervisor	SS	\$33.01012	\$34.89098	\$36.85206	\$38.93706	\$41.13141
5135	Logistics Supervisor	SS	\$30.29825	\$31.99651	\$33.79029	\$35.73660	\$37.71974
3056	Machinist Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$40.24930
2214	Mail Supervisor	SS	\$30.16692	\$31.86554	\$33.68081	\$35.57626	\$37.59567
3158	Maintenance Materiel Coordinator	SS	\$29.07340	\$30.72097	\$32.47064	\$34.28590	\$36.23239
3020	Maintenance Planner	SS	\$40.89083	\$43.20914	\$45.57116	\$48.10815	\$50.73991
2183	Marketing Representative	SS	\$30.02112	\$31.70517	\$33.49856	\$35.37943	\$37.37695
1959	Masterplan Coordinator	SS	\$40.89083	\$43.20914	\$45.57116	\$48.10815	\$50.73991
1757	Mechanical Engineer	SS	\$40.22014	\$42.45095	\$44.85671	\$47.34269	\$49.90883
2385	Medical Administrative Assistant	SS	\$27.21439	\$28.73076	\$30.38563	\$32.10612	\$33.93597
3140	Metal Technical Supervisor	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
2561	Microfilm Supervisor	SS	\$19.97519	\$21.08331	\$22.30807	\$23.56198	\$24.85964
4844	Mobile Device Management Specialist	SS	\$0.00000	\$0.00000	\$8,215.19	\$8,666.61	\$9,147.18
3081	Musical Instrument Repair Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$39.87021
3588	Network Operations Supervisor	SS	\$36.80102	\$38.87874	\$41.08038	\$43.42053	\$45.77528
1178	Network Security Administrator	SS	\$37.92372	\$40.06704	\$42.34888	\$44.74008	\$47.18959
4317	Nutrition Specialist	SS	\$31.98220	\$33.79017	\$35.69291	\$37.71231	\$39.83375
5387	Operations Compliance Manager (LA's BEST)	SS	\$41.16058	\$43.42053	\$45.80445	\$48.33415	\$50.99508
5387A	Operations Compliance Manager (LA's BEST)	SS	\$52.48229	\$53.95492	\$0.00000	\$0.00000	\$0.00000
4009	Operations Training Specialist	SS	\$6,940.29	\$7,332.21	\$7,743.96	\$8,172.04	\$8,615.30
4591	Outdoor Education Center Manager	SS	\$4,764.89	\$5,028.49	\$5,311.94	\$5,611.71	\$5,930.16
5307	Outdoor Education Programs Coordinator	SS	\$21.94356	\$23.19746	\$24.48055	\$25.88757	\$27.37478
5370	Out-of-School Program Coordinator	SS	\$22.18413	\$23.43806	\$24.76488	\$26.15730	\$27.61534
5305	Out-of-School Regional Director	SS	\$30.16692	\$31.85895	\$33.68081	\$35.57626	\$37.59567
5303	Out-of-School Senior Director	SS	\$31.98220	\$33.79237	\$35.69291	\$37.71231	\$39.83375
3147	Painting Technical Supervisor	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
3114	Paving Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$45.84090
1338	Payroll Distribution Supervisor	SS	\$19.97519	\$21.08331	\$22.30807	\$23.56198	\$24.85964
4137	Pest Management Supervisor	SS	\$27.61534	\$29.17546	\$30.85949	\$32.58728	\$34.39524
4054	Plant Manager I	SS	\$17.16847	\$18.15265	\$19.12953	\$20.24495	\$21.39680

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
4047	Plant Manager II	SS	\$20.24495	\$21.39680	\$22.60697	\$23.86817	\$25.20958
4037	Plant Manager III	SS	\$22.60697	\$23.86817	\$25.20958	\$26.67491	\$28.11837
4021	Plant Manager IV	SS	\$25.20958	\$26.67491	\$28.11837	\$29.72950	\$31.42083
3152	Plumbing Technical Supervisor	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
1121	Principal Accountant	SS	\$33.31630	\$35.25549	\$37.23844	\$39.32344	\$41.58340
5101	Principal Administrative Analyst	SS	\$6,614.85	\$6,983.45	\$7,384.70	\$7,806.96	\$8,243.19
2716	Principal Assignment Technician	SS	\$23.52554	\$24.85236	\$26.22291	\$27.73929	\$29.31397
1220	Principal Auditor, Inspector General's Office	SS	\$40.41269	\$42.69624	\$45.10840	\$47.65704	\$50.34957
2551	Principal Clerk	SS	\$21.08331	\$22.30807	\$23.56198	\$24.85964	\$26.26666
1116	Principal Financial Analyst	SS	\$6,614.85	\$6,983.45	\$7,384.70	\$7,806.96	\$8,243.19
4118	Principal Gardener	SS	\$22.06750	\$23.28496	\$24.56074	\$26.01150	\$27.47683
2251	Principal Personnel Clerk	SS	\$22.30807	\$23.56198	\$24.85964	\$26.26666	\$27.78302
2141	Principal Realty Agent	SS	\$7,438.35	\$7,855.94	\$8,294.52	\$8,741.27	\$9,226.50
5288	Principal Stock Clerk	SS	\$20.99582	\$22.18413	\$23.43806	\$24.76488	\$26.15730
5199	Procurement Customer Services Supervisor	SS	\$24.34934	\$25.69073	\$27.17793	\$28.70159	\$30.35646
1425	Program and Project Controls Manager	SS	\$9,050.99	\$9,548.43	\$10,075.67	\$10,623.89	\$11,215.28
5090	Program Evaluation and Research Coordinator	SS	\$6,934.45	\$7,341.54	\$7,733.46	\$8,159.21	\$8,612.95
1381	Quality Control and Assurance Audit Specialist	SS	\$40.41269	\$42.69624	\$45.10840	\$47.65704	\$50.34960
3523	Radio Communications Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$40.90542
5211	Receiving Inspector	SS	\$22.18413	\$23.43806	\$24.76488	\$26.15730	\$27.61534
2198	Records Manager	SS	\$6,296.42	\$6,646.35	\$7,026.61	\$7,425.53	\$7,840.78
3226	Refrigeration Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$49.20897
1663	Relocatable Housing Manufacturing Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$53.16756
4759	Reprographic Services Coordinator	SS	\$24.50972	\$25.90944	\$27.37478	\$28.93489	\$30.57517
1048	Retirement Systems Supervisor	SS	\$35.95516	\$37.99683	\$40.11809	\$42.42167	\$44.84961
3280	Roofing Inspection Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$43.00500
3281	Roofing Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$43.00500
3482	Roofing Technical Supervisor	SS	\$7,118.76	\$7,522.34	\$7,932.92	\$8,375.01	\$8,832.25
4889	SAP Workflow Analyst	SS	\$40.59922	\$42.91023	\$45.34516	\$47.89674	\$50.53580
4325	Satellite Food Service Manager	SS	\$15.14907	\$16.03119	\$16.94976	\$17.89020	\$18.88896
2500	School Administrative Assistant	SS	\$18.88896	\$19.97519	\$21.08331	\$22.30807	\$23.56198
2500A	School Administrative Assistant	SS	\$24.85964	\$26.26666	\$0.00000	\$0.00000	\$0.00000
1775	School Traffic Safety Coordinator	SS	\$27.26542	\$28.80365	\$30.42937	\$31.92386	\$33.97241
1100	Senior Accounting Analyst	SS	\$33.31630	\$35.25549	\$37.23844	\$39.32344	\$41.58340
5070	Senior Administrative Analyst	SS	\$35.25549	\$37.23844	\$39.32344	\$41.58340	\$43.95274
2062	Senior Administrative Assistant	SS	\$29.89720	\$31.59581	\$33.31630	\$35.25549	\$37.23844

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
2544	Senior Budget Technician	SS	\$20.99582	\$22.22787	\$23.46722	\$24.77216	\$26.20104
1472	Senior Building Project Estimator	SS	\$38.12056	\$40.28575	\$42.57487	\$44.95878	\$47.51036
5721	Senior Bus Dispatcher	SS	\$33.49856	\$35.37943	\$37.37695	\$39.50569	\$41.77295
2192	Senior Community Outreach Organizer	SS	\$35.25549	\$37.23844	\$39.32344	\$41.58340	\$43.95274
1147	Senior Data Center Equipment Operator	SS	\$22.71632	\$23.99940	\$25.35538	\$26.80613	\$28.33708
2615	Senior Early Childhood Education Division Fiscal Technician	SS	\$19.61097	\$20.68970	\$21.89244	\$23.12480	\$24.44430
1665	Senior Electrical Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$43.26745
5033	Senior Employment Services Manager	SS	\$6,783.91	\$7,165.47	\$7,576.62	\$8,007.73	\$8,446.79
1773	Senior Energy Specialist	SS	\$34.80205	\$36.76888	\$38.84327	\$41.03770	\$43.35870
1780	Senior Environmental Health and Safety Officer	SS	\$0.00000	\$0.00000	\$34.79622	\$36.77915	\$38.87145
1114	Senior Financial Analyst	SS	\$35.25549	\$37.23844	\$39.32344	\$41.58340	\$43.95274
1128	Senior Fiscal Specialist	SS	\$35.25549	\$37.23844	\$39.32344	\$41.58340	\$43.95274
4321	Senior Food Service Training Specialist	SS	\$27.44039	\$29.00049	\$30.61163	\$32.34670	\$34.18384
4174	Senior Gardener	SS	\$19.85855	\$20.99582	\$22.18413	\$23.43806	\$24.76488
4998	Senior Human Resources Specialist	SS	\$39.32344	\$41.58340	\$43.95274	\$46.40953	\$48.93195
3860	Senior IT Customer Services Support Representative	SS	\$25.13667	\$26.60201	\$28.07463	\$29.69306	\$31.31877
1365	Senior Labor Compliance Officer	SS	\$34.99887	\$37.03309	\$39.11599	\$41.30985	\$43.68740
2182	Senior Marketing Representative	SS	\$35.25549	\$37.23844	\$39.32344	\$41.58340	\$43.95274
2383	Senior Medical Administrative Assistant	SS	\$35.12428	\$37.10721	\$39.14848	\$41.40844	\$43.73403
4309	Senior Nutrition Specialist	SS	\$35.69291	\$37.71231	\$39.83375	\$42.10101	\$44.42659
1686	Senior Painting Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$39.17764
4143	Senior Pest Management Technician	SS	\$24.34934	\$25.69073	\$27.17793	\$28.70159	\$30.35646
5779	Senior Power Equipment Mechanic	SS	\$27.95069	\$29.53997	\$31.19484	\$32.95178	\$34.80350
5198	Senior Procurement Customer Services Supervisor	SS	\$30.02112	\$31.59581	\$33.31630	\$35.25549	\$37.23844
3809	Senior Programmer Analyst (COBOL)	SS	\$40.59922	\$42.91023	\$45.34516	\$47.89674	\$50.53580
3810	Senior Programmer Analyst (Oracle)	SS	\$40.59922	\$42.91023	\$45.34516	\$47.89674	\$50.53580
4892	Senior Programmer Analyst (SAP)	SS	\$40.59922	\$42.91023	\$45.34516	\$47.89674	\$50.53580
3811	Senior Programmer Analyst (Visual Basic)	SS	\$40.59922	\$42.91023	\$45.34516	\$47.89674	\$50.53580
3819	Senior Programmer Analyst, Filenet	SS	\$40.59922	\$42.91023	\$45.34516	\$47.89674	\$50.53580
3807	Senior Programmer Analyst, JAVA	SS	\$40.59922	\$42.91023	\$45.34516	\$47.89674	\$50.53580
4436	Senior Radiologic Technologist	SS	\$22.08935	\$23.32142	\$24.66281	\$26.07712	\$27.49142
2144	Senior Realty Agent	SS	\$37.63210	\$39.79731	\$41.96249	\$44.29537	\$46.74488
5164	Senior Stores Supervisor	SS	\$32.58728	\$34.39524	\$36.35632	\$38.42675	\$40.59193
1806	Senior Sustainability Specialist	SS	\$43.54020	\$45.98703	\$48.48749	\$51.18907	\$54.00459

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
3336	Senior Tractor and Roller Operator	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$37.39153
4131	Senior Tree Surgeon	SS	\$24.76488	\$26.15730	\$27.61534	\$29.17546	\$30.85949
4045	Senior Window/Wall Washer	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$22.70904
1126	Special Education Fiscal Analyst	SS	\$6,614.85	\$6,983.45	\$7,384.70	\$7,806.96	\$8,243.18
5175	Stores Supervisor	SS	\$29.17546	\$30.85949	\$32.58728	\$34.39524	\$36.35632
4974	Strategic Data Analyst	SS	\$6,614.85	\$6,983.45	\$7,384.70	\$7,806.96	\$8,243.18
1836	Structural Engineer	SS	\$40.22014	\$42.45095	\$44.85671	\$47.34269	\$49.90883
4736	Student Records and Data Supervisor	SS	\$21.08331	\$22.30807	\$23.56198	\$24.85964	\$26.26666
2356	Student Support Services Program Coordinator	SS	\$25.28246	\$26.72594	\$28.17670	\$29.80241	\$31.48646
4369	Summer Food Service Supervisor	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$16.00348
1129	Supervising Accountant	SS	\$28.33708	\$29.89720	\$31.59581	\$33.31630	\$35.25549
1316	Supervising Accounting Technician	SS	\$19.61068	\$20.68964	\$21.89253	\$23.12457	\$24.44411
1440	Supervising Architect	SS	\$43.69755	\$46.17624	\$48.73511	\$51.39605	\$54.23922
2720	Supervising Assignment Technician	SS	\$22.22787	\$23.46722	\$24.77216	\$26.20104	\$27.68096
5758	Supervising Automotive Body Mechanic	SS	\$32.54354	\$34.46815	\$36.37091	\$38.45591	\$40.59922
1631	Supervising Building/Construction Inspector	SS	\$8,095.72	\$8,553.50	\$9,036.56	\$9,547.16	\$10,086.35
1527	Supervising Civil Engineer	SS	\$44.78382	\$47.31353	\$49.94529	\$52.66453	\$55.58793
5648	Supervising Classified Training Representative	SS	\$31.34064	\$33.06843	\$34.97845	\$36.94682	\$39.01724
2576	Supervising Clerk	SS	\$17.89748	\$18.88896	\$19.97519	\$21.08331	\$22.30807
2575	Supervising Clerk (Non-Typing)	SS	\$17.89748	\$18.88896	\$19.97519	\$21.08331	\$22.30807
4818	Supervising Data Center Technician	SS	\$28.33708	\$29.89720	\$31.59581	\$33.31630	\$35.25549
1704	Supervising Electrical Engineer	SS	\$44.78382	\$47.31353	\$49.94529	\$52.66453	\$55.58793
1664	Supervising Electrical Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$46.00858
5214	Supervising Fleet-Parts Storekeeper	SS	\$26.46349	\$27.95800	\$29.57642	\$31.20214	\$32.97366
2773	Supervising Insurance Technician	SS	\$18.69212	\$19.74921	\$20.86460	\$22.06750	\$23.28496
2367	Supervising Job Order Contracting Specialist	SS	\$40.30762	\$42.77900	\$45.05354	\$47.56868	\$50.15669
3778	Supervising Maintenance Worker	SS	\$25.61782	\$27.07587	\$28.62870	\$30.23983	\$31.95304
1754	Supervising Mechanical Engineer	SS	\$44.78382	\$47.31353	\$49.94529	\$52.66453	\$55.58793
2211	Supervising Microcomputer Support Assistant	SS	\$31.06363	\$32.82056	\$34.70144	\$36.64794	\$38.74022
1303	Supervising Payroll Technician	SS	\$20.68964	\$21.89253	\$23.12457	\$24.44411	\$25.80008
2261	Supervising Personnel Clerk	SS	\$19.97519	\$21.08331	\$22.30807	\$23.56198	\$24.85964
1675	Supervising Plumbing Inspector	SS	\$0.00000	\$0.00000	\$0.00000	\$0.00000	\$44.41930
5783	Supervising Power Equipment Mechanic	SS	\$31.59581	\$33.31630	\$35.25549	\$37.23844	\$39.32344
5144	Supervising Purchasing Services Coordinator	SS	\$6,635.83	\$7,012.61	\$7,411.52	\$7,826.78	\$8,270.04
4560	Supervising Special Education Assistant	SS	\$21.34576	\$22.54865	\$23.83901	\$25.17312	\$26.55827
5395	Supervising Staff Development Coordinator (LA's BEST)	SS	\$40.30762	\$42.77900	\$45.05354	\$47.56868	\$50.15669

Class Code	Class Title	Unit	Step 1	Step 2	Step 3	Step 4	Step 5
1832	Supervising Structural Engineer	SS	\$44.78382	\$47.31353	\$49.94529	\$52.66453	\$55.58793
1584	Supervisor, Plots Plans and Allotments	SS	\$25.42099	\$27.00297	\$28.50475	\$30.13776	\$31.86554
1864	Survey Party Chief	SS	\$36.23556	\$38.25796	\$40.41627	\$42.70248	\$45.09260
3155	Swimming Pool Technical Supervisor	SS	\$39.89207	\$42.15204	\$44.46304	\$46.93442	\$49.50058
1053	Telecommunications Billing Supervisor	SS	\$33.67352	\$35.56899	\$37.58837	\$39.70253	\$41.94063
2952	Test Services Coordinator	SS	\$27.21439	\$28.73076	\$30.37834	\$32.10612	\$33.94325
5099	Testing Operations Manager	SS	\$35.12428	\$37.10721	\$39.14848	\$41.40844	\$43.73403
1120	Transportation Cost Analyst	SS	\$29.56913	\$31.21671	\$32.95178	\$34.84724	\$36.83018
5741	Transportation Inspector	SS	\$28.40270	\$30.02112	\$31.70517	\$33.49856	\$35.37943
5708	Transportation Planner	SS	\$33.81202	\$35.73665	\$37.71959	\$39.87021	\$42.13745
5728	Transportation Routing Center Supervisor	SS	\$33.81202	\$35.73665	\$37.71959	\$39.87021	\$42.13745
4127	Tree Maintenance Supervisor	SS	\$27.61534	\$29.17546	\$30.85949	\$32.58728	\$34.39524
5736	Truck Dispatcher	SS	\$30.29814	\$31.99678	\$33.79017	\$35.73665	\$37.71959
1180	Web Architect	SS	\$7,971.42	\$8,412.33	\$8,869.56	\$9,358.31	\$9,875.04
2047	Workers' Compensation Claims Processing Supervisor	SS	\$31.34064	\$33.06843	\$34.97845	\$36.94682	\$39.01724

APPENDIX K

RESPECTFUL TREATMENT

- 1.0 The Union and the District agree to establish a joint Committee to address respectful treatment and bullying concerns of unit members. The Committee shall be advisory in nature and will be comprised of three (3) members appointed by the Union and three (3) members appointed by the District. The Committee shall meet on a quarterly basis and will be tasked with:
 - a. <u>Providing advice to the Superintendent regarding BUL 5798.0 Workplace Violence, Bullying and Threats (Adult to Adult).</u>
 - b. <u>Providing advice to the Superintendent regarding recommended training for District staff in the above area.</u>

APPENDIX L

"Memorandum of Understanding By and Between the Los Angeles Unified School District and Teamsters Local 572 (Unit S - Classified Supervisors)

The Los Angeles Unified School District ("District") Teamsters Local 572 ("Teamsters") hereby enter into the following Memorandum of Understanding ("MOU") for the purpose of implementing the salary reopener provision in their July 1, 2014 MOU.

Background

As reflected in their July 2014 MOU, the Teamsters and the District agreed to the following salary provisions: (a) a 2% off-schedule, lump-sum salary payment, on a one-time basis, based upon and limited to all actual 2013-14 earnings paid on the base salary table; (b) a 2% on-schedule wage increase applied to the base salary table and to all pay scale groups and levels, effective July 1, 2014; (c) a 2% on-schedule wage increase applied to the base salary table and to all pay scale groups and levels, effective July 1, 2015, but contingent upon identification of sufficient funding; and (d) a 2.5% on-schedule wage increase applied to the base salary table and to all pay scale groups and levels, effective July 1, 2016, but contingent upon identification of sufficient funding.

The July 2014 MOU further provided that, "In the event that another LAUSD bargaining unit's across-the-board percentage wage increase above the previous base salary rates exceed those provided in this Agreement, the salary negotiations for this bargaining unit shall, upon request, be reopened for salary renegotiations relating to such year(s).";

On April 17, 2015, the District and United Teachers of Los Angeles ("UTLA") entered into an Agreement, later approved by the Board of Education, providing for: (a) a 4% onschedule wage increase applied to all pay scale groups and levels of the base salary tables, effective July 1, 2014; (b) a 2% on-schedule wage increase applied to all pay scale groups and levels of the base salary tables, effective January 1, 2015; (c) a 2% on-schedule wage increase applied to all pay scale groups and levels of the base salary tables, effective July 1, 2015; (d) a 2% on-schedule wage increase applied to all pay scale groups and levels of the base salary tables, effective January 1, 2016; and (e) salary negotiations reopener negotiations for the 2016-17 school year. Unlike the District-Teamsters July 2014 MOU, the UTLA Agreement did not include any provision for a 2013-14 wage increase, or for a 2.5% salary increase for the 2016-17 school year.

Agreement

The provisions below are to replace the salary provisions of the July 2014 District-Teamsters MOU with the following terms of comparable treatment for currently active employees:

- 1. The 2% off-schedule lump sum salary payment of 2013-14 will, in effect, be credited against the 2014-15 salary increases described below.
- 2. Subject to the above, and also effective July 1, 2014, Unit S employees shall receive a 4% on-schedule wage increase applied to all pay scale groups and levels of the base salary tables.
- 3. Effective January 1, 2015, Unit S employees shall receive a 2% on-schedule wage increase applied to all pay scale groups and levels of the base salary tables.
- 4. Effective July 1, 2015, Unit S employees shall receive a 2% on-schedule wage increase applied to all pay scale groups and levels of the base salary tables.
- 5. Effective January 1, 2016, Unit S employees shall receive a 2% on-schedule wage increase applied to all pay scale groups and levels of the base salary tables.
- 6. In lieu of the salary increase previously required in the 2014 MOU provisions for 2016-17, there shall be salary contract reopener negotiations, commencing April 1, 2016.

This MOU is subject to ratification by the Unit S membership and to final approval by the District's Board of Education.

Date of Agreement: <u>5 29 15</u>	
Los Angeles Unified School District	Teamsters, Local 572
By: Why Molyan Vivian Ekchian Office of Labor Relations	By: K.E. Middleton Secretary-Treasurer
(By: Ligra Car An la Adressa Business Representative

Business Representative